## Oak Brook Park District

## Tennis Center Reflective Ceiling and Insulation Project

## **Bid Packet**



Alin Pop, Tennis Center Manager 630-645-9510 apop@obparks.org

Dave Thommes, Director of Recreation 630-645-9534 <a href="https://dthommes@obparks.org">dthommes@obparks.org</a>

Laure Kosey, Executive Director 630-645-9535 lkosey@obparks.org

## INDEX TENNIS CENTER REFLECTIVE CEILING AND INSULATION PROJECT

INSTRUCTIONS TO BIDDERS	5
CONDITIONS OF THE CONTRACT	14
BID FORM	20
LIST OF SUBCONTRACTORS	23
BIDDER'S REFERENCES	24
CONTRACTOR'S CERTIFICATIONS	25
PREVAILING WAGE NOTICE	29
AIA A 101 OWNER/CONTRACTOR AGREEMENT	30
AIA A201 GENERAL CONDITIONS	31
TECHNICAL SPECIFICATIONS	32

### APPENDIX

 Oak Brook Park District Prevailing Wage Resolution and July 2015 pay rates from the Illinois Department of Labor.

<sup>\*</sup> AIA documents are separately paginated.

# INVITATION TO BID OAK BROOK PARK DISTRICT TENNIS CENTER REFLECTIVE CEILING AND INSULATION PROJECT

The Oak Brook Park District (the "District") is accepting bids for the Tennis Center Reflective Ceiling and Insulation Project.

Specifications and Contract Documents may be obtained beginning October 12, 2016, at the Administrative Office at the District's Family Recreation Center, 1450 Forest Gate Road, Oak Brook, IL 60523, Monday - Friday, 9:00 a.m. – 5:00 p.m., or in PDF format at the District's website: http://www.obparks.org/general information/bid.asp.

Each bid must be placed in a sealed opaque envelope with the Bidder's name, the date and time of the bid deadline and marked "Sealed Bid: – Oak Brook Park District Tennis Center Reflective Ceiling and Insulation Project," and addressed to the Oak Brook Park District, 1450 Forest Gate Road, Oak Brook, IL 60523, Attention: Executive Director. Bids will be received until 11:00 a.m. on November 2, 2016, at which time the bid proposals will be publicly opened and read aloud at the District's Administrative Office, located at the District's Family Recreation Center, 1450 Forest Gate Road, Oak Brook, IL 60523.

A mandatory pre-bid meeting will be held for this Project at the Tennis Center, 1300 Forest Gate Road, Oak Brook, IL 60523, on Tuesday, October 25, 2016 at 2:30 p.m. All potential Bidders for this Project must attend the pre-bid meeting. Any bid received by the Oak Brook Park District from a Bidder not in attendance at the pre-bid meeting will be considered non-responsive and returned unopened to the Bidder.

The Oak Brook Park District Board of Park Commissioners reserves the right to waive all technicalities, to accept or reject any or all bids, to accept only portions of a proposal and reject the remainder without disclosure for any reason. Failure to make such a disclosure will not result in accrual of any right, claim or cause of action by any Bidder against the Oak Brook Park District.

Bids shall not include federal excise tax or state sales tax for materials and equipment to be incorporated in, or fully consumed in the performance of, the work. An Exemption Certificate will be furnished by the Oak Brook Park District on request of the Bidder, for use in connection with this project only.

The Work of this Project is subject to the Illinois *Prevailing Wage Act*, 820 ILCS 130/0.01 *et seq*. A prevailing wage determination has been made by the District, which is the same as that determined by the Illinois Department of Labor for public works projects in DuPage County. The Contract entered into for the Work will be drawn in compliance with said law and proposals should be prepared accordingly and provide for payment of all laborers, workmen, and mechanics needed to perform the Work at no less than the prevailing rate of wages (or the prevailing rate for legal holiday and overtime work) for each craft, type of worker, or mechanic.

All bid proposals must be accompanied by a bid bond or bank cashier's check payable to the Oak Brook Park District for ten percent (10%) of the amount of the bid as provided in the Instructions to Bidders. No proposals or bids will be considered unless accompanied by such bond or check.

The Contractor selected will also be required to comply with all applicable federal, state and local laws, rules, regulations and executive orders including but not limited to those pertaining to equal employment opportunity.

The District encourages women and minority business firms to submit bids and encourages bidders to utilize minority businesses for supplies, equipment and services.

Laure Kosey, Executive Director Oak Brook Park District

# INSTRUCTIONS TO BIDDERS OAK BROOK PARK DISTRICT TENNIS CENTER REFLECTIVE CEILING AND INSULATION PROJECT

#### INSTRUCTIONS TO BIDDERS

The Oak Brook Park District and Owner are one and the same.

The words "Contractor" and "Bidder" shall mean the party bidding for or entering the Contract for the performance of the Work covered by the written Specifications and Drawings, and his/her legal representatives or authorized agents.

#### A. BID DOCUMENTS

- 1. Bid Documents for this Project will be available for examination and can be obtained from the Oak Brook Park District, 1450 Forest Gate Road, Oak Brook, IL 60523 (the "District"), Monday Friday, 9:00 a.m. 5:00 p.m., or in PDF format at the District's website: http://www.obparks.org/general\_information/bid.asp.
- 2. A mandatory pre-bid meeting will be held for this Project at the Tennis Center, 1300 Forest Gate Road, Oak Brook, IL 60523, on Tuesday, October 25, 2016 at 2:30 p.m. All potential Bidders for this Project must attend the pre-bid meeting. Any bid received by the Oak Brook Park District from a Bidder not in attendance at the pre-bid meeting will be considered non-responsive and returned unopened to the Bidder.

#### B. BID FORM

- 1. Each bid shall be made on the "Bid Form" furnished by the District. The Bid Form shall be executed properly and all writing, including all signatures, shall be with black ink. Failure to use the Bid Form provided could result in rejection of the bid.
- 2. All applicable blank spaces on the "Bid Form" shall be fully completed, including the List of Subcontractors and the Bidder's Reference List, and all amounts shall be in words as well as in figures where applicable.
- 3. The bid shall bear the legal name of the business organization. The signatures shall be in longhand and executed by a duly authorized official of the Bidder's organization and the name of the official and title shall be typed below the signature.
- 4. Erasures, interlineations, corrections, or other changes on the "Bid Form" shall be explained or noted over the signature of the Bidder. No bid submitted with deviations or reservations from the full contract called for will be considered.

- 5. Bidders' prices are to include the delivery of all materials; including plant, equipment, supplies, tools, scaffolding, transportation, insurances, bonds, warranties, and all other items and facilities, and the performance of all labor and services, necessary for the proper completion of the Work except as may be otherwise expressly provided in the Contract Documents. Bids shall not include federal excise tax or state sales tax for materials to be incorporated in, or totally consumed in the prosecution of, the Work. An exemption certificate will be furnished by the Park District upon request of the Bidder.
- 6. Bidder must acknowledge all Addenda received in the spaces provided on the Contractor Bid Form. By submitting a bid, Bidder indicates that all considerations issued by Addendum are incorporated in the bid.
- 7. Attached to the Bid Form will be one or more certifications regarding the Bidder's compliance with applicable laws. Failure of a Bidder to complete/submit a required certification shall be the basis for immediate rejection of that Bidder's bid. The certification of the successful Bidder shall become a part of the Contract with the Park District.
- 8. The bids shall be sealed in an opaque envelope, marked with the name of the Bidder, the date and time of the bid, and addressed as follows:

Sealed Bid: Tennis Center Reflective Ceiling and Insulation Project Oak Brook Park District 1450 Forest Gate Road Oak Brook, IL 60523

- 9. Bid documents shall be delivered or mailed in time for delivery to the foregoing address no later than November 2, 2016 at 11:00 a.m. Oral bids or oral modifications to bids will not be considered. It is the sole responsibility of the Bidder to see that his bid is received in proper time. **No faxed or e-mail bid or modification of a bid will be considered**. The Park District is not responsible for the premature opening of bids not marked as required. Any bid opened prematurely due to the failure of the Bidder to mark the envelope in accordance with these Bid Documents will be considered non-responsive.
- 10. No bid can be withdrawn prior to the opening of the bids unless a written request for any such withdrawal, showing good cause for said withdrawal, is first delivered to the District at the foregoing address prior to commencement of the opening of bids. No Bidder may withdraw a bid after opening of the bids.
- 11. Bids will be publicly opened on the due date.

#### C. REQUIREMENTS OF BIDDERS

Bidders must be able to demonstrate that they: 1) have experience in performing and have successfully performed and are still actively engaged in performing work similar in kind and scope to the Work of the Project; and 2) are able to show that they have adequate laborers and materials to successfully complete the Work as indicated in t the Bid Documents and within the time required by the Bid Documents. The Contractor shall not have been debarred or determined ineligible for public contracts by any governmental agency.

The following information must be attached to the bid proposal. Failure to do so may result in disqualification of the Bidder.

- 1. On a separate sheet, list all construction projects your organization has in progress, giving the name of the project, project description, project address, owner and telephone number, architect and telephone number, contract amount, percent complete, and scheduled completion date.
- 2. On the Bidder's Reference List form provided herein, list at least three (3) construction projects your organization has completed in the past five (5) years, which are comparable in scope, giving the name of the project, project description, project address, owner and telephone number.
- 3. On the List of Subcontractors form provided herein, provide a list of anticipated subcontractors, if any, including their firm names, addresses and telephone numbers. All subcontractors to be used shall be approved by the Owner. If the Contractor subcontracts any part of the Work for this project, the Contractor shall not under any circumstances be relieved of his liabilities and obligations; any subcontractor for this project will be recognized only in the capacity of an employee of the Contractor.
- 4. On a separate sheet, list all administrative proceedings and litigation filed by or against Bidder in the past five (5) years, including the name and case number, name/jurisdiction of the court or administrative agency, and a summary of each claim/case, including current status and if no longer pending, the disposition. The foregoing includes but is not limited to information regarding any proceedings and actions taken by any governmental agency to debar or disqualify the Bidder from bidding on public contracts, including the name of the agency initiating the proceeding/action, the nature of the proceeding/action, the claimed basis for the proceeding/action and the current status or disposition of the proceeding/action.
- 5. On a separate sheet, indicate all instances in which Bidder has been rejected for not being a responsible bidder, giving the name of the project, project description, project address, owner and telephone number, architect and telephone number, contract amount, and an explanation of the circumstances surrounding the rejection.

6. On a separate sheet, provide a list of all contracts to which you were a party and with respect to which you were declared to be in breach of one or more provisions, giving a the type of contract, the project location where applicable, the names and addresses of the parties to the contract, the name of the party declaring the breach, the nature of the claimed breach and current status or resolution of the claim. If a construction contract, also provide the name, address and telephone number of the architect and, if applicable also the construction manager or Owner's representative.

Other required submittals include: Bid Form; Contractor's Compliance and Certification Attachment/ Substance Abuse Prevention Program Certification. Failure of a Bidder to complete/submit these documents shall be the basis for immediate rejection of that Bidder's bid.

#### D. MODIFICATION OF BIDS

Any Bidder may modify his bid by written notice (signed by the Bidder) at any time prior to the scheduled closing time for receipt of bids, provided that such written notice is received by the District prior to the closing time. Modifications of bid submittals sent by facsimile will not be permitted.

#### E. EXAMINATION OF CONTRACT DOCUMENTS AND SITE

Each Bidder shall visit the site(s) of the proposed Work and fully acquaint himself with conditions, as they exist, and shall undertake such additional inquiry and investigation as he shall deem necessary so that he may fully understand the requirements, facilities, possible difficulties and restrictions attending the execution of the Work under the Contract. Bidder shall thoroughly examine and be familiar with all of the Bid Documents including, but not limited to, the Drawings and the written Specifications. Any conflicts or discrepancies found between or among Bid Documents including, but not limited to, the Drawings and written Specifications and the site conditions, or any errors, omissions or ambiguities in the Drawings or written Specifications shall be immediately reported to the Park District and written clarification requested prior to submission of a bid.

The failure or omission of any Bidder to obtain, receive or examine any form, instrument, or information or to visit the Project site(s), and become knowledgeable with respect to conditions there existing, or to seek needed clarification shall in no way relieve any Bidder from any obligations with respect to his bid. By submitting a bid, the Bidder agrees, represents and warrants that he has undertaken such investigation as he deemed necessary, has examined the site(s) and the Bid Documents, has obtained all needed clarifications and where the Bid Documents indicate in any part of the Work, that a given result be produced, that the Bid Documents are adequate and the required result can be produced as indicated in the Specifications and Drawing(s). Once the award has been made, failure to have undertaken and completed the foregoing tasks shall not be cause to alter the original Contract or to request additional compensation.

#### F. ACCEPTANCE OR REJECTION OF BIDS

The Park District may accept the bid of, and award the contract for the Work to, the lowest responsive and responsible Bidder as determined by and in the sole discretion of the Park District.

The Owner reserves the right to (1) reject all bids; (2) reject only certain bids which are non-conforming or non-responsive to the bid requirements; (3) accept only a portion, part or specific items of Work of all and reject others, as the Owner shall in its sole discretion determine to be in its best interest; and/or (4) award the Contract to the responsible Bidder submitting the lowest bid responsive to the bidding requirements. No bid will be accepted from or Contract awarded to any person, firm or corporation that is in arrears or is in default to the Park District upon any debt or contract, or that is a defaulter, as surety or otherwise, upon any obligation to said Park District or that has failed to perform faithfully any previous contract with the Park District.

In the event of a rejection of a portion, part, or certain items of Work of all bids, the bid of each Bidder shall automatically be deemed reduced by the amount of such rejected part or item at the unit price or other cost designated therefore by that Bidder on its submitted Contractor Bid Form. The successful Bidder so selected may not refuse to enter into a Contract with the Owner on the basis that the Owner awarded a Contract for less than all portions or items of the Work specified in the Bid Documents. The Oak Brook Park District Board of Park Commissioners reserves the right to waive any technicalities or irregularities, and to disregard any informality on the bids and bidding, when in its opinion the best interest of the Park District will be served by such actions and in accordance with applicable law.

#### G. SURETY

All bids must be accompanied by a bid bond or bank cashier's check payable to the Oak Brook Park District for ten percent (10%) of the amount of the bid and drawn on a responsive and responsible bank doing business in the United States. All bids not accompanied by a bid security, when required, will be rejected.

The bid security of all except the three (3) lowest responsive and responsible Bidders will be returned after the decision to accept or reject bids by the Oak Brook Park District Board of Park Commissioners. The bid security of the successful Bidder will be returned after acceptance by the Park District of an acceptable Performance Bond, Labor and Materials/Payment Bond and a certificate of insurance naming the Oak Brook Park District as the certificate holder and as additional insured, and the successful Bidder has executed and returned to the Park District the Contract for the Work presented by the Park District.

Prior to beginning Work, the successful Bidder shall furnish a Performance Bond, and Labor and Materials/Payment Bond in the amount of 110% of the Contract Sum, using a form similar to the AIA-A312-2010 form, or its current equivalent, or one acceptable to

Owner, cosigned by a surety company licensed to conduct business in the State of Illinois and with at least an "A" rating and a financial rating of at least "X" in the latest edition of the Best Insurance Guide. Said bond shall guarantee the faithful performance of the Work in accordance with the Contract, the payment of all indebtedness incurred for labor and materials, and guarantee correction of Work. The cost of each bond shall be included in the Contract Sum. The Bidder and all Subcontractors shall name the Park District as an obligee on all bonds. Said bonds shall meet the requirements of the Illinois Public Construction Bond Act, 30 ILCS 550/0.01 *et seq.* and any further amendments thereto. Bidder shall include in its Performance Bond and Labor and Material Payment Bond such language as shall guarantee the faithful performance of the Prevailing Wage Act as required in these Bid Documents.

The Performance Bond and Labor and Material Payment Bond will become a part of the Contract. The failure of the successful Bidder to enter into the Contract and supply the required bonds and evidence of insurance within ten (10) days after the Contract is presented for signature, or within such extended period as the Park District may grant, shall constitute a default, and the Park District may either award the Contract to the next responsible Bidder, or re-advertise for bids. In the event of a default, the Owner need not return the defaulting Bidder's bid surety and may charge against the defaulting Bidder for the full difference between the amount for the bid and the amount for which a Contract for the Work is subsequently executed, irrespective of whether the amount thus due exceeds the amount of the defaulting Bidder's bid surety, provided that the District's retention of the bid guarantee shall not preclude the District from holding the Bidder fully liable for any and all damages which are in excess of said partial liquidated damages, and which shall otherwise be incurred by the District, including reasonable attorneys' fees, arising from the Bidder's failure to enter into said Contract and to deliver the same back to the District within said ten (10) day period.

In addition to the required performance and labor and material payment bonds, the successful Bidder shall furnish a maintenance bond and/or irrevocable letter of credit in the amount of the Contract to guarantee the Work performed under the Contract against defective workmanship and/or defective materials of any nature for a period of not less than twenty-four (24) months from the date of acceptance of the Work, materials or equipment provided. The maintenance bond shall be in a form acceptable to District. A letter of credit furnished in lieu of maintenance bond shall be in a form designated by the District's attorneys. The District reserves the right to waive the maintenance bond in its own interests.

#### H. WITHDRAWAL OF BID

Bidders may withdraw or cancel their bids at any time prior to the advertised bid opening time by signing and submitting a request for said withdrawal. After the bid opening time, no bid shall be withdrawn or canceled for a period of sixty (60) calendar days.

#### I. ACCEPTANCE AND CONTRACT

Owner will award the Contract to the lowest most responsible and responsive Bidder, as determined by Owner. In considering the Bidder's responsibility, the Owner may evaluate, among other factors, the ability of the Bidder to provide experienced labor sufficient in numbers to timely and properly complete the services, conformity with the Specifications, serviceability, quality, and the financial capability of the Bidder, and the performance of the Bidder on other projects.

The Owner shall have the right to accept Alternates in any order or combination, and to determine the low Bidder on the basis of the sum of the Base Bid and Alternates accepted.

Bids will be awarded to one Bidder for the entire Project or to any series of Bidders for an appropriate proportion of the Project. If specified in the Bid Form, awards will be based upon the submitted unit prices.

The acceptance of a bid will be by a Notice of Award, signed by a duly authorized representative of the Park District; no other act by the Park District shall constitute the acceptance of a bid. The acceptance of a bid by the Park District shall bind the successful Bidder to execute and perform the Work of the Contract. The successful Bidder to whom the Contract is awarded by the Park District shall sign and deliver to the Park District for execution by the Park District all required copies of the Contract, along with all required insurance and surety documents within ten (10) days after presentation to him of the Contract for signature. In case the Bidder shall fail or neglect to do so, he will be considered as having abandoned the Contract, and as being in default to the Owner. The Owner may thereupon re-advertise or otherwise award said Contract and forfeit the Bid Security.

The Invitation to Bid, Instructions to Bidders, General Conditions, Supplementary and/or Special Conditions, if any, Drawings, Specifications, Contractor Bid Form, Addenda, if any, Contractors Compliance and Certifications Attachment, and Substance Abuse Certification and the Prevailing Wage Determination and Supersedes Notice comprise the Bid Documents. The Bid Documents, together with the Standard Form of Agreement Between Owner and Contractor AIA Document A101-2007, as modified by the Park District and included in these Bid Documents, and the Performance Bond and Labor Material Payment Bond and proof of insurance comprise the Contract Documents.

#### J. INTERPRETATION OF THE CONTRACT DOCUMENTS

The Park District shall in all cases determine the amount or quantity of the several kinds of Work which are to be paid for under this Contract, and shall decide all questions which may arise relative to the execution of the Contract on the part of the Contractor, and all estimates and decisions shall be final and conclusive. The Park District shall have the right to make alterations in the lines, grades, plans, forms, or dimensions of the Work herein contemplated either before or after the commencement of the Work. If such

alterations diminish the quantity of the Work to be done, they shall not constitute a claim for damage or for anticipated profits on the work dispensed with, or if they increase the amount of Work, such increase shall be paid according to the quantity actually done and at the price or prices stipulated for such Work in the Contract. The Park District reserves the right to approve, an equal to or superior to product or equipment required under the Specifications, or to reject as not being and equal to or superior to the product or equipment required under the Specifications. If the Bidder is in doubt as to the interpretation of any part of the Bid Documents, or finds errors, discrepancies or omissions from any part of the Contract Documents, he must submit a written request for interpretation thereof not later than five (5) days prior to opening of bids to the Park District. Address all communications to Alin Pop, apop@obparks.org. If an error or omission is discovered in the Bid Documents after the bid opening, the Park District reserves the right: i) to determine whether to require the submission of new bids; or ii) if the error or omission is of such a nature that it was reasonably discoverable upon a careful review of the Bid Documents, to award the Contract to the lowest responsive and responsible Bidder as determined by the Park District and to require that Contractor to perform the Work in accordance with an issued correction by the Park District and for the amount bid by the Contractor. Such decisions are final and not subject to recourse. Errors and omissions made by the Bidder cannot be corrected after the bid opening.

#### K. ADDENDA

Any interpretation, correction to, or addition to the Bid Documents will be made by written Addendum and will be delivered by mail or fax to each prime Bidder of record. The written Addenda constitute the only interpretations of the Bid Documents; the Park District accepts no responsibility for any other claimed interpretations or communications

It is the responsibility of each Bidder to verify that he has received all Addenda prior to submitting a bid. It is also the responsibility of each Bidder to verify that all subcontractors and material suppliers whose prices are incorporated in the Bidder's bid are familiar with the Bid Documents in their entirety, including all Addenda issued up to the time of bid opening.

In the event a conflict or omission is discovered in the Bid Documents after the issuing of the last Addendum such that an interpretation cannot be issued by the Park District prior to bidding, the Bidder is directed to estimate on and provide the quantity and quality of material and labor consistent with the overall represented and indicated Work so as to provide all materials, equipment, labor, and services necessary for the completion of the Work in accordance with the Bid Documents.

#### L. SUBSTITUTIONS DURING BIDDING

Unless otherwise indicated, the use of brand names in the Specifications is used for the purpose of establishing a grade or quality. Bidders proposing to use an alternate that is equal to or superior to in every respect to that required by the Specifications must request

approval in writing to the Park District at least seven (7) business days prior to the bid opening and mark the item as 'or approved equal'.

Additionally, Bidders requesting approval for use of an alternate must provide certification by the manufacturer that the substitute proposed is equal to or superior in every respect to that required by the Contract Documents, and that its in-place performance will be equal to or superior to the product or equipment specified in the application indicated. The Bidder, in submitting the request for substitution, waives the right to additional payment or an extension of Contract Time because of the failure of the substitute to perform as represented in the request for substitution.

The Park District may request additional information or documentation necessary for evaluation of the request for substitution. The Park District will notify all Bidders of acceptance of the proposed substitute by means of an Addendum to the Bid Documents. Park District's approval of a substitute during bidding does not relieve the Contractor of the responsibility to submit required shop drawings and to comply with all other requirements of the Contract Documents, including but not limited to proper performance of all components of the Work and suitability for the uses specified.

Bids proposing alternates not previously approved by the Park District will be considered non-responsive and rejected. The Park District reserves the right to determine whether a substituted selection, in its judgment, is equal to or better quality and therefore an acceptable alternate. Such decisions are final and not subject to recourse.

# CONDITIONS OF THE CONTRACT OAK BROOK PARK DISTRICT TENNIS CENTER REFLECTIVE CEILING AND INSULATION PROJECT

#### GENERAL CONDITIONS

The General Conditions are the General Conditions of the Contract for Construction, AIA Document A201-2007, as modified by the Park District and included in these Bid Documents (the "General Conditions").

#### SUPPLEMENTARY CONDITIONS

The General Conditions are hereby amended to include the following:

#### 1. COMMENCEMENT AND COMPLETION DATE

The Work for the Contract shall commence on November 21, 2016, or on such other date as may be agreed upon by the parties. Contractor shall achieve Substantial Completion of the entire Work not later than March 15, 2017 and shall achieve Final Completion of the entire Work not later than April 3, 2017, unless otherwise extended by agreement of the parties pursuant to the General Conditions.

#### 2. <u>USE OF THE SITE</u>

The Contractor shall confine all equipment, the storage of materials and the operations of its workers, to limits indicated by law, ordinances, permits, or directions of the Owner and shall not unreasonably encumber the site with such materials. The site shall not be utilized for the storage of vehicles, materials, equipment, or fixtures not intended for the Work to be performed.

#### 3. <u>COOPERATION WITH UTILITIES</u>

The Contractor shall notify all utility companies, public and private, as necessary in advance of commencing performance of the Work. The responsibility for moving water lines, gas lines, wire lines, service connections, water and gas meter boxes, water and gas valve boxes, light standards, cable ways, signals and all other utility appurtenances which are within the limits of the proposed construction will be assumed by the Contractor, at no additional compensation.

The Contractor shall verify the location of all utilities prior to the start of construction and shall be responsible for the preservation of existing utility installation and the cost of providing precautionary supports, braces, etc. to insure against damage to said utility installation.

The cost to repair and replace any new or existing utilities damaged will be paid for by the Contractor.

It is understood and agreed that the Contractor has considered in its bid all of the permanent and temporary utility appurtenances in their present or relocated positions and that no additional compensation will be allowed for delays, inconvenience, or damage sustained by the Contractor, due to any interference from the said utility appurtenances or the operation of moving them either by the utility company or by the Contractor, or on account of any special construction methods required in performing the Work due to the existence of said appurtenances whether in their present or relocated positions.

#### 4. PROTECTION OF PROPERTY -SAFETY RESPONSIBILITY

In accordance with the Specifications, the Contractor shall protect all existing property and improvements within the Project site and those adjacent to the Owner's property in a manner agreed upon between the Owner and Contractor. The Contractor shall be responsible for the repair cost of any damage created by its operations or the operations of any subcontractors.

Contractor shall comply with State and Federal regulations as outlined in the latest revision of the Federal Construction Safety Standards and with applicable provisions and regulations of Occupation Safety and Health Administration (OSHA), Standards of the William-Steiger Occupational Health and Safety Act of 1970 (revised). The Contractor and Owner shall each be responsible for their respective agents and employees.

The Contractor shall be obligated to indemnify, hold harmless and protect the Owner, its officers, employees and agents, from any actions or suits instituted as a direct or indirect result of any injury or damage consequent upon any failure to use or misuse by the Contractor, its agents and employees and any subcontractor, its agents and employees, of any ladder, support or other mechanical contrivance erected or constructed by any person or any or all kinds of equipment whether or not Owner or furnished by the Owner.

#### 5. <u>INSURANCE</u>

BIDDER'S ATTENTION IS DIRECTED TO THE INSURANCE REQUIREMENTS BELOW. IT IS HIGHLY RECOMMENDED THAT THE BIDDER CONFER WITH ITS INSURANCE CARRIER REGARDING THESE REQUIREMENTS. FAILURE TO MEET THESE REQUIREMENTS IS CAUSE FOR CANCELLATION OF THE CONTRACT.

The successful Bidder shall obtain insurance of the types and in the amounts listed below.

#### a. Commercial General and Umbrella Liability Insurance

The successful Bidder shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less then \$2,000,000 each occurrence. If such CGL insurance contains a general aggregate limit, it shall apply separately to this project/location.

CGL insurance shall be written on Insurance Services Office (ISO) occurrence form CG 00 01 10 93, or a substitute form providing equivalent coverage, and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract (including the tort liability of another assumed in a business contract).

The District, its elected and appointed officials, employees, agents and volunteers shall be included as an additional named insured under the CGL, using ISO additional insured endorsement CG 20 26 or a substitute providing equivalent coverage, and under the commercial umbrella, if any. This insurance shall apply as primary insurance with respect to any other insurance or self-insurance afforded to District. Any insurance or self-insurance maintained by the District shall be deemed excess of such bidder's insurance and shall not contribute with it.

### b. Business Auto and Umbrella Liability Insurance

The successful Bidder shall maintain business auto liability and, if necessary, commercial umbrella liability insurance with a limit of not less than \$1,000,000 each accident. Such insurance shall cover liability arising out of any auto including owned, hired and non-owned autos.

Business auto insurance shall be written on Insurance Services Office (ISO) form CA 00 01, CA 00 05, CA 00 12, CA 00 20, or a substitute form providing equivalent liability coverage. If necessary, the policy shall be endorsed to provide contractual liability coverage equivalent to that provided in the 1990 and later editions of CA 00 01.

#### c. Workers Compensation Insurance

The successful Bidder shall maintain workers compensation and employers liability insurance. The commercial umbrella and/or employers liability limits shall not be less than \$1,000,000 each accident for bodily injury by accident or \$1,000,000 each employee for bodily injury by disease.

Such Bidder waives all rights against District and its officers, officials, employees, volunteers and agents for recovery of damages arising out of or incident to such Bidder's activities.

#### d. General Insurance Provisions

i. <u>Evidence of Insurance:</u> The successful Bidder shall furnish the District with a certificate(s) of insurance and applicable policy

endorsement(s), executed by a duly authorized representative of each insurer, showing compliance with the insurance requirements set forth above

All certificates shall provide for 30 days' written notice to the District prior to the cancellation or material change of any insurance referred to therein. Written notice to the District shall be by certified mail, return receipt requested.

Failure of the District to demand such certificate, endorsement or other evidence of full compliance with these insurance requirements, or failure of the District to identify a coverage deficiency from evidence that is provided, shall not be construed as a waiver of such Bidder's obligation to maintain such insurance.

The District shall have the right, but not the obligation, of prohibiting such Bidder from entering the premises until such certificates or other evidence that insurance has been placed in complete compliance with these requirements is received and approved by District.

Failure to maintain the required insurance may result in termination of the Contract entered by the parties at the District's option.

Such Bidder shall provide certified copies of all insurance policies required above within 10 days of the District's written request for said copies.

- ii. <u>Acceptability of Insurers:</u> All insurance companies shall maintain a rating no less than A-VII from A.M. Best, based on the most recent edition of the A.M. Best's Key Rating Guide. If the Best's rating is less than A-VII or a Best's rating is not obtained, the District has the right to reject insurance written by an insurer it deems unacceptable.
- iii. <u>Deductibles and Self-Insured Retentions</u>: Any deductibles or self-insured retentions must be declared to the District. At the option of the District, the successful Bidder may be asked to eliminate such deductibles or self-insured retentions as respects the District, its officers, officials, employees, volunteers and agents, or such Bidder may be required to procure a bond guaranteeing payment of losses and other related costs, including, but not limited to, investigations, claims administration and defense expenses.

#### e. Subcontractors

Contractor shall cause each subcontractor employed by Contractor to purchase and maintain insurance of the type specified above. When

requested by the Owner, Contractor shall furnish copies of certificates of insurance evidencing coverage for each subcontractor.

### 6. <u>INDEMNIFICATION</u>

To the fullest extent permitted by law, the Contractor shall indemnify and hold harmless the Owner and its officers, officials, employees, volunteers and agents from and against all claims, damages, losses and expenses including but not limited to legal fees (attorney's and paralegals' fees and court costs), arising out of or resulting from the performance of the Contractor's work, provided that any such claim, damage, loss or expense (i) is attributable to bodily injury, sickness, disease or death, or injury to or destruction of tangible property, other than the work itself, including the loss of use resulting there from and (ii) is caused in whole or in part by any wrongful or negligent act or omission of the Contractor, any subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, regardless of whether or not it is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or otherwise reduce any other right or obligation of indemnity which would otherwise exist as to any party or person described in this Paragraph. Contractor shall similarly protect, indemnify and hold and save harmless the Owner, its officers, officials, employees, volunteers and agents against and from any and all claims, costs, causes, actions and expenses including but not limited to legal fees, incurred by reason of Contractor's breach of any of its obligations under, or Contractor's default of, any provision of the Contract.

Nothing contained herein shall be construed as prohibiting the District, its officers, employees or agents from defending, through the selection and use of their own agents, attorneys and experts, any claims, suits, demands, proceedings or actions brought against them. The District's participation in its defense shall not remove the successful Bidder's duty to indemnify, defend and hold the District harmless as set forth herein.

The indemnification required hereunder shall not be limited by reason of the enumeration of insurance coverage herein provided.

The successful Bidder's indemnification of the District shall survive the termination or expiration of the Contract.

#### 7. <u>WARRANTY</u>

A. The Work performed and the materials and equipment installed under this Contract shall be in compliance with the Contract Documents and must be guaranteed by the Contractor and the Surety for a period of twelve (12) months from Final Completion against defective workmanship and material of any nature. On all material or equipment incorporated, the Contractor and its Surety must guarantee that the type, quality, design and performance will fully meet the requirements of the Specifications and Drawings.

- B. A Maintenance Bond or Irrevocable Letter of Credit meeting the requirements set forth in the Instructions to Bidders shall be furnished by the Contractor to guarantee the Work performed, and the materials and equipment provided under the Contract.
- C. The Contractor shall provide the Owner with manufacturer's warranties for all materials and equipment installed under the Contract.

#### **SPECIAL CONDITIONS**

1. The Tennis Center shall remain open to the public for the duration of the Project. As such, Contractor shall maintain the Project site in a manner that ensures safe access Tennis Center amenities by the public, Park District staff and others requiring access to the Tennis Center. Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.

# BID FORM OAK BROOK PARK DISTRICT TENNIS CENTER REFLECTIVE CEILING AND INSULATION PROJECT

(Please complete in ink, and print or type)

TO:	1450 Fore	k Park District est Gate Road k, IL 60523	
FROM:	NAME O	F BIDDER	
	STREET	ADDRESS	
	CITY	STATE	ZIP
	PHONE		

FOR: Tennis Center Reflective Ceiling and Insulation Project

By submission of its bid, the Bidder acknowledges, agrees, represents, declares and warrants:

- A. That he has carefully examined the written Specifications and Drawings and is thoroughly familiar therewith, and that he has visited the site of the proposed Work to arrive at a clear understanding of the conditions under which the Work is to be done, and that he has compared the site with the Drawings and Specifications and has satisfied himself as to all conditions affecting the execution of the Work;
- B. That all modifications have been submitted with this bid;
- C. That he has checked carefully the bid figures and understands that he shall be responsible for any errors or omissions based on these Specifications and alternates as submitted on the Bid Proposal Form;
- D. That it is understood and agreed that the Oak Brook Park District reserves the right to accept or reject any or all bids, or to combine or separate any section or work, and to waive any technicalities:
- E. To hold the bid open for sixty (60) days subsequent to the date of the bid opening;
- F. To enter into and execute a Contract with the Owner within ten (10) days after the date of the Notice of Award, if awarded on the basis of this bid, and in connection therewith to:

- (a) Furnish all bonds and insurance required by the Contract Documents;
- (b) Accomplish the Work in accordance with the Contract Documents; and
- (c) Complete the Work within the time requirements as set forth in the Bid Documents.
- G. That if this bid is accepted, the Bidder is to provide all of the necessary equipment, tools, apparatus, labor, and other means of construction, and to do all of the Work and to furnish all of the materials specified in the Bid Documents in the manner and at the time therein prescribed, and in accordance with the requirements set forth;
- H. To commence Work as specified in the Instructions to Bidders, and to prosecute the Work in such a manner, and with sufficient materials, equipment and labor as will ensure its completion within reasonable time, it being understood and agreed that the completion within such reasonable time is an essential part of this Contract;
- I. That any and all prices stated in the proposal include all costs of labor, materials, equipment, insurance, bonds, overhead and profit, and any and all other costs normal to doing business.

The undersigned Bidder agrees to perform the Work for the following lump sum price:

Base Bid	Amount
Reflective Ceiling and Insulation Project	\$

The undersigned Bidder hereby acknowledges the receipt of the following addenda (if any) distributed by the Park District.

Addendum No.	 Date:
Addendum No.	 Date:

The Work for the Contract shall commence on November 21, 2016, or on such other date as may be agreed upon by the parties. Contractor shall achieve Substantial Completion of the entire Work not later than March 15, 2017 and shall achieve Final Completion of the entire Work not later than April 3, 2017, unless otherwise extended by agreement of the parties pursuant to the General Conditions.

The undersigned Bidder agrees that if this bid is accepted by the Park District, it will perform all Work in accordance with the requirements of the Contract.

DATED THIS DAY OF	, 2016.		
	(a) Individual	( )	
Full Name of Bidder (Print)	(b) Partnership	()	
	(c) Corporation	( )	

Name and Title of Authorized Agent if Corporation or Partnership (Print):	
Full Name and Title of Bidder (Signature)	
Street Address	
City/State/Zip	
Phone	
Email	

#### LIST OF SUBCONTRACTORS

Bidder submits a list of subcontractors for each trade relative to the Work to be performed under the Contract with the District, and agrees that if selected the successful Contractor, the Bidder will promptly confer with the District's agents on the question of which subcontractors the Bidder proposes to use, including submission of their qualifications. It is agreed that the District may substitute for any proposed subcontractor, another subcontractor for the trade against whose standing and ability the Bidder makes no objection in writing, and the Bidder will use all such finally selected subcontractors at the amount named in their respective subcontracts, and be in every way responsible for them and their work as if they had been originally named in the Bidder's bid, the unit, total and alternate Contract prices being adjusted to confirm thereto.

Subcontractor Name & Address	Classification of Work	Amount of Subcontract
1.		
2.		
3.		
4.		

### **BIDDER'S REFERENCE LIST**

Each Bidder must list the name, address, phone number and project name for at least three (3) projects performed for governmental entities of similar scope and complexity as the Tennis Center Reflective Ceiling and Insulation Project in the past five (5) years. Bidder may include, as a separate attachment, additional information or references on projects completed.

Contact Person		
Phone Number	E-Mail	
Description of Work performed		Project Value
Name of Park District, School District	, or Municipality	
Contact Person		
Phone Number	E-Mail	
Description of Work performed		Project Value
Name of Park District, School District	, Municipality	
Contact Person		
Phone Number	E-Mail	

#### CONTRACTOR COMPLIANCE AND CERTIFICATIONS ATTACHMENT

Note: The following certifications form an integral part of the Agreement between the Owner and Contractor. Breach by Contractor of any of the certifications may result in immediate termination of the Contractor's services by Owner.

THE UNDERSIGNED CONTRACTOR HEREBY ACKNOWLEDGES, CERTIFIES, AFFIRMS AND AGREES AS FOLLOWS:

- A. Contractor has carefully read and understands the contents, purpose and legal effect of this document as stated above and hereafter in this document. The certifications contained herein are true, complete and correct in all respects.
- B. Contractor shall abide by and comply with, and in contracts which it has with all persons providing any of the services or Work on this Project on its behalf shall require compliance with, all applicable Federal, State and local laws and rules and regulations including without limitation those relating to 1) fair employment practices, affirmative action and prohibiting discrimination in employment; 2) workers' compensation; 3) workplace safety; 4) wages and claims of laborers, mechanics and other workers, agents, or servants in any manner employed in connection with contracts involving public funds or the development or construction of public works, buildings or facilities; and 5) steel products procurement.
- C. All contracts for this Project are subject to the provisions of the Illinois Prevailing Wage Act (820 ILCS 130/0.01 et seq.), providing for the payment of the prevailing rate of wage to all laborers, workmen and mechanics engaged in the Work. Contractor shall pay prevailing rates of wages in accordance with the wage determination included with the Contract Documents and any subsequent determinations issued by the Illinois Department of Labor which shall supersede the determination included in the Contract Documents, all in accordance with applicable law. Contractor is responsible for determining the applicable prevailing wage rates at the time of bid submission and at the time of performance of the Work. Failure of Contractor to make such determination shall not relieve it of its obligations in accordance with the Contract Documents. Contractor shall also comply with all other requirements of the Act including without limitation those pertaining to inclusion of required language in subcontracts, job site posting, maintenance and submission of certified payroll records and inspection of records. Contractor is not barred from entering into public contracts under Section 11a of the Illinois Prevailing Wage Act due to its having been found to have disregarded its obligations under the Act.
- D. To the best of Contractor's knowledge, no officer or employee of Contractor has been convicted of bribery or attempting to bribe an officer or employee of the State of Illinois, or any unit of local government, nor has any officer or employee made an admission of guilt of such conduct which is a matter of record.

- E. Contractor is not barred from bidding on or entering into public contracts due to having been convicted of bid-rigging or bid rotating under paragraphs 33E-3 or 33E-4 of the Illinois Criminal Code. Contractor also certifies that no officers or employees of the Contractor have been so convicted and that Contractor is not the successor company or a new company created by the officers or owners of one so convicted. Contractor further certifies that any such conviction occurring after the date of this certification will be reported to the Owner, immediately in writing, if it occurs during the bidding process, or otherwise prior to entering into the Contract therewith.
- F. Pursuant to the Illinois Human Rights Act (775 ILCS 5/2-105), Contractor has a written sexual harassment policy that includes, at a minimum, the following information: (i) a statement on the illegality of sexual harassment; (ii) the definition of sexual harassment under State law; (iii) a description of sexual harassment utilizing examples; (iv) the Contractor's internal complaint process including penalties; (v) the legal recourse, investigative and complaint process available through the Illinois Department of Human Rights and the Human Rights Commission and directions on how to contact both; and (vi) protection against retaliation as provided by Section 6-101 of the Illinois Human Rights Act. Contractor further certifies that such policy shall remain in full force and effect. A copy of the policy shall be provided to the Illinois Department of Human Rights upon request.
- G. (i) Contractor's bid proposal was made without any connection or common interest in the profits anticipated to be derived from the Contract by Contractor with any other persons submitting any bid or proposal for the Contract; (ii) the Contract terms are in all respects fair and the Contract will be entered into by Contractor without collusion or fraud; (iii) no official, officer or employee of the Owner has any direct or indirect financial interest in Contractor's bid proposal or in Contractor, (iv) the Contractor has not directly or indirectly provided, and shall not directly or indirectly provide, funds or other consideration to any person or entity (including, but not limited to, the Owner and the Owner's employees and agents), to procure improperly special or unusual treatment with respect to this Agreement or for the purpose of otherwise improperly influencing the relationship between the Owner and the Contractor. Additionally, the Contractor shall cause all of its officers, directors, employees, (as the case may be) to comply with the restrictions contained in the preceding sentence.
- H. Contractor knows and understands the Equal Employment Opportunity Clause administrated by the Illinois Department of Human Rights, which is incorporated herein by this reference, and agrees to comply with the provisions thereof. Contractor further certifies that Contractor is an "equal opportunity employer" as defined by Section 2000 (e) of Chapter 21, Title 42 of the United States Code Annotated and Executive Orders #11246 and #11375 as amended, which are incorporated herein by this reference.
- I. Neither Contractor nor any substantially owned affiliated company is participating or shall participate in an international boycott in violation of the provisions of the U.S. Export Administration Act of 1979 or the regulations of the U.S. Department of Commerce promulgated under that Act.

- J. Contractor is not barred from contracting with the Owner because of any delinquency in the payment of any tax administrated by the Illinois Department of Revenue, unless it is being contested. Contractor further certifies that it understands that making a false statement regarding delinquency in taxes is a Class A misdemeanor and, in addition, voids the Contract and allows the Owner, a municipal entity, to recover in a civil action all amounts paid to the Contractor.
- K. If Contractor has 25 or more employees at the time of letting of the Contract, Contractor knows, understands and acknowledges its obligations under the Illinois Drug Free Workplace Act (30 ILCS 580/1 et seq.) and certifies that it will provide a drug-free workplace by taking the actions required under, and otherwise implementing on a continuing basis, Section 3 of the Drug Free Workplace Act. Contractor further certifies that it has not been debarred and is not ineligible for award of this Contract as the result of a violation of the Illinois Drug Free Workplace Act.
- L. Contractor knows, understands and acknowledges its obligations under the Substance Abuse Prevention on Public Works Act, 820 ILCS 265/1 *et seq.* A true and complete copy of Contractor's Substance Abuse Prevention Program Certification is attached to and made a part of this Contractor Compliance and Certification Attachment.
- M. The Contractor shall comply with the requirements and provisions of the Freedom of Information Act (5 ILCS 140/1 *et. seq.*) and, upon request of the Oak Brook Park District's designated Freedom of Information Act Officer (FOIA Officer), Contractor shall within two (2) business days of said request, turn over to the FOIA Officer any record in the possession of the Contractor that is deemed a public record under FOIA.

## 

#### SUBSTANCE ABUSE PREVENTION PROGRAM CERTIFICATION

The Substance Abuse Prevention on Public Works Projects Act, 820 ILCS 265/1 et seq., ("Act") prohibits any employee of the Contractor or any Subcontractor on a public works project to use, possess or be under the influence of a drug or alcohol, as those terms are defined in the Act, while performing work on the project. The Contractor/Subcontractor [circle one], by its undersigned representative, hereby certifies and represents to the Oak Brook Park District that [Contractor/Subcontractor must complete either Part A or Part B below]:

covered substancertifica Public V	The Contractor/Subcontractor [circle one] has in place for all of its employees not d by a collective bargaining agreement that deals with the subject of the Act a written ce abuse prevention program, a true and correct copy of which is attached to this ation, which meets or exceeds the requirements of the Substance Abuse Prevention on Works Act, 820 ILCS 265/1 et seq. [Contractor/Subcontractor must attach a copy of
its subs	stance abuse prevention program to this Certification.]
Ī	Name of Contractor/Subcontractor (print or type)
Ī	Name and Title of Authorized Representative (print or type)
	Dated:
;	Signature of Authorized Representative
agreeme	The Contractor/Subcontractor <b>[circle one]</b> has one or more collective bargaining ents in effect for all of its employees that deal with the subject matter of the Substance Prevention on Public Works Projects Act, 820 ILCS 265/1 <i>et seq</i> .
]	Name of Contractor/Subcontractor (print or type)
j	Name and Title of Authorized Representative (print or type)
	Dated:

Signature of Authorized Representative

## IMPORTANT NOTICE OF RESPONSIBILITY FOR PERIODIC REVISIONS TO PREVAILING WAGE RATES

Revisions of the following Prevailing Wage Rates are made periodically by the Illinois Department of Labor. These may be accessed by computer athttp://www.illinois.gov/idol/Laws-Rules/CONMED/Pages/Rates.aspx. required by the Prevailing Wage Act, any and all such revisions supersede the Park District's June determination. Bidders and contractors performing work on this Project are responsible for determining the applicable prevailing wage rates at the time of bid submission and performance of the Work. bidder/contractor to make such determination shall not relieve it of its obligations in accordance with the Contract Documents In consideration for the award to it of the contract for this Project, the contractor agrees that the foregoing notice satisfies any obligation of the public body in charge of this Project to notify the contractor of periodic changes in the prevailing wage rates and the contractor agrees to assume and be solely responsible for, as a material obligation of the contractor under the contract, the obligation to determine periodic revisions of the prevailing wage rates, to notify its subcontractors of such revisions, to post such revisions as required for the posting of wage rates under the Act, and to pay and require its subcontractors to pay wages in accordance with such revised rates.

## Standard Form of Agreement Between Owner and Contractor

AIA Document A 101-207



## Standard Form of Agreement Between Owner and Contractor where the basis of payment is a Stipulated Sum

AGREEMENT made as of the day of in the year Two Thousand Sixteen (In words, indicate day, month and year.)

#### BETWEEN the Owner:

(Name, legal status, address and other information)

Oak Brook Park District 1450 Forest Gate Road Oak Brook, IL 60523

#### and the Contractor:

(Name, legal status, address and other information)

for the following Project: (Name, location and detailed description) Tennis Center Reflective Ceiling and Insulation Project Oak Brook Park District Tennis Center 1300 Forest Gate Road Oak Brook, IL 60523

The Project includes the installation of reflective ceiling and insulation at the Tennis Center and all other and incidental and collateral work necessary to properly complete the Project as indicated in the Contract Documents.

#### The Architect

(Name, legal status, address and other information)

NA. No Architect is being used for this Project. For purposes of this Project, "Architect" shall mean "Owner."

The Owner and Contractor agree as follows.

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

AlA Document A201™-2007. General Conditions of the Contract for Construction, is adopted in this document by reference. Do not use with other general conditions unless this document is modified.

#### **TABLE OF ARTICLES**

- 1 THE CONTRACT DOCUMENTS
- 2 THE WORK OF THIS CONTRACT
- 3 DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION
- 4 CONTRACT SUM
- 5 PAYMENTS
- 6 DISPUTE RESOLUTION
- 7 TERMINATION OR SUSPENSION
- 8 MISCELLANEOUS PROVISIONS
- 9 ENUMERATION OF CONTRACT DOCUMENTS
- 10 INSURANCE AND BONDS

#### ARTICLE 1 THE CONTRACT DOCUMENTS

The Contract Documents consist of this Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of this Agreement, other documents listed in this Agreement and Modifications issued after execution of this Agreement, all of which form the Contract, and are as fully a part of the Contract as if attached to this Agreement or repeated herein. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. An enumeration of the Contract Documents, other than a Modification, appears in Article 9.

#### ARTICLE 2 THE WORK OF THIS CONTRACT

The Contractor shall fully execute the Work indicated in the Contract Documents, except as specifically stated in the Contract Documents to be the responsibility of others.

### ARTICLE 3 DATE OF COMMENCEMENT AND SUBSTANTIAL COMPLETION

§ 3.1 The date of commencement of the Work shall be the date of this Agreement unless a different date is stated below or provision is made for the date to be fixed in a notice to proceed issued by the Owner. (Insert the date of commencement if it differs from the date of this Agreement or, if applicable, state that the date will be fixed in a notice to proceed.)

The date of commencement of the Work shall be November 21, 2016 (hereinafter referred to as the "Commencement Date"). Prior to commencing the Work, the Contractor shall have obtained and provided to the Owner acceptable evidence of all licenses, permits, bonds and insurance indicated as being the Contractor's responsibility under the Contract Documents. Delay in the commencement of the Work attributable to the failure of the Contractor to have obtained and provided such evidence to the Owner shall not result in an extension of the date scheduled for Substantial Completion as provided in Section 3.3 below, or in any milestone date previously agreed to by the Parties in any Project Schedule.

If, prior to the commencement of the Work, the Owner requires time to file mortgages and other security interests, the Owner's time requirement shall be as follows:

Not applicable to this Contract.

Init.

1

§ 3.2 The Contract Time shall be measured from the Commencement Date.

AIA Document A101™ – 2007. Copyright © 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1967, 1974, 1977, 1987, 1991, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:14:24 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

§ 3.3 The Contractor shall achieve Substantial Completion of the entire Work not later than (Paragraphs deleted)

March 15, 2017 and shall achieve Final Completion of the entire Work not later than April 3, 2017, subject to adjustments of this Contract Time authorized by Change Order as provided in the Contract Documents. The Owner and Contractor agree that the amount of time given to the Contractor under the Contract to achieve Substantial Completion is a reasonable amount of time considering the requirements of the Contract Documents.

(Insert provisions, if any, for liquidated damages relating to failure to achieve Substantial Completion on time or for bonus payments for early completion of the Work.)

#### ARTICLE 4 CONTRACT SUM

§ 4.1 The Owner shall pay the Contractor the Contract Sum in current funds for the Contractor's performance of the Contract. The Contract Sum shall be\_\_\_\_\_\_ Dollars (\$ ), subject to additions and deductions as provided in the Contract Documents.

§ 4.2 The Contract Sum is based upon and includes the following alternates, if any, which are described in the Contract Documents and are hereby accepted by the Owner:

(State the numbers or other identification of accepted alternates. If the bidding or proposal documents permit the Owner to accept other alternates subsequent to the execution of this Agreement, attach a schedule of such other alternates showing the amount for each and the date when that amount expires.)

NA

§ 4.3 Unit prices are pursuant to Contractor's Proposal, as modified by this Agreement, attached to and incorporated into this Agreement by reference. :

(Identify and state the unit price; state quantity limitations, if any, to which the unit price will be applicable.)

**Item** 

**Units and Limitations** 

Price Per Unit (\$0.00)

§ 4.4 Allowances included in the Contract Sum, if any: (Identify allowance and state exclusions, if any, from the allowance price.)

Item Price

§ 4.5 Adjustments to the Contract Sum for changes in the Work (other than for changes in the Work involving items for which unit prices were provided as set forth in Section 4.3, above) shall be made in accordance with Article 7 of the General Conditions of the Contract, provided that in the case of an increase in the Contract Sum attributable to a change in the Work, "reasonable overhead and profit" for purposes of Section 7.3.7 thereof shall mean: 1. Five percent (5%) of the cost of the change in the Work involved if performed by the Contractor not involving Subcontractors, or 2. Five percent (5%) of the cost of the change in the Work involved performed by Subcontractors, plus two percent of the cost of the change in the Work for the Contractor's supervision of the work performed by the Subcontractors.

When both additions and credits covering related Work are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change. No overhead and profit shall be paid to the Contractor for an increase in the cost of the surety bond or insurance premiums resulting from a change in the Work. In the event of an increase in the Contract Sum resulting from an additive change order, the actual amount of any increase in the cost of Contractor's surety bond shall be added to the amount of the change order. Similarly in the event of a decrease in the Contract Sum resulting from a deductive change order, the amount of any decrease in the cost of Contractor's surety bond shall be added to the amount shown in the deductive change order, by which the Contract Sum is to be reduced.

§ 4.6 Overtime, if and when specifically authorized in writing in advance by the Owner shall be paid by the Owner on the basis of premium payment if any, plus the cost of insurance and taxes based on the premium payment period No overhead or profit may be charged for overtime. The Contractor shall not be entitled to any payment for overtime necessitated by the failure of the Contractor to perform the Work in accordance with the Contract Documents including without limitation to the Contractor's failure to prosecute the Work diligently and on an uninterrupted basis and with a sufficient workforce so as to achieve completion of the Work within the time and in the manner contemplated by the Contract Documents, or otherwise due to the fault of the Contractor In such instances if the Owner requires the Contractor to perform Work on an overtime basis, all costs for and associated with such overtime shall be borne by the Contractor

#### ARTICLE 5 **PAYMENTS**

#### § 5.1 PROGRESS PAYMENTS

- § 5.1.1 Based upon Applications for Payment properly completed and accompanied by all supporting documentation and other submittals required by the Contract Documents submitted to the Architect by the Contractor and Certificates for Payment issued by the Architect, and agreed to by Owner and not subsequently nullified by the Architect in accordance with the Contract Documents, the Owner shall make progress payments on account of the Contract Sum to the Contractor as provided below and elsewhere in the Contract Documents.
- § 5.1.2 The period covered by each Application for Payment shall be one calendar month ending on the last day of the month, or as follows:
- § 5.1.3 Provided that an Application for Payment, which is in proper form and accompanied by required supporting documents and submittals, is received by the Architect not later than the 10th day of a month, certified for payment by the Owner and not subsequently nullified by the Owner in accordance with the Contract Documents, the Owner shall make payment of the certified amount to the Contractor not later than the 15th day of the following month. If an Application for Payment is received by the Architect after the application date fixed above, payment shall be made by the Owner not later than forty-five (45) days after the Architect receives the Application for Payment in proper form and accompanied by required supporting documents and submittals and certifies payment to the Owner. Contractor is solely responsible for any delays in payment due in whole or in part to Contractor's failure to submit its payment application timely, in proper form and accompanied by all supporting documents and submittals required under the Contract.
- § 5.1.4 Each Application for Payment shall be based on the most recent schedule of values submitted by the Contractor in accordance with the Contract Documents. The schedule of values shall allocate the entire Contract Sum among the various portions of the Work. The schedule of values shall be prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment.
- § 5.1.5 Applications for Payment shall show the percentage of completion of each portion of the Work as of the end of the period covered by the Application for Payment.
- § 5.1.6 Subject to other provisions of the Contract Documents, the amount of each progress payment shall be computed as follows:
  - Take that portion of the Contract Sum properly allocable to completed Work as determined by multiplying the percentage completion of each portion of the Work by the share of the Contract Sum allocated to that portion of the Work in the schedule of values, less retainage of ten percent (10 %). Pending final determination of cost to the Owner of changes in the Work, amounts not in dispute shall be included as provided in Section 7.3.9 of AIA Document A201™-2007, General Conditions of the Contract for Construction;
  - Add that portion of the Contract Sum properly allocable to materials and equipment delivered and suitably stored at the site for subsequent incorporation in the completed construction (or, if approved in advance by the Owner, suitably stored off the site at a location agreed upon in writing), less retainage of ten percent (10 %);
  - .3 Subtract the aggregate of previous payments made by the Owner; and

I

- Subtract amounts, if any, for which the Architect has withheld or nullified a Certificate for Payment as provided in Section 9.5 of AIA Document A201-2007.
- § 5.1.7 The progress payment amount determined in accordance with Section 5.1.6 may be further modified (Paragraphs deleted) and the retainage may be reduced if and as provided in Paragraph 9.3 of the General Conditions.
- § 5.1.8 Reduction or limitation of retainage, if any, shall be as follows:

(If it is intended, prior to Substantial Completion of the entire Work, to reduce or limit the retainage resulting from the percentages inserted in Sections 5.1.6.1 and 5.1.6.2 above, and this is not explained elsewhere in the Contract Documents, insert here provisions for such reduction or limitation.)

§ 5.1.9 Except with the Owner's prior approval, the Contractor shall not make advance payments to suppliers for materials or equipment which have not been delivered and stored at the site.

#### § 5.2 FINAL PAYMENT

- § 5.2.1 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be made by the Owner to the Contractor when
  - .1 the Contractor has fully performed the Contract in accordance with the Contract Documents; provided that Owner shall have no obligation to make final payment if the Contractor is required to correct Work as provided in Section 12.2.2 of AIA Document A201-2007 or as otherwise specified by the Contract Documents, or to satisfy other requirements, if any, which extend beyond final payment; and
  - .2 a final Certificate for Payment has been issued by the Architect and the Owner has approved payment.
- § 5.2.2 The Owner's final payment to the Contractor shall be made no later than 45days after the conditions set forth in Section 5.2.1, above, have been met.

#### § 5.3 WAIVER PROCEDURE/FORMAT

The first Application for Payment shall be accompanied by the Contractor's Partial Waiver of Lien to date for the full amount of the payment. Each subsequent monthly payment application shall be accompanied by the Contractor's Partial Waiver of Lien and the Partial Waivers by of subcontractors and suppliers who were included in the immediately preceding payment application to the extent of that payment. The Application for Final Payment shall be accompanied by Final Waivers of lien from the Contractor, subcontractors, and suppliers who have not previously furnished such Final Waivers. Final Waivers shall be for the full amount of the Contract and the Final Waiver of a Subcontractor shall be for the full amount of its Subcontract. All applications for payment shall be accompanied by affidavits from the Contractor, in triplicate, containing such information and in such form to comply with the Illinois Mechanics Lien Act (770 ILCS 60/001 et seq.) and showing in detail the sources of all labor and materials used and contracted to be used on the Project, including names and addresses of subcontractors and materials suppliers; amounts paid and remaining to be paid to each; together with all other documents as shall be necessary, in the sole judgment of the Owner, to waive all claims of liens to date and comply with all applicable state and local laws.

- (i) All waivers (partial and final) shall include language as applicable indicating either that:
  - 1. All materials were taken from fully paid stock and delivered to job site in our own vehicles and all labor has been fully paid in accordance with prevailing wage laws; or
  - 2. Materials were provided by the following suppliers from whom waivers of lien are attached and all labor has been fully paid in accordance with prevailing wage laws.

#### ARTICLE 6 DISPUTE RESOLUTION

§ 6.1

(Paragraphs deleted)

The Parties shall make claims and resolve disputes as provided in Article 15 of the General Conditions and elsewhere in the Contract Documents.

(Paragraphs deleted)

#### TERMINATION OR SUSPENSION ARTICLE 7

§ 7.1 The Contract may be terminated by the Owner or the Contractor as provided in Article 14 of AIA Document A201-2007.

§ 7.2 The Work may be suspended by the Owner as provided in Article 14 of AIA Document A201-2007.

#### **MISCELLANEOUS PROVISIONS ARTICLE 8**

§ 8.1 Where reference is made in this Agreement to a provision of AIA Document A201-2007 or another Contract Document, the reference refers to that provision as amended, superseded or supplemented by other provisions of the Contract Documents.

§ 8.2 Payments due and unpaid under the Contract shall bear interest from the date payment is due (Paragraphs deleted) and at the rate provided under the Illinois Local Government Prompt Payment Act.

§ 8.3 The Owner's representative: (Name, address and other information)

Alin Pop Oak Brook Park District 1450 Forest Gate Road Oak Brook, IL 60523 T: 630-645-9510 Email: apop@obparks.org

§ 8.4 The Contractor's representative: (Name, address and other information)

§ 8.5 Neither the Owner's nor the Contractor's representative shall be changed without ten days written notice to the other party.

§ 8.6 Other provisions:

§ 8.6.1 Not less than the prevailing rate of wages as determined by the Illinois Department of Labor shall be paid to all laborers, workers and mechanics performing the Work. Contractor's bonds shall include a provision as will guarantee the faithful performance of this prevailing wage clause as herein provided and as provided in the General Contractor shall comply with all other requirements of the Prevailing Wage Act.

(1181773143)

Init.

- § 8.6.2 The Contractor shall be responsible for the supply and maintenance of any and all temporary equipment, utilities and facilities necessary to properly and safely complete and protect the Work, including without limitation those required by winter conditions. The Contractor shall provide and erect barricades and other safeguards adequate to warn of danger at the site and to protect persons and property from injury resulting from the Work.
- § 8.6.3 The Contractor shall limit materials and equipment storage to the immediate area of Work and such other areas as the Owner may designate. The Contractor shall promptly remove and properly dispose all construction material, trash, garbage and other debris off site.
- § 8.6.4 Except as otherwise specifically provided in the Contract Documents, if and to the extent of any inconsistency, ambiguity, conflict, discrepancy or error in the Contract Document, and otherwise in interpreting the Contract Documents, the Parties shall give precedence to the Contract Documents in the following order of priority:
  - (i) Modifications.
  - (ii) This Agreement
  - (iii) Supplementary and Special Conditions.
  - (iv) General Conditions
  - (v) Construction Drawings
- §8.6.5 The rights and remedies of the Owner stated in the Contract Documents shall be in addition to and not in limitation of any other rights of the Owner granted at law or in equity.

#### ARTICLE 9 ENUMERATION OF CONTRACT DOCUMENTS

- § 9.1 The Contract Documents, except for Modifications issued after execution of this Agreement, are enumerated in the sections below.
- § 9.1.1 The Agreement is this executed AIA Document A101–2007, Standard Form of Agreement Between Owner and Contractor, as modified by Owner.
- § 9.1.2 The General Conditions are AIA Document A201–2007, General Conditions of the Contract for Construction, as modified by Owner.
- § 9.1.3 The Supplementary, Special and other Conditions of the Contract are those included in the Project Manual dated October 12, 2016.

(Table deleted)

§ 9.1.4 The Specifications:

(Either list the Specifications here or refer to an exhibit attached to this Agreement.)
The Specifications are those included in the Project Manual dated October 12, 2016.

(Table deleted)

§ 9.1.5 The Drawings:

(Either list the Drawings here or refer to an exhibit attached to this Agreement.)
The Drawings are those included in the Project Manual dated October 12, 2016.

Number Title Date (Row deleted)
§ 9.1.6 The Addenda, if any:

Number Date Pages

Portions of Addenda relating to bidding requirements are not part of the Contract Documents unless the bidding requirements are also enumerated in this Article 9.

§ 9.1.7 Additional documents, if any, forming part of the Contract Documents:

Init.

**User Notes:** 

- 1. All other documents contained in the Project Manual dated October 12, 2016.
- 2. Certificate of Insurance and endorsements attached to and incorporated in this Agreement by this reference.
  - 3. Performance Bond, Labor and Material Payment Bond, attached to and incorporated in this Agreement by this reference.
  - 4. Prevailing Wage Determination and supersedes notice attached to and incorporated in this Agreement by this reference.
- 5. Contractor's Proposal, as modified by the Contract Documents, attached to and incorporated in this Agreement by this reference.
- 6. Contractor's Compliance and Certifications Attachment and Substance Abuse Prevention

  Program Certification, attached to and incorporated in this Agreement by this reference.

#### ARTICLE 10 INSURANCE AND BONDS

The Contractor shall purchase and maintain insurance and provide bonds as set forth in Article 11 of AIA Document A201–2007.

(Table deleted)(Paragraph deleted)

OAK DROOK BARK DICTRICE

This Agreement entered into as of the day and year first written above.

OAK BROOK PARK DISTRICT	
OWNER (Signature)	CONTRACTOR (Signature)
(Printed name and title)	(Printed name and title)

# Additions and Deletions Report for

AIA® Document A101™ - 2007

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AlA document in order to complete it, as well as any text the author may have added to or deleted from the original AlA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AlA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 11:14:24 on 10/04/2016.

#### PAGE 1

AGREEMENT made as of the day of in the year Two Thousand Sixteen

\*\*\*

Oak Brook Park District 1450 Forest Gate Road Oak Brook, IL 60523

...

Tennis Center Reflective Ceiling and Insulation Project
Oak Brook Park District Tennis Center
1300 Forest Gate Road
Oak Brook, IL 60523

The Project includes the installation of reflective ceiling and insulation at the Tennis Center and all other and incidental and collateral work necessary to properly complete the Project as indicated in the Contract Documents.

- • •

The Architect: Architect

...

NA. No Architect is being used for this Project. For purposes of this Project, "Architect" shall mean "Owner."

#### PAGE 2

The Contractor shall fully execute the Work <u>described indicated</u> in the Contract Documents, except as specifically <u>indicated stated</u> in the Contract Documents to be the responsibility of others.

••

The date of commencement of the Work shall be November 21, 2016 (hereinafter referred to as the "Commencement Date"). Prior to commencing the Work, the Contractor shall have obtained and provided to the Owner acceptable evidence of all licenses, permits, bonds and insurance indicated as being the Contractor's responsibility under the Contract Documents. Delay in the commencement of the Work attributable to the failure of the Contractor to have obtained and provided such evidence to the Owner shall not result in an extension of the date scheduled for Substantial Completion as provided in Section 3.3 below, or in any milestone date previously agreed to by the Parties in any Project Schedule.

Not applicable to this Contract.

- § 3.2 The Contract Time shall be measured from the date of commencement. Commencement Date.
- § 3.3 The Contractor shall achieve Substantial Completion of the entire Work not later than ( ) days from the date of commencement, or as follows:

(Insert number of calendar days. Alternatively, a calendar date may be used when coordinated with the date of commencement. If appropriate, insert requirements for earlier Substantial Completion of certain portions of the Work.)

Portion of Work

Substantial Completion Date

, subject March 15, 2017 and shall achieve Final Completion of the entire Work not later than April 3, 2017, subject to adjustments of this Contract Time authorized by Change Order as provided in the Contract Documents. The Owner and Contractor agree that the amount of time given to the Contractor under the Contract to achieve Substantial Completion is a reasonable amount of time considering the requirements of the Contract Documents.

#### PAGE 3

- § 4.1 The Owner shall pay the Contractor the Contract Sum in current funds for the Contractor's performance of the Contract. The Contract Sum shall be be Dollars (\$ ), subject to additions and deductions as provided in the Contract Documents.
- § 4.2 The Contract Sum is based upon and includes the following alternates, if any, which are described in the Contract Documents and are hereby accepted by the Owner:

NA

§ 4.3 Unit prices, if any: prices are pursuant to Contractor's Proposal, as modified by this Agreement, attached to and incorporated into this Agreement by reference. :

...

§ 4.5 Adjustments to the Contract Sum for changes in the Work (other than for changes in the Work involving items for which unit prices were provided as set forth in Section 4.3, above) shall be made in accordance with Article 7 of the General Conditions of the Contract, provided that in the case of an increase in the Contract Sum attributable to a change in the Work, "reasonable overhead and profit" for purposes of Section 7.3.7 thereof shall mean. 1. Five percent (5%) of the cost of the change in the Work involved if performed by the Contractor not involving Subcontractors, or 2. Five percent (5%) of the cost of the change in the Work involved performed by Subcontractors, plus two percent of the cost of the change in the Work for the Contractor's supervision of the work performed by the Subcontractors.

When both additions and credits covering related Work are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change. No overhead and profit shall be paid to the Contractor for an increase in the cost of the surety bond or insurance premiums resulting from a change in the Work. In the event of an increase in the Contract Sum resulting from an additive change order, the actual amount of any increase in the cost of Contractor's surety bond shall be added to the amount of the change order. Similarly in the event of a decrease in the Contract Sum resulting from a deductive change order, the amount of any decrease in the cost of Contractor's surety bond shall be added to the amount shown in the deductive change order, by which the Contract Sum is to be reduced.

§ 4.6 Overtime, if and when specifically authorized in writing in advance by the Owner shall be paid by the Owner on the basis of premium payment if any, plus the cost of insurance and taxes based on the premium payment period. No overhead or profit may be charged for overtime. The Contractor shall not be entitled to any payment for overtime necessitated by the failure of the Contractor to perform the Work in accordance with the Contract Documents including without limitation to the Contractor's failure to prosecute the Work diligently and on an uninterrupted basis and with a sufficient workforce so as to achieve completion of the Work within the time and in the manner contemplated by the Contract Documents, or otherwise due to the fault of the Contractor. In such instances if the Owner requires the Contractor to perform Work on an overtime basis, all costs for and associated with such overtime shall be borne by the Contractor.

#### PAGE 4

§ 5.1.1 Based upon Applications for Payment <u>properly completed and accompanied by all supporting documentation and other submittals required by the Contract Documents</u> submitted to the Architect by the Contractor and Certificates for Payment issued by the Architect, <u>and agreed to by Owner and not subsequently nullified by the Architect in accordance with the Contract Documents</u>, the Owner shall make progress payments on account of the Contract Sum to the Contractor as provided below and elsewhere in the Contract Documents.

...

§ 5.1.3 Provided that an Application for Payment Payment, which is in proper form and accompanied by required supporting documents and submittals, is received by the Architect not later than the 10th day of a month, certified for payment by the Owner and not subsequently nullified by the Owner in accordance with the Contract Documents, the Owner shall make payment of the certified amount to the Contractor not later than the 15th day of the following month. If an Application for Payment is received by the Architect after the application date fixed above, payment shall be made by the Owner not later than forty-five (45) days after the Architect receives the Application for Payment.

(Federal, state or local laws may require payment within a certain period of time.) for Payment in proper form and accompanied by required supporting documents and submittals and certifies payment to the Owner. Contractor is solely responsible for any delays in payment due in whole or in part to Contractor's failure to submit its payment application timely, in proper form and accompanied by all supporting documents and submittals required under the Contract.

...

- .1 Take that portion of the Contract Sum properly allocable to completed Work as determined by multiplying the percentage completion of each portion of the Work by the share of the Contract Sum allocated to that portion of the Work in the schedule of values, less retainage of ten\_percent (10%). Pending final determination of cost to the Owner of changes in the Work, amounts not in dispute shall be included as provided in Section 7.3.9 of AIA Document A201<sup>TM</sup>—2007, General Conditions of the Contract for Construction;
- Add that portion of the Contract Sum properly allocable to materials and equipment delivered and suitably stored at the site for subsequent incorporation in the completed construction (or, if approved in advance by the Owner, suitably stored off the site at a location agreed upon in writing), less retainage of ten percent ( 10 %);

#### PAGE 5

- § 5.1.7 The progress payment amount determined in accordance with Section 5.1.6 shall-may be further modified under the following circumstances:
  - .1 Add, upon Substantial Completion of the Work, a sum sufficient to increase the total payments to the full amount of the Contract Sum, less such amounts as the Architect shall determine for incomplete Work, retainage applicable to such work and unsettled claims; and

(Section 9.8.5 of AIA-Document A201 2007 requires release of applicable retainage upon Substantial Completion of Work with consent of surety, if any.)

Add, if final completion of the Work is thereafter materially delayed through no fault of the Contractor, any additional amounts payable in accordance with Section 9.10.3 of AIA Document A201 2007 and the retainage may be reduced if and as provided in Paragraph 9.3 of the General Conditions.

•••

- the Contractor has fully performed the Contract except for the Contractor's responsibility in accordance with the Contract Documents; provided that Owner shall have no obligation to make final payment if the Contractor is required to correct Work as provided in Section 12.2.2 of AIA Document A201-2007, and A201-2007 or as otherwise specified by the Contract Documents, or to satisfy other requirements, if any, which extend beyond final payment; and
- .2 a final Certificate for Payment has been issued by the Architect the Architect and the Owner has approved payment.

§ 5.2.2 The Owner's final payment to the Contractor shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, or as follows: 45days after the conditions set forth in Section 5.2.1, above, have been met.

...

#### § 5.3 WAIVER PROCEDURE/FORMAT

The first Application for Payment shall be accompanied by the Contractor's Partial Waiver of Lien to date for the full amount of the payment. Each subsequent monthly payment application shall be accompanied by the Contractor's Partial Waiver of Lien and the Partial Waivers by of subcontractors and suppliers who were included in the immediately preceding payment application to the extent of that payment. The Application for Final Payment shall be accompanied by Final Waivers of lien from the Contractor, subcontractors, and suppliers who have not previously furnished such Final Waivers. Final Waivers shall be for the full amount of the Contract and the Final Waiver of a Subcontractor shall be for the full amount of its Subcontract. All applications for payment shall be accompanied by affidavits from the Contractor, in triplicate, containing such information and in such form to comply with the Illinois Mechanics Lien Act (770 ILCS 60/001 et seq.) and showing in detail the sources of all labor and materials used and contracted to be used on the Project, including names and addresses of subcontractors and materials suppliers; amounts paid and remaining to be paid to each; together with all other documents as shall be necessary, in the sole judgment of the Owner, to waive all claims of liens to date and comply with all applicable state and local laws.

- (i) All waivers (partial and final) shall include language as applicable indicating either that:
  - All materials were taken from fully paid stock and delivered to job site in our own vehicles and all labor has been fully paid in accordance with prevailing wage laws; or
  - 2. Materials were provided by the following suppliers from whom waivers of lien are attached and all labor has been fully paid in accordance with prevailing wage laws.

# § 6.1 INITIAL DECISION MAKER

The Architect will serve as Initial Decision Maker pursuant to Section 15.2 of AIA Document A201–2007, unless the parties appoint below another individual, not a party to this Agreement, to serve as Initial Decision Maker. (If the parties mutually agree, insert the name, address and other contact information of the Initial Decision Maker, if other than the Architect.)

The Parties shall make claims and resolve disputes as provided in Article 15 of the General Conditions and elsewhere in the Contract Documents.

#### § 6.2 BINDING DISPUTE RESOLUTION

For any Claim subject to, but not resolved by, mediation pursuant to Section 15.3 of AIA Document A201 2007, the method of binding dispute resolution shall be as follows:

(Check the appropriate box. If the Owner and Contractor do not select a method of binding dispute resolution below, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, Claims will be resolved by litigation in a court of competent jurisdiction.)

[	-}-	Arbitration pursuant to Section 15.4 of AIA Document A201 2007
[	-}	Litigation in a court of competent jurisdiction
<u>-</u>	-}-	Other (Specify)

#### PAGE 6

- § 8.1 Where reference is made in this Agreement to a provision of AIA Document A201–2007 or another Contract Document, the reference refers to that provision as amended amended, superseded or supplemented by other provisions of the Contract Documents.
- § 8.2 Payments due and unpaid under the Contract shall bear interest from the date payment is due at the rate stated below, or in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

(Insert rate of interest agreed upon, if any.)

- % and at the rate provided under the Illinois Local Government Prompt Payment Act.

...

Alin Pop
Oak Brook Park District
1450 Forest Gate Road
Oak Brook, IL 60523
T: 630-645-9510
Email: apop@obparks.org

...

- § 8.6.1 Not less than the prevailing rate of wages as determined by the Illinois Department of Labor shall be paid to all laborers, workers and mechanics performing the Work. Contractor's bonds shall include a provision as will guarantee the faithful performance of this prevailing wage clause as herein provided and as provided in the General Conditions. Contractor shall comply with all other requirements of the Prevailing Wage Act.
- § 8.6.2 The Contractor shall be responsible for the supply and maintenance of any and all temporary equipment, utilities and facilities necessary to properly and safely complete and protect the Work, including without limitation those required by winter conditions. The Contractor shall provide and erect barricades and other safeguards adequate to warn of danger at the site and to protect persons and property from injury resulting from the Work.
- § 8.6.3 The Contractor shall limit materials and equipment storage to the immediate area of Work and such other areas as the Owner may designate. The Contractor shall promptly remove and properly dispose all construction material, trash, garbage and other debris off site.

inconsistency, ambiguity,	se specifically provided in t conflict, discrepancy or en Parties shall give precedence	ror in the Contract Docum	f and to the extent of any ent, and otherwise in interpreting the ents in the following order of priority:
(iii) Supplem (iv) General	ations. reement nentary and Special Condit Conditions ction Drawings	ions.	
\$8.6.5 The rights and reme limitation of any other rig	dies of the Owner stated in	the Contract Documents law or in equity.	shall be in addition to and not in
PAGE 7			
§ 9.1.1 The Agreement is and Contractor.Contractor	this executed AIA Docume, as modified by Owner.	nt A101–2007, Standard I	Form of Agreement Between Owner
§ 9.1.2 The General Condi- Construction. Construction	itions are AIA Document A	201–2007, General Cond	itions of the Contract for
§ 9.1.3 The Supplementary Supplementary, Special ar 12, 2016.	and other Conditions of the Co	ne Contract: Contract are those included	l in the Project Manual dated October
Document Document	Title	Date	Pages
The Specifications are the	ose included in the Project l	Manual dated October 12,	<u> 2016.</u>
Section	<del>Title</del>	Date	Pages
•••			
The Drawings are those in	ncluded in the Project Man	ual dated October 12, 201	<u>6.</u>
PAGE 8			
.1 AIA Document E.  1. All oth	201TM 2007, Digital Data I er documents contained in	Protocol Exhibit, if complete Project Manual dated	eted by the parties, or the following: October 12, 2016
2. Certifica reference	te of Insurance and endors	ements attached to and in	corporated in this Agreement by this
Payment Bond, at	tached to and incorporated:	in this Agreement by th	ce Bond, Labor and Material is reference. part of the Contract Documents.
AIA Document A	201 2007 provides that bid	ding requirements such as	s advertisement or invitation to bid,

	Instructi	ens to Bidders, sar	aple forms and the Contractor's bid are not part of the Contract Documents	
	<del>unless e</del>	numerated in this /	greement. They should be listed here only if intended to be part of the	
	Contrac	t Documents.)4. I	revailing Wage Determination and supersedes notice attached to and	
	incorpor	ated in this	Agreement by this reference.	
	this	5. Contractor's Pr Agreement by this	oposal, as modified by the Contract Documents, attached to and incorporated in	ņ
	uns	-	mpliance and Certifications Attachment and Substance Abuse Prevention	
	Program		Certification, attached to and incorporated in this Agreement by this reference.	
			the state of the s	
•••				
<del>(State b</del> A201–2	onding re 1007.)	quirements, if any,	and limits of liability for insurance required in Article 11-of AIA Document	
	Type of i	nsurance or bond	Limit of liability or bond amount (\$0.00)	
OAK B	ROOK PA	ARK DISTRICT		
		<del></del>		

# **Certification of Document's Authenticity**

AIA® Document D401™ - 2003

I, Nicole L. Karas, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 11:14:24 on 10/04/2016 under Order No.  $1549258185\_1$  from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA® Document  $A101^{TM} - 2007$ , Standard Form of Agreement Between Owner and Contractor—where the basis of payment is a Stipulated Sum, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

Nicole	Kaus			
(Signed)				 
Attorney				
Title)			-	 
10/4/16				
(Dated)	-	_		 _

# General Conditions of the Contract for Construction AIA Document A201 - 2007



# General Conditions of the Contract for Construction

#### for the following PROJECT:

(Name and location or address)
Tennis Center Reflective Ceiling and Insulation Project
Oak Brook Park District Tennis Center
1300 Forest Gate Road
Oak Brook, IL 60523

#### THE OWNER:

(Name and address)
Oak Brook Park District
1450 Forest Gate Road
Oak Brook, IL 60523

#### THE ARCHITECT:

(Name and address)

NA. No Architect is being used for this Project. For purposes of this Project, "Architect" shall mean "Owner."

# TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 OWNER
- 3 CONTRACTOR
- 4 ARCHITECT
- 5 SUBCONTRACTORS
- 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS
- 7 CHANGES IN THE WORK
- 8 TIME

User Notes:

- 9 PAYMENTS AND COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- 12 UNCOVERING AND CORRECTION OF WORK
- 13 MISCELLANEOUS PROVISIONS

#### **ADDITIONS AND DELETIONS:**

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

1

- 14 **TERMINATION OR SUSPENSION OF THE CONTRACT**
- 15 **CLAIMS AND DISPUTES**

INDEX	Architect's Additional Services and Expenses
(Numbers and Topics in Bold are Section Headings)	2.4.1, 11.3.1.1, 12.2.1, 13.5.2, 13.5.3, 14.2.4
,	Architect's Administration of the Contract
	3.1.3, 4.2, 3.7.4, 15.2, 9.4.1, 9.5
Acceptance of Nonconforming Work	Architect's Approvals
9.6.6, 9.9.3, <b>12.3</b>	2.4.1, 3.1.3, 3.5.1, 3.10.2, 4.2.7
Acceptance of Work	Architect's Authority to Reject Work
9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3, 12.3	3.5.1, 4.2.6, 12.1.2, 12.2.1
Access to Work	Architect's Copyright
<b>3.16,</b> 6.2.1, 12.1	1.1.7, 1.5
Accident Prevention	Architect's Decisions
10	3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 4.2.14,
Acts and Omissions	6.3.1, 7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2.1, 9.4.1, 9.5, 9.8.4,
3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5,	9.9.1, 13.5.2, 15.2, 15.3
10.2.8, 13.4.2, 13.7.1, 14.1, 15.2	Architect's Inspections
Addenda	3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 13.5
1.1.1, 3.11.1	Architect's Instructions
Additional Costs, Claims for	3.2.4, 3.3.1, 4.2.6, 4.2.7, 13.5.2
3.7.4, 3.7.5, 6.1.1, 7.3.7.5, 10.3, 15.1.4	Architect's Interpretations
Additional Inspections and Testing	4.2.11, 4.2.12
9.4.2, 9.8.3, 12.2.1, <b>13.5</b>	Architect's Project Representative
Additional Insured	4.2.10
11.1.4	Architect's Relationship with Contractor
Additional Time, Claims for	1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5.1,
3.2.4, 3.7.4, 3.7.5, 3.10.2, 8.3.2, <b>15.1.5</b>	3.7.4, 3.7.5, 3.9.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, 3.18,
Administration of the Contract	4.1.2, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5,
3.1.3, <b>4.2</b> , 9.4, 9.5	9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13.4.2, 13.5,
Advertisement or Invitation to Bid	15.2
1.1.1	Architect's Relationship with Subcontractors
Aesthetic Effect	1.1.2, 4.2.3, 4.2.4, 4.2.6, 9.6.3, 9.6.4, 11.3.7
4.2.13	Architect's Representations
Allowances	9.4.2, 9.5.1, 9.10.1
<b>3.8</b> , 7.3.8	Architect's Site Visits
All-risk Insurance	3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5
11.3.1, 11.3.1.1	Asbestos
Applications for Payment	10.3.1
<b>4.</b> 2.5, 7.3.9, 9.2, <b>9.3</b> , 9.4, 9.5.1, 9.6.3, 9.7.1, 9.10,	Attorneys' Fees
11.1.3	3.18.1, 9.10.2, 10.3.3
Approvals	Award of Separate Contracts
2.1.1, 2.2.2, 2.4, 3.1.3, 3.10.2, 3.12.8, 3.12.9, 3.12.10,	6.1.1, 6.1.2
4.2.7, 9.3.2, 13.5.1	Award of Subcontracts and Other Contracts for
Arbitration	Portions of the Work
8.3.1, 11.3.10, 13.1.1, 15.3.2, <b>15.4</b>	5.2
ARCHITECT	Basic Definitions
4	1.1
Architect, Definition of	Bidding Requirements
4.1.1	1.1.1, 5.2.1, 11.4.1
Architect, Extent of Authority	Binding Dispute Resolution
2.4.1, 3.12.7, 4.1, 4.2, 5.2, 6.3.1, 7.1.2, 7.3.7, 7.4,	9.7.1, 11.3.9, 11.3.10, 13.1.1, 15.2.5, 15.2.6.1, 15.3.1,
9.2.1, 9.3.1, 9.4, 9.5, 9.6.3, 9.8, 9.10.1, 9.10.3, 12.1,	15.3.2, 15.4.1
12.2.1, 13.5.1, 13.5.2, 14.2.2, 14.2.4, 15.1.3, 15.2.1	Boiler and Machinery Insurance
Architect, Limitations of Authority and	11.3.2
Responsibility	Bonds, Lien
2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2,	7.3.7.4, 9.10.2, 9.10.3
4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, 7.4.1,	Bonds, Performance, and Payment
9.4.2, 9.5.3, 9.6.4, 15.1.3, 15.2	7.3.7.4, 9.6.7, 9.10.3, 11.3.9, 11.4

AlA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatles. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale. User Notes:

Init.

1

3.7.1 Capitalization 1.3 3.7.1 Capitalization 1.3 Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5 Certificates for Payment 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 14.1.1, 3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Inspection, Testing or Approval 13.5.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1, 13.3, 11.3.4, 11.3, 11.3, 11.3, 9.10.3, 10.3.2, 11.3.1, 11.3, 11.3, 11.3, 9.1.3, 10.3.2, 15.1.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3, 11.3, 11.3, 11.3, 11.3, 11.3, 11.3, 6, 11.4.1 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3, 11.3, 7.1, 1.3, 11.3, 11.3, 6, 11.4.1 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3, 11.3, 7.1.3, 7.3, 10.3.2, 15.1.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3, 5.1.1.3, 7.1.3, 7.1.3, 10.3, 2, 15.1.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3, 5.1.1.3, 7.1.3, 7.4, 7.4.1, 15.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Construction Change Directives 1.1.1, 3.4.2, 3.12.8, 3.1.2, 2.3, 1.3, 1.3, 1.3, 1.3, 1.3, 2.2, 6.1.3, 15.1.5, 2 Contract Documents, Copies Furnished and Use of 1.3, 2.2, 2.3, 2.3, 2.3, 2.3, 2.3, 2.3, 2	Duilding Damais	0 14 01 41
Capitalization 1.3 Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5 Certificates for Payment 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10, 1.9.10, 3.1, 1.4.1.3, 1.42.4, 1.5.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 1.1.3 Change Orders 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 7.2.1, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.11, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 11.3.1, 11.3.9 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK CHANGES IN THE WORK CLaims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.2.1, 11.1, 11.3.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.24 Commencement of the Work, Definition of 8.1.2 Communications Relating to 3.7.1, 3.13, 3.15, 3.2, 2.2, 2.2, 5.3 Contract Documents, Definition of 9.1 Contract Documents, Definition of 9.1 Contract Documents, Definition of 1.1.1 Contract Documents, Definition of 1.1.1 Contract Documents, Definition of 9.1 Contract Sum, Defini	Building Permit	Completion, Substantial
Corrificate of Substantial Completion 9.83, 9.84, 9.8.5 Certificates for Payment 4.21, 4.22, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.101, 9.103, 1.4.1.13, 1.4.2.4, 1.5.1.3 Certificates of Insurance 113.5.4 Certificates of Insurance 110.2, 1.11.3 Change Orders 11.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 3.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.1, 5.1.1, 5.1.1, 5.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 11.3.1, 11.3.9, 12.1.3, 11.3.9, 12.1.3, 11.3.9, 13.1.3.2, 19.3.3, 19.3.1, 11.3.1, 11.3.9 Claims pefinition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims pefinition of 15.1.1 Claims for Additional Time 15.4.1 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.1, 11.3.6, 11.4.1 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Consenement of the Work, Conditions Relating to 2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communication, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.3, 11.3.5, 11.3.1, 13.2, 13.2, 2.2, 2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.3.1.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.2. Contract Sum, Definition of 9.1 Co		
Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5 Certificates for Payment 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 1, 14.1.3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.1.3 Change Orders 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9 Claims Por Additional Cost 13.2.4, 3.7.4, 6.1.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5 Concealed or Unknown Conditions CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7.7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1 CLAIMS AND DISPUTES 2.2.4, 6.1.1, 6.3.1, 9.10.3, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5 Concealed or Unknown Conditions CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7.7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.10.2, 10.13, 11.3.1, 13.2, 13.4.2, 15.4.4.2 Construction Change Directive, Definition of 7.3.1 CLAIMS AND DISPUTES 2.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5 Concealed or Unknown Conditions Construction Special or Unknown Condition of Change Directives 1.1.1, 6.1.4 Consent, Written 3.4.2, 3.1.2, 8.3.1, 1.3.1, 2.1.3.4, 2.1.3.4, 11.3.9, 11.4 Construction Change Directive, Definition of 7.3.1 Construction Change Directive, Definition of 1.1.1, 6.1.1.4 Construction Change Directive, Definition of 1.1.3, 1		
9.83, 9.84, 9.85 Certificates for Payment 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 1.4.1.3, 1.1.3, 4.1.3, 1.5.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Inspection, Testing or Approval 13.5.4 Cartificates of Inspection, Testing or Approval 13.5.4 Cartificates of Insurance 9.10.2, 11.1.3 Change Orders 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1, 9.10.3, 10.3.2, 113.1.2, 113.4, 11.3.9, 11.3.1, 11.3.9 Claims Orders, Definition of 13.1.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7.7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1, 11.3.9 Claims perintion of 15.1.1 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.7.4, 5.1.1, 7.1.2, 7.1.3, 10.1, 3.12.6, 5.2.1, 5.2.3 Concealed or Unknown Conditions, Claims for 3.7.4 Consented on Unknown Condition of Claims 15.4.1 Claims for Additional Time 3.2.4, 3.11, 1.1.3.9, 11.2.2 Consultation or Joinedr 15.1.3 Construction Change Directive, Definition of 1.1.2 Construction Change Directives 2.1.1, 3.2.4, 2.4.2, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1, 1.1.3, 11.3.4, 11.3.5, 11.4.1.4, 12.2.2 Continuing Contract Performance 15.1.3 Contract Oucuments, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.1.1 Completion, Conditions Relating to 3.4, 1.3.1, 3.1, 3.1, 3.1, 3.1, 3.1, 3.1, 3		
Certificates for Payment 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.3 Certificates of Inspurance 9.10.2, 11.1.3 Change Orders 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1, 2, 11.3.4, 11.3.9, 2.1.2, 15.1.3 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Construction Schedules, Contractor's 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Contract Documents, Contract Administration 3.13, 4, 9, 4, 9.5 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2 Contract Documents, Copies Furnished and Use of 1.5.2 Communications Facilitating Contract Administration 3.9.1, 4.24 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 2, 9.10.3, 11.3.1, 13.2, 13.4.2, 15.4.2 Construction Change Directives 1.1.1, 6.1.4 Construction Change Directives 1.1.1, 6.1.2 Construction Change Directives 1.1.1, 6.2 Continuing Contract Performance 15.1.3 Contract Decuments, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.2 Contract Documents, Copies Furnished and Use of 1.5.2 Contract Sum, Definition		
4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 14.1.1, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.1.3 Change Orders 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.12, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 12.1, 13.1, 19.03, 10.3.2, 11.3.1, 2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9, 9.3.1.1, 11.3.9, 9.3.1, 1, 1.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 15.3.1, 15.4.1 Claims for Additional Time 23.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.9, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 3.1 (2.2, 13.7, 3.1.1, 13.1, 1.1.4, 11.3.1,		
9.10.1, 19.10.3, 14.1.1.3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.1.3 Change Orders 1.1.1, 24.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 72.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7.7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.1.4, 13.4.2, 4, 15.1.6 Claims for Damages 3.2.4, 3.1.4, 13.4.2, 4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.5 Commencement of the Work, Conditions Relating to 2.2.1, 3.2, 3.4.1, 3.7.1, 3.101, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.13 Change Orders 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.2, 7.3.10, 3.2, 9.1.3, 11.3, 11.3.9, 12.1.2, 15.1.3 CIAIMS SID THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 113.9 CIAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.1.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3, 6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Commencement of the Work Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.22, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 51.2, 51.3.1, 52.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2, 7.3, 7.3, 7.3, 7.3, 7.4, 9.1, 9.3.2, 15.1.4, 15.2.5 Completion, Conditions Relating to 2.4.1, 3.11, 3.15, 4.22, 4.2.9, 8.2, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
13.5.4 Certificates of Insurance 9.10.2, 11.1.3 Change Orders 1.1.1, 24.1, 34.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.9, 12.1.1, 15.1.3 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 22.1, 3.2.2, 3.4.1, 3.7.1, 3.10, 3.12, 6.5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications, Relating to 2.4.1, 3.11, 3.15, 4.22, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
Cortificates of Insurance 9.10.2, 11.1.3 1.1.2, 24.1, 34.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 2.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Cost 13.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims for Damages 3.2.4, 3.1.3, 1.3.1, 1.1.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Conmencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Relating to 3.4.1, 3.11, 3.15, 4.22, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
9.10.2, 11.1.3 Change Orders 1.1.1, 24.1, 34.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1, 1, 9.10.3, 10.3.2, 11.3.1, 2.11.3.4, 11.3.9, 12.1.2, 15.1.3 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46, 1.1, 3.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.1, 3.2.2, 3.4.1, 3.7.1, 3.10, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Relating to 3.4.1, 3.7.1, 3.10, 3.12, 4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
Change Orders  1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3  Change Orders, Definition of 7.2.1  CHANGES IN THE WORK  2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9  Claims, Definition of 15.1.1  CLAIMS AND DISPUTES  3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4  Claims and Timely Assertion of Claims 15.4.1  Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4  Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5  Concealed or Unknown Conditions, Claims for 3.7.4  Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Cleaning Up 3.15, 6.3  Commencement of the Work, Conditions Relating to 15.1.3  Commencement of the Work, Definition of 8.1.2  Communications Facilitating Contract Administration 3.9.1, 4.2.4  Communications Facilitating Contract Administration 3.9.1, 4.2.4  Communications Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND		
1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.13.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3  Change Orders, Definition of 7.2.1  CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9  Claims, Definition of 15.1.1  CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4  Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4  Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5  Concealed or Unknown Conditions, Claims for 3.7.4  Claims for Damages 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Claims Subject to Arbitration 15.3.1, 15.4.1  Claims Subject to Arbitration 6.1.3.1  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10, 1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 11.3.1  Commencement of the Work, Definition of 8.1.2  Communications Facilitating Contract Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.01, 12.2, 13.7, 14.1.2  COMPLETION BY OWNER OR BY SEPARATE CONTRACTORS  1.1.4, 6  Construction Change Directive, Definition of 7.3.1  Construction Change Directives 1.1.4, 6  Construction Change Directives 1.1.1, 3.6  Construction Change Directives 1.1.1, 4.6  Construction Change Directives 1.1.1, 4.6		3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5, 9.9.1,
5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.1, 13.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims for Additional Time 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 2.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.9, 14 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.1, 11.3.1, 11.3.6, 11.4.1 Claims Subject to Arbitration 15.3.1, 15.4.1 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Communications Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.01, 12.2, 13.7, 14.1.2 COMPLETION BY OWNER OR BY SEPARATE CONTRACTORS 1.1.4, 6 Construction Change Directive, Definition of 7.3.1 Contract Definition Change Directives Contractor's 3.10.2, 12.2, 6.1.3, 151.5.2 Continuing Contract Pe		9.10.2, 9.10.3, 11.3.1, 13.2, 13.4.2, 15.4.4.2
8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.1, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Communications Facilitating Contract Administration 3.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 91.0, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND	1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8,	Consolidation or Joinder
12.12, 15.1.3   SEPARATE CONTRACTORS     Change Orders, Definition of     7.2.1   Change St N THE WORK     2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1,     3.3.1, 11.3.9     Claims, Definition of     15.1.1   Construction Change Directives     1.1, 3.4.2, 3.1.2, 8.4.2, 8, 7.1.1, 7.1.2, 7.1.3, 7.3,     9.3.1.1   Construction Change Directives     1.1, 3.4.2, 3.1.2, 8.4.2, 8, 7.1.1, 7.1.2, 7.1.3, 7.3,     9.3.1.1   Construction Change Directives     1.1.1, 3.4.2, 3.1.2, 8.4.2, 8, 7.1.1, 7.1.2, 7.1.3, 7.3,     9.3.1.1   Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.5, 2     Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.5, 2.1, 7.1.3, 7.3,     9.3.1.1   Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.5, 2.2, 1.3, 1.5, 1.5, 2.2     Continuing Contract Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.3, 1.5, 1.5, 2.2     Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.5, 2.2     Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.5, 1.5     Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.5, 1.5, 1.5, 1.5     Construction Change Directives     1.1.1, 3.4.2, 1.3, 1.5, 1.3, 1.5, 1.5, 1.5, 1.5, 1.5, 1.5, 1.5, 1.5		
Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 9.3.1.1, 11.3.9 9.3.1.1, 11.3.9 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		CONSTRUCTION BY OWNER OR BY
7.2.1 CHANGES IN THE WORK 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.7.4, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Displect to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10, 3.12.6, 5.2.1, 5.2.3, 6.1.1, 11.3.6, 11.4.1 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  Construction Change Directives 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Change Directives 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Change Directives 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Change Directives 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Schedules, Contractor's 3.10, 3.12.1, 3.12.2, 6.1.3, 151.5.2 Continuing Contract Assignment of Subcontracts 54, 14.2.2.2 Continuing Contract Performance 15.1.3 Contract, Definition of 1.1.2 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Administration 3.7.4, 3.19, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Contract Documents, Definition of 1.5.2, 2.2.5, 5.3 Contract Sum, Definition of 9.1 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 2.2.3, 7.2.1, 7.3.1, 7.3.5, 7.4, 9.1 9.2.1.1, 13.1.5, 14.2.4, 14.3.2, 15.1.4 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 2.2, 2.3, 7.2.1, 3, 7		SEPARATE CONTRACTORS
CHANGES IN THE WORK  2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9  Claims, Definition of 15.1.1  CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4  Claims and Timely Assertion of Claims 15.4.1  Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4  Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5  Concealed or Unknown Conditions, Claims for 3.7.4  Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Cleaning Up 3.15, 6.3  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 8.1.2  Communications Facilitating Contract  Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.1, 12.2, 13.7, 14.1.2, 12.3, 7.1.3, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.3.1.1  Construction Change Directives 1.1.1, 3.4.2, 4, 2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1  Construction Change Directives 1.1.1, 3.4.2, 4, 1.2, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1  Construction Change Directives 1.1.1, 3.4.2, 4, 2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1  Construction Change Directives 1.1.1, 3.4.2, 4, 2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1  Contract Decuments, Copies Currients of Subcontracts 5.4, 14.2.2.2  Continuing Contract Performance 1.1.2  Contract Administration 3.1.3, 4, 9.4, 9.5  Contract Administration of 1.1.2  Contract Sum 3.7.4, 3.1, 3.1.3, 14.2.4, 15.2.4  1.1.1  Contract Sum, Definition of 9.1  Contract Sum, Definition of 9.1  Contract Time 3.7.4, 3.1, 1.2.2, 1.3.1, 1.2.2, 1.3.1, 1.3.2, 1.3.3, 1, 7.3.5, 7.4, 9.1  Sum of Chalms of Subcontract Administration 3.1.3, 4, 9.4, 9.5  Contract Sum, Definition of 9.1  Contract Sum, Defini		1.1.4, <b>6</b>
2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.3, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10, 1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		Construction Change Directive, Definition of
9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		7.3.1
Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		Construction Change Directives
15.1.1 CLAIMS AND DISPUTES 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 910, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  Contract Tocuments Schedules, Contractor's 3.1.0, 3.12.1, 3.12.2, 6.1.3, 151.5.2 Contingent Assignment of Subcontracts 5.4, 14.2.2.2 Continuing Contract Performance 15.1.3 Contract, Definition of 1.1.2 Contract Award and Execution, Conditions Relating to 3.1.3, 4, 9.4, 9.5 Contract Documents, The 1.1.1 Contract Documents, Copies Furnished and Use of 1.1.1 Contract Documents, Definition of 1.1.1 Contract Documents, Definition of 1.1.1 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3,
CLAIMS AND DISPUTES 3.24, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.24, 3.74, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.24, 3.74, 6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.24, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 11.5.1.4 Commencement of the Work, Definition of 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.1.3, 3.1, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 91, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Contingent Assignment of Subcontracts 54, 14.2.2.2 Contingent Assignment of Subcontract Performance 15.1.3 Contract, Definition of 11.2 Contract Administration 3.1.3, 1, 1.1.3, 1, 1.3.6, 11.4.1, 11.5.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 Contract Sum 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.1, 15.2.5		9.3.1.1
3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4  Claims and Timely Assertion of Claims 15.4.1  Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4  Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5  Concealed or Unknown Conditions, Claims for 3.7.4  Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Cleaning Up 3.15, 6.3  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 8.1.2  Communications Facilitating Contract Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 910, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND		Construction Schedules, Contractor's
15.4 Claims and Timely Assertion of Claims 15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.0, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
15.4   15.5   15.4		Contingent Assignment of Subcontracts
15.4.1 Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 3.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		
Claims for Additional Cost 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.4.6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 3.9.1, 4.2.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  Contract Addition of 1.1.2 Contract, Definition of 5.4.1.1, 11.3.9, 14 Contract Administration 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3, 6, 11.4.1 Contract Documents, Copies Furnished and Use of 1.5.2, 5.3 Contract Documents, Definition of 1.1.1 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		Continuing Contract Performance
3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		15.1.3
Claims for Additional Time 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5  Concealed or Unknown Conditions, Claims for 3.7.4  Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Cleaning Up 3.15, 6.3  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Communications Facilitating Contract  Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND  CONTRACT, TERMINATION OR SUSPENSION OF THE Sun, 11.3.9, 14, 2.9, 4, 9, 5 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Documents, Copidations Relating to 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.1.1 Contract Sum Suspension Su		Contract, Definition of
3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5  Concealed or Unknown Conditions, Claims for 3.7.4  Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Cleaning Up 3.15, 6.3  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 8.1.2  Communications Facilitating Contract  Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.7.4, 3.1, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND  SUSPENSION OF THE 5.4.1.1, 11.3.9, 14  Contract Administration 3.1.3, 4, 9.4, 9.5  Contract Administration 3.1.3, 4, 9.4, 9.5  Contract Award and Execution, Conditions Relating to 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1  Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3  Contract Documents, Definition of 1.5.2, 2.2.5, 5.3  Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5  Contract Sum, Definition of 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5, 1, 15.2.5		1.1.2
Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Time 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND		CONTRACT, TERMINATION OR
Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Time 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  5.4.1.1, 11.3.9, 14 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Award and Execution, Conditions Relating to 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1 Contract Documents, The 1.1.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.1.1 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  5.4.1.1, 11.3.9, 14 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 9.6.7, 9.7, 10.3.2, 12.3.1, 7.3.5, 7.4, 9.1, 9.1, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 9.6.7, 9.7, 10.3.2, 12.1.1, 14.3.2, 15.1.4, 9.6.7, 9.7, 10.3.2, 12.1.1, 14.3.2, 15.1.4, 11.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1 1.1 1		SUSPENSION OF THE
Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  3.1.3, 4, 9.4, 9.5 Contract Award and Execution, Conditions Relating to 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1 Contract Documents, The 1.1.1 Contract Documents, Copies Furnished and Use of 1.5.2, 5.3 Contract Documents, Definition of 1.5.2, 2.2.5, 5.3 Contract Sum Contract Sum Contract Sum Contract Sum Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, COMPLETION, PAYMENTS AND		5.4.1.1, 11.3.9, 14
3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6  Claims Subject to Arbitration 15.3.1, 15.4.1  Cleaning Up 3.15, 6.3  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, Contract Sum, Definition of 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND		Contract Administration
11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  to Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.4, 1.1.1 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		3.1.3, 4, 9.4, 9.5
Claims Subject to Arbitration 15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.2, 2.2.5, 5.3 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		Contract Award and Execution, Conditions Relating
15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.2, 2.2.5, 5.3 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		
15.3.1, 15.4.1 Cleaning Up 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 Communications Facilitating Contract Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1
3.15, 6.3  Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5  Communications Facilitating Contract Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND  Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3  Contract Documents, Definition of 1.5.2, 2.2.5, 5.3  Contract Sum  3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5		
Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5  Communications Facilitating Contract 4.2.4  Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND 1.5.2.5  Contract Documents, Definition of 1.1.1  Contract Sum  Solution of 1.5.2, 2.2.5, 5.3  Contract Sum  Solution of 1.1.1  Contract Sum  Solution of 9.1  Contract Time  3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		1.1.1
2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5  Communications Facilitating Contract Administration 3.9.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND  Contract Documents, Definition of 1.1.1  Contract Sum 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 15.2.5  Contract Sum, Definition of 9.1  Contract Sum, Definition of 9.1  Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		Contract Documents, Copies Furnished and Use of
6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4  Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5  Communications Facilitating Contract 15.2.5  Administration 29.1, 4.2.4  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND 11.4.1, 11.3.1, 11.3.6, 11.4.1, 11.3.1, 11.3.1, 11.3.1, 11.3.1, 11.3.1, 11.3.1, 11.3.1, 11.3.1, 12.3, 13.7, 13.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5		1.5.2, 2.2.5, 5.3
15.1.4 Commencement of the Work, Definition of 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Communications Facilitating Contract 15.2.5 Contract Sum, Definition of 3.9.1, 4.2.4 9.1 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, COMPLETION, PAYMENTS AND 15.1.5.1, 15.2.5		Contract Documents, Definition of
Commencement of the Work, Definition of       3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4,         8.1.2       9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4,         Communications Facilitating Contract       15.2.5         Administration       Contract Sum, Definition of         3.9.1, 4.2.4       9.1         Completion, Conditions Relating to       Contract Time         3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1,       3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4,         9.10, 12.2, 13.7, 14.1.2       8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2,         COMPLETION, PAYMENTS AND       15.1.5.1, 15.2.5		1.1.1
8.1.2 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5  Administration Contract Sum, Definition of 9.1  Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, COMPLETION, PAYMENTS AND 15.1.5.1, 15.2.5		Contract Sum
Communications Facilitating Contract       15.2.5         Administration       Contract Sum, Definition of         3.9.1, 4.2.4       9.1         Completion, Conditions Relating to       Contract Time         3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1,       3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4,         9.10, 12.2, 13.7, 14.1.2       8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2,         COMPLETION, PAYMENTS AND       15.1.5.1, 15.2.5		3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, <b>9.1</b> , 9.4.2, 9.5.1.4,
AdministrationContract Sum, Definition of3.9.1, 4.2.49.1Completion, Conditions Relating toContract Time3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1,3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4,9.10, 12.2, 13.7, 14.1.28.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2,COMPLETION, PAYMENTS AND15.1.5.1, 15.2.5		9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4,
3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 COMPLETION, PAYMENTS AND  9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		
Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND  Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		Contract Sum, Definition of
3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2  COMPLETION, PAYMENTS AND  3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		
9.10, 12.2, 13.7, 14.1.2 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, COMPLETION, PAYMENTS AND 15.1.5.1, 15.2.5		Contract Time
9.10, 12.2, 13.7, 14.1.2 <b>COMPLETION, PAYMENTS AND</b> 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5		
COMPLETION, PAYMENTS AND 15.1.5.1, 15.2.5		
9 Contract Time, Definition of		
	9	Contract Time, Definition of

AIA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatles. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

Init.

8.1.1	Costs
CONTRACTOR	2.4.1, 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3,
3	7.3.3.3, 7.3.7, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6,
Contractor, Definition of	11.3, 12.1.2, 12.2.1, 12.2.4, 13.5, 14
3.1, 6.1.2	Cutting and Patching
Contractor's Construction Schedules	3.14, 6.2.5
<b>3.10</b> , 3.12.1, 3.12.2, 6.1.3, 15.1.5.2	Damage to Construction of Owner or Separate
Contractor's Employees	Contractors
3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3,	3.14.2, 6.2.4, 10.2.1.2, 10.2.5, 10.4, 11.1.1, 11.3,
11.1.1, 11.3.7, 14.1, 14.2.1.1,	12.2.4
Contractor's Liability Insurance	Damage to the Work
11.1	3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4.1, 11.3.1, 12.2.4
Contractor's Relationship with Separate Contractors	Damages, Claims for
and Owner's Forces	3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1,
3.12.5, 3.14.2, 4.2.4, 6, 11.3.7, 12.1.2, 12.2.4	11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6
Contractor's Relationship with Subcontractors	Damages for Delay
1.2.2, 3.3.2, 3.18.1, 3.18.2, 5, 9.6.2, 9.6.7, 9.10.2,	6.1.1, 8.3.3, 9.5.1.6, 9.7, 10.3.2
11.3.1.2, 11.3.7, 11.3.8	Date of Commencement of the Work, Definition of
Contractor's Relationship with the Architect	8.1.2
1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5.1,	Date of Substantial Completion, Definition of
3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.3, 4.2, 5.2,	8.1.3
6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6,	Day, Definition of
10.3, 11.3.7, 12, 13.5, 15.1.2, 15.2.1	8.1.4
Contractor's Representations	Decisions of the Architect
3.2.1, 3.2.2, 3.5.1, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2	3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 15.2, 6.3,
Contractor's Responsibility for Those Performing the	7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2.1, 9.4, 9.5.1, 9.8.4, 9.9.1,
Work	13.5.2, 14.2.2, 14.2.4, 15.1, 15.2
3.3.2, 3.18, 5.3.1, 6.1.3, 6.2, 9.5.1, 10.2.8	Decisions to Withhold Certification
Contractor's Review of Contract Documents	9.4.1 <b>, 9.5</b> , 9.7, 14.1.1.3
3.2	Defective or Nonconforming Work, Acceptance,
Contractor's Right to Stop the Work	Rejection and Correction of
9.7	2.3.1, 2.4.1, 3.5.1, 4.2.6, 6.2.5, 9.5.1, 9.5.2, 9.6.6,
Contractor's Right to Terminate the Contract	9.8.2, 9.9.3, 9.10.4, 12.2.1
14.1, 15.1.6	Defective Work, Definition of
Contractor's Submittals	3.5.1
3.10, 3.11, 3.12.4, 4.2.7, 5.2.1, 5.2.3, 9.2, 9.3, 9.8.2,	Definitions
9.8.3, 9.9.1, 9.10.2, 9.10.3, 11.1.3, 11.4.2	1.1, 2.1.1, 3.1.1, 3.5.1, 3.12.1, 3.12.2, 3.12.3, 4.1.1,
Contractor's Superintendent	15.1.1, 5.1, 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1
3.9, 10.2.6	Delays and Extensions of Time
Contractor's Supervision and Construction	3.2., 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4.1, <b>8.3</b> , 9.5.1, 9.7.1,
Procedures	10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5
1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4,	Disputes
7.1.3, 7.3.5, 7.3.7, 8.2, 10, 12, 14, 15.1.3 Contractual Liability Insurance	6.3.1, 7.3.9, 15.1, 15.2
11.1.1.8, 11.2	Documents and Samples at the Site
Coordination and Correlation	3.11
1.2, 3.2.1, 3.3.1, 3.10, 3.12.6, 6.1.3, 6.2.1	Drawings, Definition of
Copies Furnished of Drawings and Specifications	1.1.5
1.5, 2.2.5, 3.11	Drawings and Specifications, Use and Ownership of 3.11
Copyrights	Effective Date of Insurance
1.5, <b>3.1</b> 7	8.2.2, 11.1.2
Correction of Work	Emergencies
2.3, 2.4, 3.7.3, 9.4.2, 9.8.2, 9.8.3, 9.9.1, 12.1.2, <b>12.2</b>	<b>10.4</b> , 14.1.1.2, 15.1.4
Correlation and Intent of the Contract Documents	Employees, Contractor's
1.2	3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2,
Cost, Definition of	10.3.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1
7.3.7	Equipment, Labor, Materials or
- <del> ·</del>	-quipmont, Lacon, intacentals of

Init.

1.1.3, 1.1.6, 3.4, 3.5.1, 3.8.2, 3.8.3, 3.12, 3.13.1,	Insurance
3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3,	3.18.1, 6.1.1, 7.3.7, 9.3.2, 9.8.4, 9.9.1, 9.10.2, 11
9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2	Insurance, Boiler and Machinery
Execution and Progress of the Work	11.3.2
1.1.3, 1.2.1, 1.2.2, 2.2.3, 2.2.5, 3.1, 3.3.1, 3.4.1, 3.5.1,	Insurance, Contractor's Liability
3.7.1, 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.5, 8.2,	11.1
9.5.1, 9.9.1, 10.2, 10.3, 12.2, 14.2, 14.3.1, 15.1.3	Insurance, Effective Date of
Extensions of Time	8.2.2, 11.1.2
3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 7.4.1, 9.5.1, 9.7.1,	Insurance, Loss of Use
10.3.2, 10.4.1, 14.3, 15.1.5, 15.2.5	11.3.3
Failure of Payment	Insurance, Owner's Liability
9.5.1.3, <b>9.7</b> , 9.10.2, 13.6, 14.1.1.3, 14.2.1.2	11.2
Faulty Work	Insurance, Property
(See Defective or Nonconforming Work)	10.2.5, 11.3
Final Completion and Final Payment	Insurance, Stored Materials
4.2.1, 4.2.9, 9.8.2, <b>9.10</b> , 11.1.2, 11.1.3, 11.3.1, 11.3.5,	9.3.2, 11.4.1.4
12.3.1, 14.2.4, 14.4.3	INSURANCE AND BONDS
Financial Arrangements, Owner's	11
2.2.1, 13.2.2, 14.1.1.4	Insurance Companies, Consent to Partial Occupancy
Fire and Extended Coverage Insurance	9.9.1, 11.4.1.5
11.3.1.1	Insurance Companies, Settlement with
GENERAL PROVISIONS	11.4.10
1	Intent of the Contract Documents
Governing Law	1.2.1, 4.2.7, 4.2.12, 4.2.13, 7.4
13.1	Interest
Guarantees (See Warranty)	13.6
Hazardous Materials	Interpretation
10.2.4, 10.3	1.2.3, 1.4, 4.1.1, 5.1, 6.1.2, 15.1.1
Identification of Subcontractors and Suppliers	Interpretations, Written
5.2.1	4.2.11, 4.2.12, 15.1.4
Indemnification	Judgment on Final Award
3.17.1, <b>3.18</b> , 9.10.2, 10.3.3, 10.3.5, 10.3.6, 11.3.1.2, 11.3.7	15.4.2
Information and Services Required of the Owner	Labor and Materials, Equipment
2.1.2, <b>2.2</b> , 3.2.2, 3.12.4, 3.12.10, 6.1.3, 6.1.4, 6.2.5,	1.1.3, 1.1.6, 3.4, 3.5.1, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1,
9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2.1, 11.4, 13.5.1,	4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3,
13.5.2, 14.1.1.4, 14.1.4, 15.1.3	9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2
Initial Decision	Labor Disputes 8.3.1
15.2	Laws and Regulations
Initial Decision Maker, Definition of	1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13.1, 4.1.1, 9.6.4, 9.9.1,
1.1.8	10.2.2, 11.1.1, 11.3, 13.1.1, 13.4, 13.5.1, 13.5.2,
Initial Decision Maker, Decisions	13.6.1, 14, 15.2.8, 15.4
14.2.2, 14.2.4, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5	Liens
Initial Decision Maker, Extent of Authority	2.1.2, 9.3.3, 9.10.2, 9.10.4, 15.2.8
14.2.2, 14.2.4, 15.1.3, 15.2.1, 15.2.2, 15.2.3, 15.2.4,	Limitations, Statutes of
15.2.5	12.2.5, 13.7, 15.4.1.1
Injury or Damage to Person or Property	Limitations of Liability
<b>10.2.8</b> , 10.4.1	2.3.1, 3.2.2, 3.5.1, 3.12.10, 3.17.1, 3.18.1, 4.2.6,
Inspections	4.2.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 10.2.5, 10.3.3,
3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3,	11.1.2, 11.2.1, 11.3.7, 12.2.5, 13.4.2
9.9.2, 9.10.1, 12.2.1, 13.5	Limitations of Time
Instructions to Bidders	2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7,
1.1.1	5.2, 5.3.1, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, 9.2.1, 9.3.1,
Instructions to the Contractor	9.3.3, 9.4.1, 9.5, 9.6, 9.7.1, 9.8, 9.9, 9.10, 11.1.3,
3.2.4, 3.3.1, 3.8.1, 5.2.1, 7, 8.2.2, 12, 13.5.2	11.3.1.5, 11.3.6, 11.3.10, 12.2, 13.5, 13.7, 14, 15
Instruments of Service, Definition of 1.1.7	Loss of Use Insurance
	11.3.3

AIA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale. User Notes:

lnit.

2.1.2, <b>2.2</b> , 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2,
9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2.1, 11.3, 13.5.1,
13.5.2, 14.1.1.4, 14.1.4, 15.1.3
Owner's Authority
1.5, 2.1.1, 2.3.1, 2.4.1, 3.4.2, 3.8.1, 3.12.10, 3.14.2,
4.1.2, 4.1.3, 4.2.4, 4.2.9, 5.2.1, 5.2.4, 5.4.1, 6.1, 6.3.1,
7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.1, 9.3.2, 9.5.1, 9.6.4,
9.9.1, 9.10.2, 10.3.2, 11.1.3, 11.3.3, 11.3.10, 12.2.2,
12.3.1, 13.2.2, 14.3, 14.4, 15.2.7
Owner's Financial Capability
2.2.1, 13.2.2, 14.1.1.4
Owner's Liability Insurance
11.2
Owner's Loss of Use Insurance
11.3.3
Owner's Relationship with Subcontractors
1 1 2 5 2 5 3 5 4 0 6 4 0 10 2 14 2 2
1.1.2, 5.2, 5.3, 5.4, 9.6.4, 9.10.2, 14.2.2
Owner's Right to Carry Out the Work
<b>2.4</b> , 14.2.2
Owner's Right to Clean Up
6.3
Owner's Right to Perform Construction and to
Award Separate Contracts
6.1
Owner's Right to Stop the Work
2.3
Owner's Right to Suspend the Work
14.3
Owner's Right to Terminate the Contract
14.2
Ownership and Use of Drawings, Specifications
and Other Instruments of Service
1.1.1, 1.1.6, 1.1.7, <b>1.5</b> , 2.2.5, 3.2.2, 3.11.1, 3.17.1,
4.2.12, 5.3.1
Partial Occupancy or Use
9.6.6, <b>9.9</b> , 11.3.1.5
Patching, Cutting and
3.14, 6.2.5
Patents
3.17
Payment, Applications for
4.2.5, 7.3.9, 9.2.1, <b>9.3</b> , 9.4, 9.5, 9.6.3, 9.7.1, 9.8.5,
9.10.1, 14.2.3, 14.2.4, 14.4.3
Payment, Certificates for
4.2.5, 4.2.9, 9.3.3, <b>9.4</b> , 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1,
9.10.3, 13.7, 14.1.1.3, 14.2.4
Payment, Failure of
9.5.1.3, <b>9.7</b> , 9.10.2, 13.6, 14.1.1.3, 14.2.1.2
Payment, Final
4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.4.1, 11.4.5,
12.3.1, 13.7, 14.2.4, 14.4.3
Payment Bond, Performance Bond and
7.3.7.4, 9.6.7, 9.10.3, 11.4.9, <b>11.4</b>
Payments, Progress
9.3, <b>9.6</b> , 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3
/···, /···, /····, /·.··, 10.0, 17.2.J, 10.1.J
PAVMENTS AND COMDITETION
PAYMENTS AND COMPLETION 9

init.

Payments to Subcontractors	Rights and Remedies
5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, 11.4.8,	1.1.2, 2.3, 2.4, 3.5.1, 3.7.4, 3.15.2, 4.2.6, 4.5, 5.3, 5.4
14.2.1.2	6.1, 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.2,
PCB	12.2.4, <b>13.4</b> , 14, 15.4
10.3.1	Royalties, Patents and Copyrights
Performance Bond and Payment Bond	3.17
7.3.7.4, 9.6.7, 9.10.3, 11.4.9, <b>11.4</b>	Rules and Notices for Arbitration
Permits, Fees, Notices and Compliance with Laws	15.4.1
2.2.2, <b>3.7</b> , 3.13, 7.3.7.4, 10.2.2	Safety of Persons and Property
PERSONS AND PROPERTY, PROTECTION	10.2, 10.4
OF	Safety Precautions and Programs
10	3.3.1, 4.2.2, 4.2.7, 5.3.1, <b>10.1</b> , 10.2, 10.4
Polychlorinated Biphenyl	Samples, Definition of
10.3.1	3.12.3
Product Data, Definition of	Samples, Shop Drawings, Product Data and
3.12,2	3.11, <b>3.12</b> , 4.2.7
Product Data and Samples, Shop Drawings	Samples at the Site, Documents and
3.11, <b>3.12</b> , <b>4.</b> 2.7	3.11
Progress and Completion	Schedule of Values
4.2.2, <b>8.2</b> , 9.8, 9.9.1, 14.1.4, 15.1.3	9.2, 9.3.1
Progress Payments	Schedules, Construction
9.3, <b>9.6</b> , 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3	1.4.1.2, 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2
Project, Definition of the	Separate Contracts and Contractors
1.1.4	
Project Representatives	1.1.4, 3.12.5, 3.14.2, 4.2.4, 4.2.7, 6, 8.3.1, 11.4.7, 12.1.2
4.2.10	
Property Insurance	Shop Drawings, Definition of 3.12.1
10.2.5, <b>11.3</b>	
PROTECTION OF PERSONS AND PROPERTY	Shop Drawings, Product Data and Samples
10	3.11, <b>3.12</b> , 4.2.7
Regulations and Laws	Site, Use of
	<b>3.13</b> , 6.1.1, 6.2.1
1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 9.9.1,	Site Inspections
10.2.2, 11.1, 11.4, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14,	3.2.2, 3.3.3, 3.7.1, 3.7.4, 4.2, 9.4.2, 9.10.1, 13.5
15.2.8, 15.4	Site Visits, Architect's
Rejection of Work	3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5
3.5.1, 4.2.6, 12.2.1	Special Inspections and Testing
Releases and Waivers of Liens	4.2.6, 12.2.1, 13.5
9.10.2	Specifications, Definition of the
Representations	1.1.6
3.2.1, 3.5.1, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.4.2, 9.5.1,	Specifications, The
9.8.2, 9.10.1	1.1.1, <b>1.1.6</b> , 1.2.2, 1.5, 3.11, 3.12.10, 3.17, 4.2.14
Representatives	Statute of Limitations
2.1.1, 3.1.1, 3.9, 4.1.1, 4.2.1, 4.2.2, 4.2.10, 5.1.1,	13.7, 15.4.1.1
5.1.2, 13.2.1	Stopping the Work
Responsibility for Those Performing the Work	2.3, 9.7, 10.3, 14.1
3.3.2, 3.18, 4.2.3, 5.3.1, 6.1.3, 6.2, 6.3, 9.5.1, 10	Stored Materials
Retainage	6.2.1, 9.3.2, 10.2.1.2, 10.2.4, 11.4.1.4
9.3.1, 9.6.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3	Subcontractor, Definition of
Review of Contract Documents and Field	5.1.1
Conditions by Contractor	SUBCONTRACTORS
<b>3.2</b> , 3.12.7, 6.1.3	5
Review of Contractor's Submittals by Owner and	Subcontractors, Work by
Architect	1.2.2, 3.3.2, 3.12.1, 4.2.3, 5.2.3, 5.3, 5.4, 9.3.1.2,
	9.6.7
3.10.1, 3.10.2, 3.11, 3.12, 4,2, 5.2, 6.1.3, 9.2, 9.8.2	
	Subcontractual Relations
3.10.1, 3.10.2, 3.11, 3.12, 4.2, 5.2, 6.1.3, 9.2, 9.8.2 Review of Shop Drawings, Product Data and Samples by Contractor	<b>Subcontractual Relations 5.3,</b> 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, 11.4.7, 11.4.8,

Payments to Subcontractors

Init.

e American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

(1950958438)

Submittals	212 222 422 426 420 042 002 002
3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 7.3.7, 9.2, 9.3,	3.1.3, 3.3.3, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2,
9.8, 9.9.1, 9.10.2, 9.10.3, 11.1.3	9.10.1, 10.3.2, 11.4.1.1, 12.2.1, <b>13.5</b>
Submittal Schedule	TIME
3.10.2, 3.12.5, 4.2.7	8 T' D. I. I. I.
	Time, Delays and Extensions of
Subrogation, Waivers of	3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4.1, 8.3, 9.5.1, 9.7.1,
6.1.1, 11.4.5, 11.3.7	10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5
Substantial Completion	Time Limits
4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, <b>9.8</b> , 9.9.1, 9.10.3,	2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2,
12.2, 13.7	4.4, 4.5, 5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1,
Substantial Completion, Definition of	9.3.3, 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3,
9.8.1	11.4.1.5, 11.4.6, 11.4.10, 12.2, 13.5, 13.7, 14, 15.1.2,
Substitution of Subcontractors	15.4
5.2.3, 5.2.4	Time Limits on Claims
Substitution of Architect	3.7.4, 10.2.8, <b>13.7</b> , 15.1.2
4.1.3	Title to Work
Substitutions of Materials	9.3.2, 9.3.3
3.4.2, 3.5.1, 7.3.8	Transmission of Data in Digital Form
Sub-subcontractor, Definition of	1.6
5.1.2	UNCOVERING AND CORRECTION OF
Subsurface Conditions	WORK
3.7.4	12
Successors and Assigns	Uncovering of Work
13.2	12.1
Superintendent	Unforeseen Conditions, Concealed or Unknown
3.9, 10.2.6	
Supervision and Construction Procedures	3.7.4, 8.3.1, 10.3
1.2.2, <b>3.3</b> , 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4,	Unit Prices
	7.3.3.2, 7.3.4
7.1.3, 7.3.7, 8.2, 8.3.1, 9.4.2, 10, 12, 14, 15.1.3	Use of Documents
Surety	1.1.1, 1.5, 2.2.5, 3.12.6, 5.3
5.4.1.2, 9.8.5, 9.10.2, 9.10.3, 14.2.2, 15.2.7	Use of Site
Surety, Consent of	<b>3.13</b> , 6.1.1, 6.2.1
9.10.2, 9.10.3	Values, Schedule of
Surveys	<b>9.2,</b> 9.3.1
2.2.3	Waiver of Claims by the Architect
Suspension by the Owner for Convenience	13.4.2
14.3	Waiver of Claims by the Contractor
Suspension of the Work	9.10.5, 11.4.7, 13.4.2, 15.1.6
5.4.2, 14.3	Waiver of Claims by the Owner
Suspension or Termination of the Contract	9.9.3, 9.10.3, 9.10.4, 11.4.3, 11.4.5, 11.4.7, 12.2.2.1,
5.4.1.1, 11.4.9, 14	13.4.2, 14.2.4, 15.1.6
Taxes	Waiver of Consequential Damages
3.6, 3.8.2.1, 7.3.7.4	14.2.4, 15.1.6
Termination by the Contractor	Waiver of Liens
14.1, 15.1.6	9.10.2, 9.10.4
Termination by the Owner for Cause	Waivers of Subrogation
5.4.1.1, <b>14.2</b> , 15.1.6	6.1.1, 11.4.5, 11.3.7
Termination by the Owner for Convenience	Warranty
14.4	3.5, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.4, 12.2.2, 13.7.1
Termination of the Architect	Weather Delays
4.1.3	15.1.5.2
Termination of the Contractor	Work, Definition of
14.2.2	1.1.3
TERMINATION OR SUSPENSION OF THE	Written Consent
CONTRACT	
14	1.5.2, 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5,
	9.9.1, 9.10.2, 9.10.3, 11.4.1, 13.2, 13.4.2, 15.4.4.2
Tests and Inspections	Written Interpretations

Init.

1

AlA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

9

4.2.11, 4.2.12 Written Notice 2.3, 2.4, 3.3.1, 3.9, 3.12.9, 3.12.10, 5.2.1, 8.2.2, 9.7, 9.10, 10.2.2, 10.3, 11.1.3, 11.4.6, 12.2.2, 12.2.4, 13.3, 14, 15.4.1

Written Orders 1.1.1, 2.3, 3.9, 7, 8.2.2, 11.4.9, 12.1, 12.2, 13.5.2, 14.3.1, 15.1.2

(1950958438)

# ARTICLE 1 GENERAL PROVISIONS

#### § 1.1 BASIC DEFINITIONS

#### § 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement, Advertisement for Bids, Invitation & Instructions to Bidders, Conditions (General, Supplementary and Special Supplementary), Proposal, Surety Bond, Performance Bond, Labor and Material Payment Bond, Plans, Drawings, Specifications, Addenda, Bulletins, Supplemental Plans, Supplemental Specifications, and Standards & Specifications pursuant to Paragraph 1.5.4 below, and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract approved by the Owner and signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect.

#### § 1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor provided, however, Owner shall be a third party beneficiary of any Subcontract agreement under the circumstances set forth in Article 5 herein, (3) between the Owner and the Architect or the Architect's consultants or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

#### § 1.1.3 THE WORK

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

#### § 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by separate contractors.

#### § 1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

#### § 1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

#### (Paragraphs deleted)

- §1.1.7 "As selected", or "as approved", or "satisfactory to", or "as required", or "as directed" mean as selected, approved, required or directed by the Owner or by the Architect acting for the Owner, unless otherwise specified in the specific provision in which the term is used.
- "Final Completion" means the date the Contract has been fully performed, all the Work has been completed and a final Certificate for Payment approved by the Owner has been issued by the Architect.
- §1.1.9 "Indicated" and "shown" mean as described, detailed, discussed, scheduled, referenced, or called for in, or reasonably inferable from, the Contract Documents, in order to produce a first class Work product.
- "Or equal", "approved equal", or "equal to" mean that the determination whether an alternative product or system is equal to that indicated or specified shall be made by the Owner.
- §1.1.11 "Provide" or derivatives thereof means the Contractor shall properly fabricate, supply, furnish or procure all labor, materials, equipment, apparatus, and accessory appurtenances necessary to transport, deliver, install, erect and

construct the specified item, complete, in place, fully tested and ready for operation and use, including any final connections in strict accordance with the Drawings, Specifications and other Contract Documents. The words "Contractor shall" are implied and shall be so understood wherever the direction "provide" is used.

- §1.1.12 "Punch List Items" shall mean and shall be limited to uncompleted items of the Work (a) that do not interfere with the use and occupancy of any area of the Site for its intended purpose and (b) that, as a group, are capable of being completed by the Contractor within thirty (30) days of issuance of any Punch List. The "Punch List" is the list containing the Punch list items.
- \$1.1.13 "Substantial Completion" means the date that all of the Work has been completed to the point where it can be occupied and used for all purposes intended by Owner and bas been accepted by Owner and Architect as such, subject only to minor Punch List Items which do not affect full use and enjoyment," and the Contractor has completed all Work necessary in order for the Owner to receive all required occupancy permits.

#### § 1.1.14 THE PROJECT MANUAL

The Project Manual is a volume assembled for the Work which may include the bidding requirements, sample forms, Conditions of the Contract and Specifications.

# § 1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

§ 1.2.1 The Contract Documents include all items necessary for the proper execution and completion of the Work by the Contractor. The Work shall consist of all items specifically included in the Contract Documents as well as all additional items of work "indicated" in or from the Contract Documents. The Contract Documents are complementary, and what is required by any one Contract Document shall be as binding as if required by all. By its execution of the Contract, the Contractor shall represents, acknowledges and agrees that any differences between the requirements of the Drawings and the Specifications or any differences noted within the Drawings themselves or within the Specifications themselves or any conflicts between the Contract Documents and applicable standards, codes and ordinances, have been referred to the Owner and Architect by Contractor prior to submission of bids and have been clarified by an Addendum issued to all bidders.

If any such differences or conflicts were not called to the Owner's and Architect's attention prior to submission of bids, the Architect shall decide which of the conflicting requirements will govern based upon the most stringent or highest quality of the requirements, and, subject to the approval of the Owner, the Contractor shall perform the Work at no additional cost and/or time to the Owner in accordance with the Architect's decision. Work not indicated in the Contract Documents will not be required unless it is consistent therewith and is reasonably inferable therefrom as being necessary to produce the intended results.

Large scale drawings shall take precedence over small scale drawings; figured dimensions on drawings over scaled dimensions and noted material over graphic representations.

- § 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.
- § 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings. Unless specifically stated as an obligation of the Owner, it is understood to be an obligation of the Contractor.
- **§1.2.4** All Work shall conform to the Contract Documents. No change therefrom shall be made without review and written acceptance by Architect and Owner.

#### § 1.3 CAPITALIZATION

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles or (3) the titles of other documents published by the American Institute of Architects.

(Paragraphs deleted)

User Notes:

# § 1.4.1 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

#### § 1.4.2 HEADINGS.

The headings for each paragraph of the Contract Documents are for convenience and reference purposes only and in no way define, limit or describe the scope or intent of said paragraphs or of the Contract Documents nor in any way affect the Contract Documents.

# § 1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHERPROJECT DOCUMENTS

§ 1.5.1 The Contract Documents are owned and copyrighted by the Owner. No Contractor, Subcontractor or Materialman shall retain or claim ownership or copyright of Contract Documents. No Contractor, Subcontractor or Materialman shall use any Contract Document for another project. All Contractors and Subcontractors have a limited license to reproduce portions of Contract Documents for this project, which cannot be obtained from Architect or Owner.

The Architect and Owner shall be deemed the owners of the Project Documents, including the Drawings and Specifications, and will retain all common law, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Project Documents. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Architect or Owner's reserved rights.

§ 1.5.2 The Contractor hereby specifically acknowledges and declares, and the execution of the Contract by the Contractor is a representation of the Contractor that the Contract Documents are full and complete, are sufficient to have enabled the Contractor to determine the cost of the Work and that the Contract Documents are sufficient to enable it to construct the Work outlined therein, in accordance with applicable laws and regulations, and otherwise to fulfill all its obligations hereunder, including, but not limited to, Contractor's obligation to construct the Work for an amount not in excess of the Contract Sum on or before the date(s) of Substantial Completion established in the Agreement. The Contractor further acknowledges and declares that it has visited and examined the site, examined all physical and other conditions affecting the Work and is fully familiar with all of the conditions thereon and thereunder affecting the same. In connection therewith, Contractor specifically represents and warrants to Owner that it has, by careful examination, satisfied itself as to: (1) the nature, location, and character of the Project and the site, including, without limitation, surface conditions of the site and subsurface conditions readily observable or ascertainable upon the exercise reasonable diligence and all structures and obstructions thereon and thereunder, both natural and manmade, and all surface and subsurface water conditions of the site and the surrounding area; (2) the nature, location, and character of the general area in which the Project is located, including without limitation, its climatic conditions, available labor supply and labor costs, and available equipment supply and equipment costs; and (3) the quality and quantity of all materials, supplies, tools, equipment, labor, and professional services necessary to complete the Work in the manner and within the cost and time frame required by the Contract Documents. In connection with the foregoing, and having carefully examined all Contract Documents, as aforesaid, and having visited the site, the Contractor acknowledges and declares that it has no knowledge of any discrepancies, omissions, ambiguities, or conflicts in said Contract Documents and that if it becomes aware of any such discrepancies, omissions, ambiguities, or conflicts, it has an obligation to and will promptly notify Owner and Architect of such fact, and will not proceed until it shall have received the written interpretation of Owner or Architect.

Claims for additional compensation or extensions of time because of the failure of the Contractor to familiarize himself with the conditions enumerated in the Contract Documents and other conditions of the site which might affect the Work, or because of the failure of the Contractor to obtain a needed interpretation before proceeding with the Work, will not be allowed.

User Notes:

# § 1.6 TRANSMISSION OF DATA IN DIGITAL FORM

If the parties intend to transmit Instruments of Service or any other information or documentation in digital form, they shall endeavor to establish necessary protocols governing such transmissions, unless otherwise already provided in the Agreement or the Contract Documents.

#### ARTICLE 2 OWNER

#### § 2.1 GENERAL

§ 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization, excluding matters requiring approval by the Board of Park Commissioners or the Executive Director. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.

#### § 2.1.2 Intentionally omitted.

# § 2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER 8 2 2 1

Decision of the Owner. All work done under this Contract shall be done to the satisfaction of the Owner who shall in all cases determine the amount of work done which is to be paid for under this Contract. The Owner shall decide all questions that may arise as to the measurements of quantities and the fulfillment of this Contract on the part of the Contractor, and shall determine all questions concerning the true intent or meaning of the Plans and Specifications and his determination and decision shall be final and conclusive.

- § 2.2.2 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.
- § 2.2.3 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The furnishing of surveys by the Owner is not a guarantee of the accuracy of the information contained therein, and shall not relieve the Contractor from its duties under the Contract Documents in general. The submission of a bid for the work implies that the Contractor had examined the site, taking into consideration all such conditions that may affect the work, regardless of the information contained in the surveys.
- § 2.2.4 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services required to be furnished by Owner and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.
- § 2.2.5 Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.

#### § 2.3 OWNER'S RIGHT TO STOP THE WORK

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2, or fails or refuses to provide a sufficient amount of properly supervised and coordinated labor, materials, or equipment so as to be able to complete the Work within the Contract Time or fails to remove and discharge (within ten days) any lien filed upon Owner's funds by anyone claiming by, through, or under Contractor, or disregards the instructions of Architect or Owner when based on the requirements of the Contract Documents, or fails to carry out any portion of the Work in accordance with the Contract Documents, or in the event an emergency arises that requires the Work to be stopped, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated or the emergency no longer exists; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3. The Owner and Architect shall at all times have access to the Project Site for purposes of inspection whenever the Work is in preparation and progress.

(1950958438)

#### § 2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a seven-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, or fails within such seven-day period to eliminate (or diligently commence to eliminate) the cause of any stop work order issued under Section 2.3 thereof, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the actual cost of correcting such deficiencies, including Owner's expenses and compensation for the Architect's additional services made necessary by such default, neglect or failure. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

# §2.5 OWNER'S REMEDIES NOT EXCLUSIVE

The rights and remedies of Owner stated in this Article 2 shall be in addition to and not in limitation of, any other rights of the Owner granted in the Contract Documents or at law or in equity

# ARTICLE 3 CONTRACTOR

#### § 3.1 GENERAL

§ 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

It is understood and agreed that the relationship of Contractor to Owner shall be that of an independent contractor. Nothing contained herein or inferable herefrom shall be deemed or construed to (1) make Contractor the agent, servant, or employee of the Owner, or (2) create any partnership, joint venture, or other association between Owner and Contractor. Any direction or instruction by Owner in respect of the Work shall relate to the results the Owner desires to obtain from the Work, and shall in no way affect Contractor's independent contractor status as described herein.

- § 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents. High quality craftsmanship will be expected in all phases of work. Any elements found unacceptable and not in compliance with the Contract Documents will be removed and replaced by the Contractor until satisfactory results are obtained.
- § 3.1.3 The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

# § 3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

§ 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, evaluated the local conditions under which the Work is to be performed and, except for concealed conditions or those unknown to Contractor, found the site suitable for completing the Work in compliance with requirements of the Contract Documents. Furthermore, execution of the Contract by the Contractor is a representation that the Contract Documents include the construction details, means, methods, procedures and techniques necessary to perform the Work.

Before submitting its bid proposal the Contractor shall carefully study and compare the Contract Documents with each other and with information furnished by Owner pursuant to Section 2.2.2 and shall at once report to the Architect and Owner errors, inconsistencies or omissions discovered. The Contractor shall not be liable to the Owner or Architect for damage resulting from errors, inconsistencies or omissions in the Contract Documents that could not have been discovered by a reasonably diligent contractor experienced in performance of the type of work of the Work of this Project in advance of performance and that are not of the nature of items described in and intended to be covered in Sections 1.2.1 and 1.5.2 hereof, unless the Contractor recognized or reasonably should have recognized such error, inconsistency or omission and failed to report it to the Owner and Architect. If the Contractor performs any construction activity involving an error, inconsistency or omission in the Contract Documents that the Contractor recognized or reasonably should have recognized and of which Contractor failed to

**User Notes:** 

Init.

notify the Owner and Architect, the Contractor shall assume complete responsibility for such performance and shall bear the full amount of the attributable costs for correction.

- §3.2.1.1. If any errors, inconsistencies, or omissions in Contract Documents are recognized or reasonably should have been recognized by the Contractor, any employee or agent of the Contractor, or any of its Subcontractors, the Contractor shall be responsible for notifying the Owner and Architect in writing of such error, inconsistency, or omission before proceeding with the Work. The Architect will take such notice under advisement and within a reasonable time commensurate with job progress render an interpretation. The Architect's interpretation shall be subject to Owner's approval. If the Contractor fails to give such notice and proceeds with such Work, it shall correct any errors, inconsistencies, or omissions at no additional cost to the Owner.
- § 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall evaluate any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents, except to the extent that such errors, omissions or inconsistencies affect Contractor's services; however, the Contractor shall promptly report to the Architect and the Owner any errors, inconsistencies or omissions discovered by or made known to the Contractor. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.
- § 3.2.3 Both Architect and the Contractor are required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities. The Contractor shall promptly report to the Architect and the Owner any nonconformity discovered by or made known to the Contractor.
- § 3.2.4 If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner as would have been avoided, including any increases in construction costs, if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

#### § 3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

- § 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner and Architect and shall not proceed with that portion of the Work without further written instructions from the Owner and Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures.
- § 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.
- § 3.3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

- §3.3.4 The Contractor has the responsibility to ensure that all material suppliers and Subcontractors, their agents, and employees adhere to the Contract Documents, and that they order materials on time, taking into account the current market and delivery conditions and that they provide materials on time. The Contractor shall coordinate its Work with that of all others on the Project including deliveries, storage, installations, and construction utilities. The Contractor shall be responsible for the space requirements, locations, and routing of its equipment. In areas and locations where the proper and most effective space requirements, locations, and routing cannot be made as indicated, the Contractor shall meet with all others involved, before installation, to plan the most effective and efficient method of overall installation.
- §3.3.5 The Contractor shall establish and maintain bench marks and all other grades, lines and levels necessary for the Work, report errors or inconsistencies to the Owner and Architect before commencing Work, and review the placement of the building(s) and permanent facilities on the site with the Owner and Architect after all lines are staked out and before foundation Work is started. The Contractor shall provide access to the Work for the Owner, Architect, other persons designated by Owner, and governmental inspectors. Any encroachments made by the Contractor or its Subcontractors (of any tier) on adjacent properties due to construction as revealed by an improvement survey, except for encroachments arising from errors or omissions not reasonably discoverable by the Contractor in the Contract Documents shall be the sole responsibility of the Contractor, and the Contractor shall correct such encroachments within thirty (30) days of the improvement survey (or as soon thereafter as reasonably possible), at the Contractor's sole cost and expense, either by the removal of the encroachment (and subsequent reconstruction on the Project site) or agreement with the adjacent property owner(s) (in form and substance satisfactory to Owner in its sole discretion) allowing the encroachments to remain.

### § 3.4 LABOR AND MATERIALS

- § 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.
- § 3.4.2 The Contractor may make substitutions equivalent to or superior to the specified materials only with the consent of the Owner, after evaluation by the Architect and approval by the Owner and in accordance with a Change Order or Construction Change Directive.
- § 3.4.3 Except as provided in 3.4.2, the materials specified have been determined to have characteristics appropriate for the purposes of this Project. No work will be acceptable which utilizes an alternate not approved during the bidding process.
- § 3.4.4 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.
- §3.4.5 Before ordering any material or doing any Work, the Contractor shall verify all measurements at the Project Site and he shall be responsible for the correctness of same. No extra charge or compensation will be allowed to the Contractor on account of any difference between actual dimensions and the measurements shown by the Project Drawings.
- §3.4.6 The Contractor shall carefully inspect all materials delivered on and to the Project Site and reject defective materials without waiting for the Architect or other representative of Owner to observe the materials.
- §3.4.7 Contractor shall maintain harmonious labor relations on the job site. If a labor problem arises or any person employed by the Contractor on the Work shall appear to the Owner to be incompetent or conduct himself in a disorderly or improper manner, such person or persons shall be removed from the Work immediately on the request of the Owner. Said removal shall not create any additional cost to Owner and shall not extend the time for completion of the Work.

(1950958438)

User Notes:

#### § 3.5 WARRANTY

§3.5.1 The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. The warranty will not be affected by the specification of any product or procedure, unless the Contractor objects promptly to such product or procedures and advises the Architect and Owner of possible substitute products or procedures which will not affect the warranty. This warranty shall not be restricted by the limitations of any manufacturer's warranty. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, or improper operation. If required by the Architect or the Owner, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

All warranties shall include labor and materials and shall be signed by the manufacturer or Subcontractor as the case may be and countersigned by the Contractor. All warranties shall be addressed to the Owner and delivered to the Architect upon completion of the Work and before or with the submission of request for final payment. Except as otherwise provided in these Supplementary Conditions, elsewhere in the Contract Documents, or in any Certificate of Substantial Completion approved by Owner and Contractor and/or Subcontractor, as applicable, all warranties shall become effective on the date of Final Completion of the entire Work, and shall run for a twelve (12) month period, unless a longer period is provided for in the Contract Documents or by law. Where warranties overlap, the more stringent requirement shall govern.

- §3.5.2.1 If materials or equipment are replaced during the original warranty period, a new warranty period thereon shall then begin from the date that such corrective action is completed and approved.
- §3.5.2.2 For concrete work, warranty protection for a repaired item shall be for twenty-four months after final acceptance of concrete work or the length of the original warranty period, whichever is longer. This will cover structural failures, as well as surface erosion due to spalling caused by frost popping soft aggregates within the concrete and surface erosion due to faulty workmanship. All concrete work not meeting high industry standards will be removed and replaced at no charge to the Owner.
- §3.5.2.3 Defective materials, equipment or workmanship occurring within the Warranty period may be repaired where such produces results conforming to the Contract Documents relating to appearance, performance and reliability. Where the nature of the defective materials, equipment or workmanship is such that acceptable results cannot be obtained by repair, such defective items shall be removed and replaced with new materials, equipment or workmanship complying with the Contract Documents.
- §3.5.2.4 Correction of defective or non-conforming Work shall include, in addition to that described in Article 12, any damage to the Project or other property that may result from such defective or nonconforming Work or from such corrective action, including without limitation any damage to any contents, to the work of other contractors, or to adjacent property.
- §3.5.2.5 The Contractor shall furnish maintenance and twenty-four (24) hour callback service for the equipment provided by Contractor, Subcontractor or supplier for a period of at least six (6) months after Final Completion and acceptance of the Work, or for such longer period as shall otherwise be provided in any of the Contract Documents. This service shall include regular examinations of the installation by competent and trained employees of the Contractor, or manufacturer, and shall include all necessary adjustments, greasing, oiling, cleaning, supplies and parts to keep the equipment in proper operation except such parts made necessary by misuse, accidents or negligence not caused by the Contractor or any of its Subcontractors.
- §3.5.2.6 The warranty provided in this Paragraph 3.5 shall be in addition to and not in limitation of, any other warranty or remedy required under the Contract Documents or under applicable law.

# § 3.6 TAXES

**User Notes:** 

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled

init.

AlA Document A201TM - 2007. Copyright @ 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treatles. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

to go into effect. The Illinois State sales tax is not applicable to the Project with respect to materials, equipment and supplies incorporated in the Work or totally consumed in the performance of the Work. The Contractor shall pay unemployment and Social Security taxes and other taxes imposed by local, city, state or federal government with respect to Contractor's own personnel and certify to Owner that this has been done before payment is made to the Contractor.

#### § 3.7 PERMITS, FEES, NOTICES, AND COMPLIANCE WITH LAWS

- § 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded except as indicated in 2.2.2 herein.
- § 3.7.2 Contractor shall comply with and give notices required by laws, ordinances, codes, rules, regulations and lawful orders of public authorities having jurisdiction over all or any part of the Work or any insurance organizations (collectively, "Legal Requirements") relating to the Work of the performance thereof. If the Contractor fails to give any such notice or to permit any inspection, it shall be responsible for and shall indemnify and hold harmless, the Owner, the Architect and their respective consultants, employees, officers and agents against and from any resulting Work delays, fines, penalties, judgments or damages, including without limitation reasonable attorneys' fees and court costs imposed on, or suffered, sustained or incurred by, any of the parties indemnified hereunder. The Contractor shall also be liable to the Owner for any delay in the performance of the Work or increase in the cost of the Work resulting from the Contractor's failure to fully comply with the provisions of this Subparagraph 3.7.2. If the Contractor performs Work that it knows or reasonably should have known would be contrary to Legal Requirements, without such notice to the Architect and the Owner, the Contractor shall assume full responsibility for such Work and delays and shall bear the attributable costs.

#### (Paragraphs deleted)

- § 3.7.2.1 If the Contractor observes that portions of the Contract Documents are at variance with Legal Requirements, the Contractor shall promptly notify the Architect and Owner in writing, and necessary changes shall be accomplished by appropriate Modification.
- §3.7.2.2 Copies of any and all permits, licenses and certificates shall be delivered to the Architect and Owner as soon as they are obtained. The Contractor shall deliver the originals of such permits, licenses and certificates to the Architect together with the Contractor's application for final payment.
- §3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 7 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor in writing, stating the reasons and the Contract Sum or time for performance of the Work shall not be adjusted. If either party disputes this determination or recommendation, that party may proceed as provided in Article 15.
- § 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15.

(1950958438)

User Notes:

§ 3.7.6 Contractor shall comply with all public and private utility requirements relating to the Work or the performance thereof. If the Contractor performs Work contrary to applicable utility requirements, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

#### § 3.8 ALLOWANCES

- § 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct. § 3.8.2 Unless otherwise provided in the Contract Documents,
  - allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
  - .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
  - .3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.
- § 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

### § 3.9 SUPERINTENDENT

- § 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. The Contractor's competent superintendent shall have the knowledge and control of all work under this Contract and shall communicate directly to the Owner upon request.
- § 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner and Architect the name and qualifications of a proposed superintendent. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to the proposed superintendent or (2) that the Architect or Owner requires additional time to review. Failure of the Architect to reply within the 14 day period, or any extension of said time period, shall constitute notice of no reasonable objection.
- § 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall be which shall not unreasonably be withheld or delayed.

#### § 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

- § 3.10.1 The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall not exceed time limits current under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.
- § 3.10.2 The Contractor shall prepare a submittal schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit the schedule(s) for the Architect's and Owner's approval. The Architect's and Owner's approval shall not unreasonably be delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Owner and Architect reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.
- § 3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Architect.

User Notes:

- § 3.10.4 Construction Engineerting Check. The Contractor shall notify the Owner three (3) business days in advance of all grading, drainage, and other major items of construction for field checking of construction engineering. All questions pertaining to the Plans, Specifications and details of the work shall be directed to the Owner and cleared prior to construction.
- § 3.10.5 Contractor's Construction Schedule. The Contractor shall provide regular monitoring and updating of the Progress Schedule with monthly Update Reports submitted contemporaneously with the monthly pay application, or more frequently as required by the conditions of the Work. The update report shall indicate progress achieved and activities commenced or completed within the last month.

#### § 3.11 DOCUMENTS AND SAMPLES AT THE SITE

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Architect and Owner and shall be delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

#### § 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

- § 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- § 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.
- § 3.12.3 Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.
- § 3.12.4 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.
- § 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of separate contractors.
- § 3.12.6 By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- § 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved by the Owner and Architect.
- § 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Owner and the Architect in writing of such deviation at the time of submittal and (1) the Owner and Architect have given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility

for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Owner's or Architect's approval thereof.

- § 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such written notice, the Architect's approval of a resubmission shall not apply to such revisions.
- § 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner will and the Architect shall specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Owner and Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Architect has specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review, approve or take other appropriate action on submittals only for the purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

#### § 3.13 USE OF SITE

- § 3.13.1 The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment
- § 3.13.2 General Use. The Contractor shall enforce the Owner's instructions regarding the conduct and use of the site by his employees.
- § 3.13.3 Property Corners. Existing property corners on the site shall be replaced by a registered Land Surveyor at the Contractor's expense.

#### § 3.13.4 Parking & Traffic.

- .1 Parking of construction vehicles on the site by the Contractor shall not inhibit construction nor prevent access for emergency or other official vehicles. Parking of private vehicles on the site by the Contractor is prohibited unless said vehicle is necessary in the execution of the Contract. No construction vehicles shall be parked near or under any existing vegetation on the site.
- .2 Construction traffic and staging shall be permitted only within construction limits as indicated on plan. The contractor is responsible for repair of any areas disturbed outside of this area, including grading and sodding. No staging will be permitted on the existing asphalt.
- § 3.13.5 Fencing. The Contractor will be responsible for erecting and maintaining all construction fencing required by applicable law, regulation, rule, ordinance or code at all times of construction. Failure to erect or maintain this fencing will result in the correction of the problem by the Owner at the expense of the Contractor. The Contractor's expense will be back charged to the contract, and may include, but are not limited to, the cost of any materials and staff time. Required fencing must be installed and fully erected before construction operations beginning and tied-up at the end of each working day. All construction fencing must conform to the Specifications and as required by applicable law. specification.

**User Notes:** 

(1950958438)

§ 3.13.6 Water Removal. If, during construction, standing water caused by heavy rains or poor drainage becomes a hazard in the proper execution of the Contract, it shall be the responsibility of the Contractor to provide and make payment for removal of said water to existing drainage swales, storm sewers or other natural or man-made drainage ways.

# § 3.14 CUTTING AND PATCHING

- § 3.14.1 The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting and patching shall be restored to the condition existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents.
- § 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner or a separate contractor the Contractor's consent to cutting or otherwise altering the Work.

#### § 3.15 CLEANING UP

- § 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials from and about the Project.
- § 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Contractor.
- §3.15.3 The Contractor shall walk the site at the close of every work day to assure it is either free of waste material and rubbish, or the waste material and rubbish is secured in a container that is inaccessible to park patrons.

# § 3.16 ACCESS TO WORK

The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located.

#### § 3.17 ROYALTIES, PATENTS AND COPYRIGHTS

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if the Contractor knows or has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Owner and Architect.

#### § 3.18 INDEMNIFICATION

§ 3.18.1 To the fullest extent permitted by law, the Contractor shall waive all right of contribution and shall indemnify and hold harmless the Owner and the Architect and their officers, officials, employees, volunteers and agents from and against all claims, damages, losses and expenses, including but not limited to legal fees (attorney's and paralegals fees and court costs), arising out of or resulting from the performance of the Contractor's work, provided that any such claim, damage, loss or expense (i) is attributable to bodily injury, sickness, disease or death, or injury to or destruction of tangible property, other than the work itself, including the loss of use resulting therefrom and (ii) is caused in whole or in part by any wrongful or negligent act or omission of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, regardless of whether or not it is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or otherwise reduce any other right or obligation of indemnity which would otherwise exist as to any party or person described in this Paragraph. Contractor shall similarly protect, indemnify and hold and save harmless the Owner, its officers, officials, employees, volunteers and agents against and from any

Init.

and all claims, costs, causes, actions and expenses including but not limited to legal fees, incurred by reason of Contractor's breach of any of its obligations under, or Contractor's default of, any provision of the Contract.

In any and all claims against the Owner or the Architect or any of their agents or employees and consultants by any employee of any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the indemnification obligation under this Paragraph shall not be limited in anyway by any limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or any Subcontractor under workers' or workmen's compensation acts, disability benefit acts or other employee benefit

"Claims," "damages,"" losses" and "expenses" as these words are used in this Agreement shall be construed to include, but not limited to (1) injury or damage consequent upon the failure of or use or misuse by Contractor, its Subcontractors, agents, servants or employees, of any hoist, rigging, blocking, scaffolding, or any and all other kinds of items of equipment or other mechanical or structural contrivance erected or constructed by any person, or any or all other kinds of equipment whether or not the same be owned, furnished or loaned by Owner; (2) all attorneys' fees and costs incurred in bringing an action to enforce the provisions of this indemnity or any other indemnity contained herein; (3) (3) all costs and expenses incurred by the indemnified party and (4) error or omission or defect in any submission made to Architect / Architect for its approval or review.

- § 3.18.2 In claims against any person or entity indemnified under this Section 3.18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under Section 3.18.1 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts.
- The indemnification obligations of the Contractor under this Contract are limited only to the extent required under the Construction Contract Indemnification for Negligence Act (740 ILCS 35/0.01 et seq.).

# ARTICLE 4 ARCHITECT

#### § 4.1 GENERAL

- § 4.1.1 The Owner shall retain an architect lawfully licensed to practice architecture or an entity lawfully practicing architecture in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number.
- § 4.1.2 Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner and Architect. Consent shall not be unreasonably withheld.
- § 4.1.3 If the employment of the Architect is terminated, the Owner shall employ a successor architect as to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.

#### § 4.2 ADMINISTRATION OF THE CONTRACT

- § 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate for Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.
- § 4.2.2 The Architect as a representative of the Owner, will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, to supervise and to keep the Owner informed about the progress and quality of the portion of the Work completed, to endeavor to guard the Owner against defects and deficiencies in the Work, and to determine in general if the Work is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since

(1950958438)

these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.

§ 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and report to the Owner (1) known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, and (2) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of and will not be responsible for acts or omissions of the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

# § 4.2.4 COMMUNICATIONS FACILITATING CONTRACT ADMINISTRATION

Except as otherwise provided in the Contract Documents or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate with each other about matters arising out of or relating to the Contract. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and material suppliers shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.

- § 4.2.5 Based on the Architect's observations and evaluations of the of the progress and quality of Work and Contractor's Applications for Payment, the Architect will review and, after consultation with Owner, certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.
- § 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Owner and the Architect considers it necessary or advisable, the Owner or the Architect will have the responsibility and authority, subject to Owner's approval, to require inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Owner or the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Owner or the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Owner or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5 and 3.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- § 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Owner and Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4. All Change Orders, Construction Change Directives, and field directives shall require the written approval of Owner in order to be binding on Owner.
- § 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and approval, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.

- § 4.2.10 If the Owner and Architect agree, the Architect will provide one or more project representatives to assist in carrying out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.
- § 4.2.11 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.
- § 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable to the Contractor for results of interpretations or decisions rendered in good faith.

#### (Paragraph deleted)

§ 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Owner's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

#### ARTICLE 5 SUBCONTRACTORS

#### § 5.1 DEFINITIONS

- § 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor.
- § 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

# § 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

- § 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Architect may reply within 30 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to any such proposed person or entity or (2) that the Architect or Owner requires additional time for review. Failure of the Owner or Architect to reply within the 30 day period or any extension thereof shall constitute notice of no reasonable objection.
- § 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.
- § 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required.
- § 5.2.4 The Contractor shall not substitute a Subcontractor, person or entity previously selected if the Owner or Architect makes reasonable objection to such substitution.

**§5.2.5** In the event of a conflict between the Owner and Architect regarding the selection of Subcontractors, the Owner's decision shall govern.

#### § 5.3 SUBCONTRACTUAL RELATIONS

By appropriate written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

# § 5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS § 5.4.1

(Paragraphs deleted)

All subcontract agreements shall conform to the requirements of the Contract Documents and the Contractor hereby assigns to Owner (and Owner's permitted assigns) all its interest in any subcontract agreements and purchase orders now existing or hereinafter entered into by the Contractor for performance of any part of the Work, which assignment will be effective in the event of the Contractor's failure to perform the Work in accordance with the Contract Documents and upon acceptance by Owner in writing and only as to those subcontract agreements and purchase orders that Owner designates in said writing. It is agreed and understood that the Owner may accept said assignment at any time during the course of construction prior to Final Completion. Upon such acceptance by Owner, (1) the Contractor shall promptly furnish to the Owner true and correct copies of the designated subcontract agreements and purchase orders, and (2) the Owner shall only be required to compensate the designated Subcontractor(s) or supplier(s) for compensation accruing to such party(ies) for Work done or materials delivered from and after the date on which the Owner accepts the subcontract agreement(s) or purchase order(s). All sums due and owing by the Contractor to the designated Subcontractor(s) or supplier(s) for work performed or material supplied prior to Owner's acceptance of the subcontract agreement(s) or purchase order(s) shall constitute a debt between such parties and the Contractor. It is further agreed that no subcontract agreement or purchase order shall contain any restriction that would prohibit assignment under the terms and conditions stated hereinabove. It is further agreed and understood that such assignment is part of the consideration to Owner for entering into the Contract with the Contractor and may not be withdrawn prior to Final Completion.

- § 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for increases in cost resulting from the suspension.
- § 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity.

# ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS § 6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

§ 6.1.1 The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation. If the Contractor claims that delay or additional cost is involved because of such action by the Owner, the Contractor shall make such Claim as provided in Article 15.

(1950958438)

- § 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.
- § 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, separate contractors and the Owner until subsequently revised.
- § 6.1.4 Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces, the Owner shall be deemed to be subject to the same obligations and to have the same rights that apply to the Contractor under the Conditions of the Contract, including, without excluding others, those stated in Article 3, this Article 6 and Articles 10, 11 and 12.

#### § 6.2 MUTUAL RESPONSIBILITY

- § 6.2.1 The Contractor shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.
- § 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect and Owner apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.
- § 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for costs the Contractor incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction.
- § 6.2.4 The Contractor shall promptly remedy damage the Contractor causes to completed or partially completed construction or to property of the Owner, separate contractors as provided in Section 10.2.5.
- § 6.2.5 The Owner and each separate contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.
- **§6.2.6** Should the Contractor cause damage to the work or property of any separate contractor and/or in the event of any other claim, dispute, or matter in question between the Contractor and any separate contractor, the Contractor shall promptly attempt to settle with such other contractor by agreement, or otherwise to resolve the dispute. In any event, the Contractor shall indemnify, defend, and hold harmless the Owner, its officers, park commissioners, employees and agents, to the full extent as agreed to under Section 3.18 of these General Conditions.

# § 6.3 OWNER'S RIGHT TO CLEAN UP

If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and will allocate the cost among those responsible.

# ARTICLE 7 CHANGES IN THE WORK

#### § 7.1 GENERAL

§ 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.

- § 7.1.2 A Change Order shall be based upon agreement among the Owner, Architect and Contractor; a Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Architect alone.
- § 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Directive or order for a minor change in the Work.

#### § 7.2 CHANGE ORDERS

- § 7.2.1 A Change Order is a written instrument prepared and signed by the Owner, Architect and Contractor stating their agreement upon all of the following:
  - .1 The change in the Work;
  - .2 The amount of the adjustment, if any, in the Contract Sum; and
  - .3 The extent of the adjustment, if any, in the Contract Time.

Methods used in determining adjustments to the Contract sum may include those listed in Section 7.3.3. Notwithstanding any provision to the contrary in the Contract Documents, the overhead and profit for Contractor and Subcontractors on any Change Order or Construction Change Directive shall be determined as specified in Section 4.5 of the Agreement.

## § 7.3 CONSTRUCTION CHANGE DIRECTIVES

- § 7.3.1 A Construction Change Directive is a written order prepared and signed by the Architect and Owner and directed to the Contractor, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.
- § 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.
- § 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:
  - .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
  - .2 Unit prices stated in the Contract Documents or subsequently agreed upon;
  - .3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
  - .4 As provided in Section 7.3.7.

#### (Paragraph deleted)

- § 7.3.5 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.
- § 7.3.6 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.
- § 7.3.7 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect and Owner shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall be limited to the following:

**User Notes:** 

Init.

- .1 Costs of labor, including social security, old age and unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance;
- .2 Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- .3 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others:
- 4 Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to the Work; and
- .5 Additional costs of supervision and field office personnel directly attributable to the change.
- § 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.
- § 7.3.9 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.
- § 7.3.10 When the Owner and Contractor agree with a determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive.
- §7.3.11 No change in the Work, whether by way of alteration or addition to the Work, shall be the basis of an addition to the Contract Sum or a change in the Contract Time unless and until such alteration or addition has been authorized by a Change Order executed and issued in accordance with and in strict compliance with the requirements of the Contract Documents and applicable law. Accordingly, no course of conduct or dealing between the parties, nor any express or implied acceptance of alterations or additions to the Work and no claim that the Owner has been unjustly enriched shall be the basis of any claim to an increase in the Contract Sum or change in the Contract Time.

#### § 7.4 MINOR CHANGES IN THE WORK

The Architect has authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order signed by the Architect and shall be binding on the Owner and Contractor.

# §7.5 CONTINUATION OF WORK PENDING RESOLUTION

Pending final determination of cost to the Owner or extension of time to the Contractor, unless otherwise directed by Owner, Contractor shall continue to perform the Work in accordance with the Contract Documents.

# ARTICLE 8 TIME

#### § 8.1 DEFINITIONS

- § 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.
- § 8.1.2 The date of commencement of the Work is the date established in the Agreement.
- § 8.1.3 The date of Substantial Completion is the date certified by the Architect and approved by Owner in accordance with Section 9.8.
- § 8.1.4 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

User Notes:

lnit.

1

#### § 8.2 PROGRESS AND COMPLETION

- § 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for performing the Work.
- § 8.2.2 The Contractor shall not, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.
- § 8.2.3 The Contractor shall carry the Work forward regularly, diligently, uninterruptedly and expeditiously and in a good workmanlike and professional manner at such a rate of progress and with an adequate work force as will insure the completion of the Work in accordance with the Contract Documents by the date established in the Contract. It is expressly understood and agreed by and between Contractor and Owner that the time for completion of the Work is a reasonable time, taking into consideration the average climatic range, usual industrial conditions, and all other conditions and factors prevailing in this locality.

#### § 8.3 DELAYS AND EXTENSIONS OF TIME

- § 8.3.1 If the Contractor is delayed at any time in the progress of the Work prior to Substantial Completion by: any wrongful act or neglect of Owner; changes ordered in the Work which are not caused by the wrongful or negligent acts, errors or omissions of Contractor, its agents, employees or Subcontractors; or by occurrences beyond the control and without the fault or negligence of the Contractor and which by the exercise of reasonable diligence the Contractor is unable to prevent or provide against, including regional labor disputes (not disputes limited to the work force of, or provided by, the Contractor or its Subcontractors) as they affect the Work that cannot be resolved by Contractor's agreeing to the wages, hours, working conditions and other terms as they have been or will be established as the pattern settlement with respect to said dispute, provided that prior to execution of the Contract by Owner, Contractor has advised Owner in writing of the expiration during the Contract Time of applicable labor contracts; fire, unusual delay in deliveries not reasonably foreseeable, unavoidable casualties, adverse weather conditions not reasonably foreseeable, unusual subsurface or concealed Site conditions which could not have been discovered in the exercise of due diligence and which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the type contemplated in the Contract Documents, or by other occurrences which the Architect, subject to the Owner's approval, determines may justify delay, then, provided that the Contractor is in compliance with all other relevant provisions of the Contract Documents, the Contract Time shall be extended by Change Order or Construction Change Directive for the length of time actually and directly caused by such occurrence as determined by the Architect and approved by the Contractor and Owner; provided, however, that such extension of Contract Time shall be net of any delays caused by or due to the fault or negligence of the Contractor or which are otherwise the responsibility of the Contractor and shall also be net of any contingency or "float" time allowance included in the Contractor's construction schedule. The Contractor shall, in the event of any occurrence likely to cause a delay, cooperate in good faith with the Architect and Owner to minimize and mitigate the impact of any such occurrence and do all things reasonable under the circumstances to achieve this goal.
- § 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.
- § 8.3.3 Extension of time provided for the completion of the Work shall be the Contractor's sole remedy for delay (except for the Contractor's right to terminate the Contract pursuant to the provisions of Article 14 hereof), unless the same shall have been caused by acts constituting intentional interference by Owner with Contractor's performance of the Work and where to the extent that such acts of the Owner continue after Contractor's written notice to Owner of such interference. The Owner's exercise of any of its rights under the Contract, including, without limitation, its rights under Article 7, 'Changes in the Work,' regardless of the extent or number of such Changes, or the Owner's exercise of any of its remedies of suspension of the Work, or requirement of correction or re-execution of any defective Work, shall not under any circumstances be construed as intentional interference with Contractor's performance of the Work.

# ARTICLE 9 PAYMENTS AND COMPLETION

## § 9.1 CONTRACT SUM

The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

#### § 9.2 SCHEDULE OF VALUES

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit to the Architect and Owner, before the first Application for Payment, a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Architect and Owner may require. This schedule, unless objected to by the Architect or Owner, shall be used as a basis for reviewing the Contractor's Applications for Payment.

#### § 9.3 APPLICATIONS FOR PAYMENT

- § 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2., for completed portions of the Work. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents. Until Final Completion, the Owner will pay ninety percent (90%) of the amount due the Contractor on account of progress payments. No interest will be paid on retention amounts. Provided, however, that after the Work is seventy percent (70%) complete, Owner may, without reduction of previous retainage, determine to pay Contractor remaining progress payments for each work category in full.
- § 9.3.1.1 As provided in Section 7.3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders.
- § 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.
- §9.3.1.3 The following forms must be used for pay requests: (1) AIA Application & Certificate for Payment (G702 & G703); and (2) a Sworn Statement from Contractor and Subcontractor to Owner.
- **§9.3.1.4** When the contract work has been awarded on a unit price bid basis, the form of each application shall follow the Bid Proposal Form, listing each item number, the total quantity of units completed to date of the estimate, the unit price and subtotal. The subtotal column shall be added to show the total cost of work completed to date, less ten (10%) percent to be withheld giving the total amount requested for payment. Previous applications for payment paid by the Owner shall be shown on each subsequent request and subtracted after the ten (10%) percent has been withheld.
- § 9.3.1.5 Failure to supply waivers of lien or acceptable evidence of payment of all current accounts incurred by this Contract work will be considered grounds for withholding final payment.
- § 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.
- § 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims,

security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

#### § 9.4 CERTIFICATES FOR PAYMENT

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, and subject to Owner's approval, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Section 9.5.1.

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, and not Contractor, based on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation to the Owner and not the Contractor, that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§9.4.3 All applications for payment shall be accompanied by affidavits, in triplicate, from the Contractor and Subcontractors containing such information and in such form as to comply with the Illinois Mechanics Lien Act (770 ILCS 60/0.01 et seq.) and showing in detail the sources of all labor and materials used and contracted to be used on the Project, including names and addresses of subcontractors and material suppliers; amounts paid and remaining to be paid to each; lien waivers in a form acceptable to Owner; together with all other documents as shall be necessary, in the sole judgment of the Architect and Owner, to waive all claims of liens to date and comply with all applicable state and local laws.

# § 9.5 DECISIONS TO WITHHOLD CERTIFICATION

§ 9.5.1 The Architect, after consultation with Owner, may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of, but not limited to:

defective Work not remedied; .1

Init.

- .2 third party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Contractor;
- .3 failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment:
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- damage to the Owner or a separate contractor,
- reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay;
- .7 repeated failure to carry out the Work in accordance with the Contract Documents.

§ 9.5.2 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

(1950958438)

§ 9.5.3 If the Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Architect will reflect such payment on the next Certificate for Payment.

§9.5.4 No interest will be paid on payments withheld.

#### § 9.6 PROGRESS PAYMENTS

- § 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents and shall so notify the Architect.
- § 9.6.2 The Contractor shall pay each Subcontractor no later than seven days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.
- § 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.
- § 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. The Owner shall not have an obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law.
- § 9.6.5 Contractor payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.
- § 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.
- § 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors and suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust or shall entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.
- **§9.6.8** Anything to the contrary contained or implied herein notwithstanding, no progress payment need be made by Owner until such time as Contractor, Subcontractors or any other persons performing the Work or furnishing materials or equipment for the Project furnishes such documents as Owner may reasonably require (including without limitation sworn notarized contractor's statements, affidavits and waivers of lien).

#### § 9.7 FAILURE OF PAYMENT

If the Architect does not issue a recommendation for a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner either does not reject the Architect's recommendations for the Certificate of Payment or does not pay the Contractor within 10 days after the date established in the Contract Documents the amount accepted for payment by the Owner, then the Contractor may, upon seven additional days' written notice to the Owner and Architect stop the Work. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs incurred of shutdown, and start-up, plus interest as provided for in the Contract Documents.

(1950958438)

#### § 9.8 SUBSTANTIAL COMPLETION

- § 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use.
- § 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect and Owner a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.
- § 9.8.3 Upon receipt of the Contractor's list, the Owner and Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Owner's and Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Owner and the Architect to determine Substantial Completion.
- § 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Final Completion of the Work or designated portion thereof unless otherwise provided in the Contract Documents or the Certificate of Substantial Completion.
- § 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. Upon such acceptance and consent of surety, if any, the Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

# § 9.9 PARTIAL OCCUPANCY OR USE

- § 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect and Owner as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.
- § 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.
- § 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

#### § 9.10 FINAL COMPLETION AND FINAL PAYMENT

§ 9.10.1 Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect find the Work acceptable under the Contract Documents and the Contract fully performed, and

the Architect has advised Owner of that finding and Owner has not advised Architect of any objection to such finding, the Architect will promptly issue a final Certificate for Payment stating that to the best of Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation to Owner and not Contractor, that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled. Owner's failure to object to and Owner's acceptance of Architect's findings hereunder shall not limit Architect's obligation to properly perform his duties under the Contract Documents and shall not constitute Owner's acceptance of Work not complying with the requirements of the Contract Documents or Owner's waiver of any claims or remedies it may have with respect to any such defective or delayed Work.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) consent of surety, if any, to final payment and (5), if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

§ 9.10.3 The final payment by Owner shall not relieve the Contractor of the responsibility for the correction of any and all defects in the work performed. Contractor shall correct all defects as notified for the applicable warranty period after final payment.

#### § 9.10.4

(Paragraphs deleted)

Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

(Paragraph deleted)

# ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY § 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.

#### § 10.2 SAFETY OF PERSONS AND PROPERTY

§ 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to

- .1 employees on the Work and other persons who may be affected thereby;
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.

§ 10.2.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their

User Notes:

init.

protection from damage, injury or loss. Contractor shall be responsible for security of all tools, materials, and equipment left on site.

- § 10.2.3 The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.
- § 10.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, such activities shall only be done with written consent of Owner and the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- § 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose wrongful acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.
- § 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect. The person designated as responsible for prevention of accidents shall hold regularly scheduled meetings with representatives of Subcontractors, and in the event of separate contracts, hold meetings with other contractors, to promote compliance with governing safety regulations.
- § 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

#### § 10.2.8 INJURY OR DAMAGE TO PERSON OR PROPERTY

If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

#### § 10.3 HAZARDOUS MATERIALS

- § 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner and Architect in writing.
- § 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or the Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor has no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Contractor's reasonable additional costs of shut-down, and start-up.

- § 10.3.3 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's consultants and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking indemnity.
- § 10.3.4 The Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling of such materials or substances.
- § 10.3.5 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.
- § 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall indemnify the Contractor for all cost and expense thereby incurred.

#### § 10.4 EMERGENCIES

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

#### ARTICLE 11 INSURANCE AND BONDS

(Paragraphs deleted)

- §11.1 CONTRACTOR'S INSURANCE REQUIREMENTS. Contractor shall procure and maintain for the duration of the contract, insurance against claims for death, injuries, sickness to persons, or damages to property which may arise from or in connection with the performance of work hereunder by the Contractor, his agents, representatives, employees or subcontractors, anyone directly or indirectly employed by them or anyone for whose acts any of them may be liable, of the types and in the amounts listed below.
- §11.1.1 Commercial General and Umbrella Liability Insurance. Contractor shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less that \$1,000,000 each occurrence. If such CGL insurance contains a general aggregate limit, it shall apply separately to this project/location. CGL insurance shall be written on Insurance Services Office (ISO) occurrence form CG 00 01, or a substitute form providing equivalent coverage, and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract (including the tort liability of another assumed in a business contract). Owner, its elected and appointed officials, employees, agents and volunteers, and Architect shall be included as an insured under the CGL, using ISO additional insured endorsement CG 20 10 or a substitute providing equivalent coverage, and under the commercial umbrella, if any. This insurance shall apply as primary insurance with respect to any other insurance or self-insurance afforded to Owner and Architect. There shall be no endorsement or modification of the CGL limiting the scope of coverage for liability arising from pollution, explosion, collapse, or underground property damage.
- §11.1.2 Continuing Completed Operations Liability Insurance. Contractor shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella liability insurance with a limit of not less than \$1,000,000 each occurrence for at least three years following substantial completion of the work. Continuing CGL insurance shall be written on ISO occurrence form CG 00 01, or substitute form providing equivalent coverage, and shall, at minimum, cover liability arising from products-completed operations and liability assumed under an insured

(1950958438)

contract. Continuing CGL insurance shall have a products-completed operations aggregate of at least two times its each occurrence limit. Continuing commercial umbrella coverage, if any, shall include liability coverage for damage to the insured's completed work equivalent to that provided under ISO form CG 00 01.

- §11.1.3 Business Auto and Umbrella Liability Insurance. Contractor shall maintain business auto liability and, if necessary, commercial umbrella liability insurance with a limit of not less than \$1,000,000 each accident. Such insurance shall cover liability arising out of any auto including owned, hired and non-owned autos. Business auto insurance shall be written on Insurance Services Office (ISO) form CA 00 01, CA 00 05, CA 00 12, CA 00 20, or a substitute form providing equivalent liability coverage. If necessary, the policy shall be endorsed to provide contractual liability coverage equivalent to that provided in the 1990 and later editions of CA 00 01.
- §11.1.4 Workers Compensation Insurance. Contractor shall maintain workers compensation as required by statute and employers liability insurance. The commercial umbrella and/or employers liability limits shall not be less than \$1,000,000 each accident for bodily injury by accident or \$1,000,000 each employee for bodily injury by disease. If Owner has not been included as an insured under the CGL using ISO additional insured endorsement CG 20 10 under the Commercial General and Umbrella Liability Insurance required in this Contract, the Contractor waives all rights against Owner and its officers, officials, employees, volunteers and agents for recovery of damages arising out of or incident to the Contractors work.
- §11.1.5 Contractor's Obligation to Insure for Bodily Injury Claims. In addition to the above, the Owner will require all Contractor's to purchase insurance to cover claims and expenses asserted against Architect, its employees and consultants for bodily injury, sickness, disease, or death cause by any negligent act or omission of the Contractor, any Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts any of them may be liable.

#### §11.1.6 General Insurance Provisions

- .1 Evidence of Insurance Prior to beginning work, Contractor shall furnish Owner with a certificate(s) of insurance and applicable policy endorsement(s), executed by a duly authorized representative of each insurer, showing compliance with the insurance requirements set forth above. All certificates shall provide for 30 days written notice to Owner prior to the cancellation or material change of any insurance referred to therein. Written notice to Owner shall be by certified mail, return receipt requested. Failure of Owner to demand such certificate, endorsement or other evidence of full compliance with these insurance requirements or failure of Owner to identify a deficiency from evidence that is provided shall not be construed as a waiver of Contractor's obligation to maintain such insurance. Owner shall have the right, but not the obligation, of prohibiting Contractor or any subcontractor from entering the project site until such certificates or other evidence that insurance has been placed in complete compliance with these requirements is received and approved by Owner. Failure to maintain the required insurance may result in termination of this Contract at Owner's option. With respect to insurance maintained after final payment in compliance with a requirement above, an additional certificate(s) evidencing such coverage shall be promptly provided to Owner whenever requested. Contractor shall provide certified copies of all insurance policies required above within 10 days of Owner's written request for said copies.
- . 2 Acceptability of Insurers. For insurance companies which obtain a rating from A.M. Best, that rating should be no less than A VII using the most recent edition of the A.M. Bests Key Rating Guide. If the Bests rating is less than A VII or a Best's rating is not obtained, the Owner has the right to reject insurance written by an insurer it deems unacceptable.
- .3 Cross-Liability Coverage. If Contractor's liability policies do not contain the standard ISO separation of insureds provision, or a substantially similar clause, they shall be endorsed to provide cross-liability coverage.
- .4 Deductibles and Self-Insured Retentions. Any deductibles or self-insured retentions must be declared to the Owner. At the option of the Owner, the Contractor may be asked to eliminate such deductibles or self-insured retentions as respects the Owner, its officers, officials, employees, volunteers and agents or required to procure a bond guaranteeing payment of losses and other related costs including but not limited to investigations, claim administration and defense expenses.

1

.5 Subcontractors. Contractor shall cause each subcontractor employed by Contractor to purchase and maintain insurance of the type specified above. When requested by the Owner, Contractor shall furnish copies of certificates of insurance evidencing coverage for each subcontractor.

# §11.2 PERFORMANCE AND PAYMENT BONDS

- § 11.2.1 The Contractor shall deposit with the Owner before commencing any Work an AIA A312-2010 Performance Bond and Payment Bond for 110% of the Contract Sum, guaranteeing the faithful performance of the work in accordance with the Contract, the payment of all indebtedness incurred for labor and materials, payment of the prevailing wage in accordance with paragraph 13.8.1, and guarantee correction of work. The Surety must be approved by the Owner, and be licensed to conduct business in the State of Illinois and be named in the current list of "Surety Companies Acceptable on Federal Bonds" as published in the Federal Register by the Audit Staff Bureau of Accounts, U.S. Treasury.
- § 11.2.2 The Contractor and all subcontractors shall name the Owner as an obligee on all bonds.

#### § 11.3 PROPERTY INSURANCE

- § 11.3.1 Unless otherwise provided, the Owner shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance in the amount of the initial Contract Sum as well as subsequent modifications thereto for the entire Work at the site on a replacement cost basis without voluntary deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Paragraph 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Paragraph 11.3 to be covered, whichever is earlier. This insurance shall include the respective interests of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Work.
- § 11.3.1.1 Property insurance shall be on a course of construction policy form and shall insure against the perils of fire and extended coverage and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, false work, windstorm, testing and start-up, temporary buildings and debris removal, including demolition, and shall cover reasonable compensation for the Architect's, any of the Owner's Consultant's services and expenses required as a result of such insured loss. Owner shall not be required to provide coverage for other perils unless otherwise provided in the Contract Documents. Property insurance provided by the Owner shall not cover Contractor's, Subcontractor's or Sub-subcontractor's liability or any tools, apparatus, machinery, scaffolding, hoists, forms, staging, shoring or other similar items commonly referred to as construction equipment, which may be on the site and the capital value of which is not included in the Work. The Contractor shall make his own arrangements for any insurance he may require on such construction equipment.
- § 11.3.1.2 The Contractor shall effect insurance which will protect the interests of the Contractor, Subcontractors and Sub-subcontractors in the Work.
- § 11.3.1.3 If the property insurance requires minimum deductibles and such deductibles are identified in the Contract Documents, the Contractor shall pay costs not covered because of deductibles. If the Owner or insurer increases the required minimum deductibles above the amounts so identified or if the Owner elects to purchase this insurance with voluntary deductible amounts, the Owner shall be responsible for payment of the additional costs not covered because of such increased or voluntary deductibles. If deductibles are not identified in the Contract Documents, the Owner shall pay costs not covered because of deductibles.
- § 11.3.1.4 Unless otherwise provided in the Contract Documents, this property insurance shall cover portions of the Work stored off the site and paid for by Owner after written approval of the Owner at the value established in the approval, and also portions of the Work in transit and paid for by Owner.
- § 11.3.1.5 Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse or reduction of insurance.

#### § 11.3.2 BOILER AND MACHINERY INSURANCE

The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner; this insurance shall include interests of the Owner, Contractor Subcontractors and Sub-subcontractors in the Work, and the Owner and the Contractor shall be named insureds.

### (Paragraphs deleted)

§ 11.3.4 If the Contractor requests in writing that insurance for risks other than those described herein or other special causes of loss be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by appropriate Change Order.

# (Paragraphs deleted)

§11.3.11 Notwithstanding any provision contained in Section 11.3 including paragraphs 11.3.1 through and including 11.3.11, the Owner's obligation to purchase insurance shall herein be deemed satisfied by the Owner's membership in a self-insured risk management agency or pool. The Contractor agrees that any obligation the Owner has to purchase property insurance shall be satisfied by the Owner's membership in a self-insured risk management agency or pool. The Contractor further agrees that it will only have rights allowable to it under any coverage provided through the Owner's membership in a self-insured risk management agency or pool.

## ARTICLE 12 UNCOVERING AND CORRECTION OF WORK § 12.1 UNCOVERING OF WORK

§ 12.1.1 If a portion of the Work is covered contrary to the Owner's or Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Owner or the Architect, be uncovered for the Owner's and Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.

§ 12.1.2 If a portion of the Work has been covered that the Owner or the Architect has not specifically requested to examine prior to its being covered, the Architect, with the consent of the Owner, may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

## § 12.2 CORRECTION OF WORK

init.

1

# § 12.2.1 BEFORE OR AFTER SUBSTANTIAL COMPLETION

The Contractor shall promptly correct Work rejected by the Owner or Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

# § 12.2.2 AFTER SUBSTANTIAL COMPLETION

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so. The Owner shall give such notice promptly after discovery of the condition. During the applicable period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.4. Notwithstanding the foregoing, Contractor shall correct Work deficiently or defectively performed, and replace defective or nonconforming materials, even though such deficiency, defect or nonconformity may be discovered more than one year after final completion, if the correction is of a latent defect and arises from poor workmanship or improper materials or is required to be made to

AIA Document A201™ - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorize reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale. User Notes:

(1950958438)

workmanship or materials covered by Contractor or Subcontractors contrary to the Architect's request or to requirements specifically expressed in the Contract Documents and was therefore not visible for inspection by Architect or Owner at the time the Work was performed.

§ 12.2.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

### (Paragraph deleted)

- § 12.2.3 The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.
- § 12.2.4 The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.
- § 12.25 Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

#### § 12.3 ACCEPTANCE OF NONCONFORMING WORK

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

# ARTICLE 13 MISCELLANEOUS PROVISIONS

#### § 13.1 GOVERNING LAW

The Contract shall be governed by the law of the place where the Project is located.

#### § 13.2 SUCCESSORS AND ASSIGNS

- § 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to covenants, agreements and obligations contained in the Contract Documents. This Contract is nonassignable in whole or in part by Contractor, and an assignment shall be void without the prior written consent of Owner, which consent, shall not be unreasonably withheld.
- § 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate such assignment.

# § 13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to, the last business address known to the party giving notice.

#### § 13.4 RIGHTS AND REMEDIES

- § 13.4.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.
- § 13.4.2 No action or failure to act by the Owner, Architect or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach there under, except as may be specifically agreed in writing.

User Notes:

1

#### § 13.5 TESTS AND INSPECTIONS

- § 13.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Owner and Architect timely notice of when and where tests and inspections are to be made so that the Owner and Architect may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor.
- § 13.5.2 If the Owner, Architect, or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Section 13.5.1, the Owner or Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Owner and Architect of when and where tests and inspections are to be made so that the Owner and Architect may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.
- § 13.5.3 If such procedures for testing, inspection or approval under Sections 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure including those of repeated procedures and compensation for the Architect's services and expenses shall be at the Contractor's expense.
- § 13.5.4 Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Owner and Architect.
- § 13.5.5 If the Architect is to observe tests, inspections or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.
- § 13.5.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.
- § 13.5.7 Retests. The cost of a retest will be borne by the party requesting the retest, unless the retest shows that the original test or the Work being tested was in error or defective, and in such event, the cost of the retest shall be borne by the other party.

#### § 13.6 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at the legal rate established in the Illinois Local Governmental Prompt Payment Act.

#### § 13.7 TIME LIMITS ON CLAIMS

The Owner and Contractor shall commence all claims and causes of action, whether in contract, tort, breach of warranty or otherwise, against the other arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in the Agreement within the time period specified by applicable law.

#### § 13.8. COMPLIANCE WITH LAWS

43

Contractor shall abide by and comply with all applicable Federal, State and local laws and rules and regulations including without limitation those relating to 1) fair employment practices, affirmative action and prohibiting discrimination in employment; 2) workers' compensation; 3) workplace safety; 4) wages and claims of laborers, mechanics and other workers, agents, or servants in any manner employed in connection with contracts involving public funds or the development or construction of public works, buildings or facilities; and 5) steel products procurement.

§ 13.8.1 As a condition of the award of the Contract to contractor, Contractor shall certify, affirm and agree as follows, which certifications, affirmations and agreements shall be incorporated in and hereunder as a part of the Contract:

.1 The Contractor shall comply with the requirements of the Illinois Prevailing Wage Act (820 ILCS 130/0.01 et seq.) and the Owner's Ordinances requiring payment of prevailing wages. The Contractor shall pay or cause to be paid not less than the prevailing rate of hourly wage in the county the work is performed as determined by the Illinois Department of Labor for the month in which the work is performed including but not limited to all laborers, workers and mechanics. All contractors and subcontractors rendering services under this contract must comply with all requirements under the Act, including but not limited to, all wage, notice and record keeping duties.

The Contractor is required to verify current prevailing wage prior to the first day of each month and to pay the then-current prevailing wage rate as determined by the Illinois Department of Labor, regardless of the rates contained in the Contract Documents. Any increases in costs to the Contractor due to the changes in the prevailing wage during the term of this Contract shall be at the expense of Contractor and not at the expense of Owner. Current prevailing wage rates are published at the following website: <a href="https://www.illinois.gov/idol/Laws-Rules/CONMED/Rates/2015/july/COUNTY.HTM">https://www.illinois.gov/idol/Laws-Rules/CONMED/Rates/2015/july/COUNTY.HTM</a>. The Contractor agrees to indemnify and hold harmless the Owner for any violations of the Prevailing Wage Act.

The Contractor shall also: (1) insert into each subcontract and the project specifications for each subcontract, a written stipulation that the subcontractor shall not pay less than the prevailing rate of hourly wage to all laborers, workers, and mechanics performing work under the contract; and (2) require each subcontractor to insert into each lower-tiered contract and the project specifications for each lower-tiered subcontract, a stipulation that the subcontractor shall not pay less than prevailing rate of hourly wage to all laborers, workers, and mechanics performing work under the contract.

The Contractor shall include on all bonds and shall cause all subcontractors' bonds required under the Contract Documents to guarantee compliance with the Prevailing Wage Act.

Additionally, the Contractor and each subcontractor shall make and keep, for a period of not less than five (5) years from the date of the last payment on a contract or subcontract, records of all laborers, mechanics, and other workers employed by them on the Project; the records shall include each worker's name, address, telephone number when available, social security number, classification or classifications, the hourly wages paid in each pay period, the number of hours worked each day, and the starting and ending times of work each day. The Contractor shall submit monthly, no later than the 10th day of each calendar month, in person, by mail, or electronically a certified payroll to the Owner with each monthly pay request in the form attached to the Contract Documents. The certified payroll shall be accompanied by a statement signed by the Contractor or subcontractor which states that: (i) he or she has examined the certified payroll and such records are true and accurate; (ii) the hourly rate paid to each worker is not less than the general prevailing rate of hourly wages required by this Act; and (iii) the Contractor or subcontractor is aware that filing a certified payroll that he or she knows to be false is a Class A misdemeanor. The Contractor may rely on the certification of a lower tier subcontractor, provided the Contractor does not knowingly rely upon a subcontractor's false certification. The records submitted in accordance with this payroll submittal provision shall be considered public records pursuant to Section 5 of the Prevailing Wage Act, 820 ILCS 130/5 (2004, as amended by P.A. 94-515). The Owner may, at its option, immediately terminate the Contract in the event that Contractor violates any provision of this paragraph or the Prevailing Wage Act.

Contractor shall also post the prevailing wage rates for each craft or type of worker or mechanic needed to complete the project at either: (1) a location on the project site easily accessible to the workers engaged on the project; or (2) in lieu of posting on the project site, if the Contractor has a business location where

laborers, workers, ands mechanics may regularly visit, the Contractor may either post the prevailing rate of wages in each county the Contractor works in a conspicuous location or provide the laborers, workers or mechanics engaged on the project a written notice indicating the prevailing rate of wages for the project.

Upon seven business days notice, the Contractor and each subcontractor shall make available for inspection and copying at a location within this State during reasonable hours, the records identified in 820 ILCS 130/5(a)(1) to the Owner, and its officers and agents.

- To the best of Contractor's knowledge, no officer or employee of Contractor has been convicted of bribery or attempting to bribe an officer or employee of the State of Illinois, or any unit of local government, nor has any officer or employee made an admission of guilt of such conduct which is a matter of record.
- Contractor is not barred from bidding on or entering into public contracts due to having .3 been convicted of bid-rigging or bid rotating under paragraphs 33E-3 or 33E-4 of the Illinois Criminal Code. Contractor also certifies that no officers or employees of the Contractor have been so convicted and that Contractor is not the successor company or a new company created by the officers or owners of one so convicted. Contractor further certifies that any such conviction occurring after the date of this certification will be reported to the Owner, immediately in writing, if it occurs during the bidding process, or otherwise prior to entering into the Contract therewith.
- Pursuant to the Illinois Human Rights Act (775 ILCS 5/2-105), Contractor has a written sexual harassment policy that includes, at a minimum, the following information: (i) a statement on the illegality of sexual harassment; (ii) the definition of sexual harassment under State law; (iii) a description of sexual harassment utilizing examples; (iv) the Contractor's internal complaint process including penalties; (v) the legal recourse, investigative and complaint process available through the Illinois Department of Human Rights and the Human Rights Commission and directions on how to contact both; and (vi) protection against retaliation as provided by Section 6-101 of the Illinois Human Rights Act. Contractor further certifies that such policy shall remain in full force and effect. A copy of the policy shall be provided to the Illinois Department of Human Rights upon request.
- Contractor shall abide by the "Employment of Illinois Workers on Public Works Act" (30) ILCS 570/0.01 et. seq.) which stipulates that whenever there is a period of excessive unemployment in Illinois, defined as any month immediately following two (2) consecutive calendar months during which the level of unemployment in Illinois exceeds five percent (5%) as measured by the U.S. Bureau of Labor Statistics in its monthly publication of employment and unemployment figures, the Contractor shall employ only Illinois laborers unless otherwise exempted as so stated in the Act. ("Illinois laborer" means any person who has resided in Illinois for at least 30 days and intends to become or remain an Illinois resident). Other laborers may be used if Illinois laborers are not available or are incapable of performing the particular type of work involved if so certified by the Contractor and approved by the Architect.
- (i) Contractor's bid proposal was made without any connection or common interest in the profits anticipated to be derived from the Contract by Contractor with any other persons submitting any bid or proposal for the Contract; (ii) the Contract terms are in all respects fair and the Contract will be entered into by Contractor without collusion or fraud; and (iii) no official, officer or employee of the Owner has any direct or indirect financial interest in Contractor's bid proposal or in Contractor.
- Contractor knows and understands the Equal Employment Opportunity Clause administrated by the Illinois Department of Human Rights, which is incorporated herein by this reference, and agrees to comply with the provisions thereof. Contractor further certifies that Contractor is an "equal opportunity employer" as defined by Section 2000 (e) of Chapter 21, Title 42 of the United States Code Annotated and Executive Orders #11246 and #11375 as amended, which are incorporated herein by this reference.
- Neither Contractor nor any substantially owned affiliated company is participating or shall participate in an international boycott in violation of the provisions of the U.S. Export Administration Act of 1979 or the regulations of the U.S. Department of Commerce promulgated under that Act.

Init.

- Contractor is not barred from contracting with the Owner because of any delinquency in the payment of any tax administrated by the Illinois Department of Revenue, unless it is being contested. Contractor further certifies that it understands that making a false statement regarding delinquency in taxes is a Class A. misdemeanor and, in addition, voids the Contract and allows the Owner, a municipal entity, to recover in a civil action all amounts paid to the Contractor.
- If Contractor has 25 or more employees at the time of letting of the Contract, Contractor knows, understands and acknowledges its obligations under the Illinois Drug Free Workplace Act (30 ILCS 580/1 et seq.) and certifies that it will provide a drug-free workplace by taking the actions required under. and otherwise implementing on a continuing basis, Section 3 of the Drug Free Workplace Act. Contractor further certifies that it has not been debarred and is not ineligible for award of this Contract as the result of a violation of the Illinois Drug Free Workplace Act.
- The Contractor shall cause all Subcontractors to comply with the requirements and provisions of the Illinois Substance Abuse Prevention on Public Works Projects Act (820 ILCS 265/1 et seq.) (the "Act") by:
  - Prohibiting the use, possession, distribution or delivery of any drug or alcohol (as defined under the Act) or allowing any employee to be under the influence of any said drug or alcohol while performing the Work;
  - Filing a written substance abuse prevention program with the Owner for the prevention of substance abuse among its employees prior to the commencement of the Work. Said program shall be available to the general public and, at a minimum, contain the following:
    - A minimum requirement of a 9 panel urine drug test plus a test for alcohol. Testing an employee's blood may only be used for post-accident testing, however, blood testing is not mandatory for the employer where a urine test is sufficient;
    - A prohibition against the actions for the use, possession, distribution or delivery of any drug or alcohol (as defined under the Act) or any employee under the influence of any said drug or alcohol while performing the Work;
    - A requirement that employees performing the Work submit to pre-hire, random, reasonable suspicion, and post-accident drug and alcohol testing. Testing of an employee before commencement of the Work is not required if the employee participated in a random testing program during the ninety (90) days preceding the date on which the employee commenced work hereunder; and
    - A procedure for notifying an employee that he or she may not perform any of the Work if he or she: 1) uses, possess, delivers or is under the influence of a drug or alcohol as prohibited under the Act; 2) tests positive for the presence of a drug as outlined in the Act; or 3) refuses to submit to drug or alcohol testing as required under the Contractor's substance abuse program until the employee tests negative for the presence of drugs or alcohol as outlined in the Act or has been approved to commence or return to work in accordance with the Contractor's substance abuse program.
  - Immediately removing and/or prohibiting access to the Work site of any employee who: 1) uses, possess, delivers or is under the influence of a drug or alcohol as prohibited under the Act; 2) tests positive for the presence of a drug as outlined in the Act; or 3) refuses to submit to drug or alcohol testing as required under the Contractor's substance abuse program. Said employee shall be prohibited from the Work site until he or she tests negative for the presence of drugs or alcohol as outlined in the Act or has been approved to commence or return to work in accordance with the Contractor's substance abuse program; and
    - d. Complying with all other requirements of the Act.
- The Contractor hereby acknowledges that pursuant to 5 ILCS 140/7(2) any record of the Contractor that relates directly to a governmental function being performed by the Contractor pursuant to

1

this Agreement is considered a public record of the Owner for purposes of the Freedom of Information Act (FOIA), and upon request of the Owner's FOIA Officer, Contractor shall within two (2) business days turn over to the FOIA Officer any record requested that is in possession of the Contractor.

§ 13.8.2 Contractor and any subcontractor shall keep and maintain accurate books of record and account, in accordance with sound accounting principles, of all expenditures made and all costs, liabilities and obligations incurred under this Contract, and all papers, files, accounts, reports, cost proposals with backup data and all other material relating to work under this Contract and shall make all such materials available at the office of the Owner at any reasonable time during the term of this contract and for the length of time established by law or five (5) years. whichever is longer from the date of final payment to Contractor or termination of this Contract for audit, inspection and copying upon Owner's request.

§ 13.8.3 The Contractor shall be required to remain for the entirety of the Contract in compliance with the foregoing legal requirements. A violation is grounds for the immediate termination of the Contractor for cause. However, any forbearance in delay by the Owner in terminating Contractor or canceling the Contract shall not constitute a waiver of any right the Owner may have, including without limitation termination of Contractor, cancellation of the Contract and recovery of damages.

# §13.8.4 Record Keeping.

§13.8.4.1 Contractor and any subcontractor shall keep and maintain accurate books of record and account, in accordance with sound accounting principles, of all expenditures made and all costs, liabilities and obligations incurred under this Contract, and all papers, files, accounts, reports, cost proposals with backup data and all other material relating to work under this Contract and shall make all such materials available at the office of the Park District Executive Director at any reasonable time during the term of this contract and for the length of time established by law or five (5) years, whichever is longer from the date of final payment to Contractor or termination of this Contract for audit, inspection and copying upon Owner's request.

# ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT

# § 14.1 TERMINATION BY THE CONTRACTOR

- § 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 90 consecutive days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons:
  - Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped;
  - An act of government, such as a declaration of national emergency that requires all Work to be stopped;or
  - Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents.
  - The Owner has failed to furnish to the Contractor promptly, upon the Contractor's request, reasonable evidence as required by Section 2.2.1.
- § 14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, repeated suspensions, delays or interruptions of the entire Work by the Owner as described in Section 14.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.
- § 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner, terminate the Contract and recover from the Owner payment for Work executed,
- § 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract

with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner terminate the Contract and recover from the Owner as provided in Section 14.1.3.

# § 14.2 TERMINATION BY THE OWNER FOR CAUSE § 14.2.1

(Paragraphs deleted)

If the Contractor shall institute proceedings or consent to proceedings requesting relief or arrangement under the Federal Bankruptcy Act or any applicable Federal or State Law, or if a petition under any federal or state insolvency law is filed against the Contractor and such petition is not dismissed within sixty (60) days from the date of said filing, or if the Contractor admits in writing his inability to pay his debts generally as they become due, or if he makes a general assignment for the benefit of his creditors, or if a receiver, liquidator, trustee, or assignee is appointed on account of his bankruptcy or insolvency; or if a receiver of all or any substantial portion of the Contractor's properties is appointed; or if the Contractor abandons the Work; or if he fails, except in cases for which extension of time is provided, to prosecute promptly and diligently the Work or to supply enough properly skilled workmen or proper materials for the Work; or if the Contractor submits an application for payment, sworn statement, waiver of lien, affidavit or document of any nature whatsoever which is intentionally falsified; or if the Contractor fails to make prompt payment to Subcontractors for materials or labor (Paragraphs deleted)

or otherwise breaches obligations under any subcontract with a Subcontractor; or if a mechanic's or materialman's lien or a notice of lien is filed against any part of the Project and is not promptly bonded or insured over by the Contractor in a manner reasonably satisfactory to the Owner; or if the Contractor disregards any laws, statutes, ordinances, rules, regulations or orders of any governmental body or public or quasi-public authority having jurisdiction of the Work or the Project premises; or if the Contractor otherwise violates any material provision of the Contract Documents, then, without prejudice to any right or remedy available Owner may, after giving the Contractor seven (7) days' written notice, terminate the employment of the Contractor, and take possession of the Project and of all materials, equipment, tools, construction equipment and machinery thereon owned by the Contractor and accept assignment of Subcontracts and may complete the Work by whatever reasonable method the Owner may deem expedient. If requested by the Owner, the Contractor shall remove any part or all of his equipment, machinery and supplies from the Project within seven (7) days from the date of such request, and in the event of the Contractor's failure to so, the Owner shall have the right to remove or store, or remove and store, such equipment, machinery and supplies at the Contractor's expense. In case of such termination, the Contractor shall not be entitled to receive any further payment for Work performed by the Contractor through the date of termination until final completion of the Work

# § 14.2.2 When

(Paragraphs deleted)

the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.

§ 14.2.3 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. This obligation for payment shall survive termination of the Contract.

(Paragraph deleted)

expires on 08/24/2017, and is not for resale.

§14.2.4 The Owner's right to terminate the Contract pursuant to Section 14.2 shall be in addition to and not in limitation of its right to stop the Work without terminating the Contract pursuant to Section 2.3.

## § 14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

§ 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.

User Notes:

Init.

- § 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 14.3.1. Adjustment of the Contract Sum shall not include profit. No adjustment shall be made to the extent
  - .1 that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
  - .2 that an equitable adjustment is made or denied under another provision of the Contract.

#### § 14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

- § 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.
- § 14.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall
  - .1 cease operations as directed by the Owner in the notice;
  - .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and
  - .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.
- § 14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed.

#### ARTICLE 15 CLAIMS AND DISPUTES

§ 15.1 CLAIMS

§ 15.1.1 DEFINITION

A "Claim" is a written demand or assertion by the Contractor seeking adjustment to interpretation of Contract terms, payment of money, extension of time or other relief with respect to the terms arising out of the Contract.

#### § 15.1.2 NOTICE OF CLAIMS

Claims by the Contractor must be initiated by written notice to the Owner with a copy sent to the Architect. Claims by Contractor must be initiated within 15 days after occurrence of the event giving rise to such Claim or within 15 days after the Contractor first recognizes the condition giving rise to the Claim, whichever is later.

#### § 15.1.3 CONTINUING CONTRACT PERFORMANCE

Pending final resolution of a Contractor's Claim, , the Contractor shall proceed diligently with the performance required of him under the Contract.

#### § 15.1.4 CLAIMS FOR ADDITIONAL COST

If the Contractor wishes to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

#### § 15.1.5 CLAIMS FOR ADDITIONAL TIME

- § 15.1.5.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.
- § 15.1.5.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated and had an adverse effect on the scheduled construction.
- § 15.1.5.3 Unit Prices. The Contractor shall be responsible for notifying the Owner of any discrepancies or additions to work items completed on a unit price basis. This notification must take place prior to the execution of the Work. The purpose of this requirement is to make sure the Owner is aware of the extra items affecting the cost of the original contract amount. Discrepancies in the multiplication of units of work and unit prices will be resolved in favor of the unit prices. Discrepancies between the indicated sum of any column of figures and the corrected sum

init.

1

thereof will be resolved in favor of the corrected sum.

#### § 15.1.6 CLAIMS

The Contractor waives Claims against the Owner for consequential damages arising out of or relating to this Contract. This

(Paragraphs deleted)

waiver includes, but is not limited to damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit arising from the Work.

This waiver is applicable, without limitation, to all consequential damages due to Owner's termination in accordance with Article 14.

In any suit or action arising under this Contract the Owner shall be entitled to an award of reasonable attorney's fees and costs of litigation.

(Paragraphs deleted)

#### §15.1.7 RESOLUTION OF DISPUTES

§15.1.7.1 Venue. Any suit or action arising under this Contract shall be commenced in DuPage County, Illinois, but only after exhausting all possible administrative remedies.

§15.1.7.2 Limitations On Contractor's Claims. No suit or action shall be maintained by Contractor, its successors or assigns, against Owner on any claim based upon or arising out of this Contract or out of anything done in connection with this Contract unless such action shall be commenced within one year of the termination of this Contract.

§15.1.7.3 Waiver of Punitive Damages. The Contractor and Owner waive all claims against each other for all punitive damages arising out of or relating to this Contract, but nothing in this paragraph shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

(Paragraph deleted)

# Additions and Deletions Report for

AIA® Document A201™ - 2007

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 11:11:21 on 10/04/2016.

#### PAGE 1

Tennis Center Reflective Ceiling and Insulation Project
Oak Brook Park District Tennis Center
1300 Forest Gate Road
Oak Brook, IL 60523

(Name, legal status (Name and address)
Oak Brook Park District
1450 Forest Gate Road
Oak Brook, IL 60523

(Name, legal status and address) (Name and address)

NA. No Architect is being used for this Project. For purposes of this Project, "Architect" shall mean "Owner."

#### PAGE 3

(Topics and numbers in bold are section headings.)(Numbers and Topics in Bold are Section Headings)

3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 10.2.8, 13.4.2, <del>13.7, 13.7.1,</del> 14.1, 15.2 ...

1.1.1, 3.113.11.1

---

4.2.5, 7.3.9, 9.2, **9.3**, 9.4, 9.5.1, 9.6.3, <del>9.7, 9.7.1, 9.10</del>, 11.1.3

8.3.1, 11.3.10, <del>13.1, 13.1.1,</del> 15.3.2, **15.4** 

<del>2.4, 2.4.1,</del> 3.12.7, 4.1, 4.2, 5.2, <del>6.3, 6.3.1,</del> 7.1.2, 7.3.7, 7.4, <del>9.2, 9.2.1,</del> 9.3.1, 9.4, 9.5, 9.6.3, 9.8, 9.10.1, 9.10.3, 12.1, 12.2.1, 13.5.1, 13.5.2, 14.2.2, 14.2.4, 15.1.3, 15.2.1

Additions and Deletions Report for AIA Document A201<sup>TM</sup> = 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Toder No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

```
2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2, 4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, <del>7.4, 7.4.1,</del> 9.4.2,
9.5.3, 9.6.4, 15.1.3, 15.2
<del>2.4, 2.4.1,</del> 11.3.1.1, 12.2.1, 13.5.2, 13.5.3, 14.2.4
<del>2.4, 3.1.3, 3.5, <u>2.4.1, 3.1.3, 3.5.1, 3.10.2, 4.2.7</u></del>
<del>3.5, 3.5.1, 4.2.6, 12.1.2, 12.2.1</del>
3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 4.2.14, <del>6.3, 6.3.1,</del> 7.3.7, 7.3.9, 8.1.3, 8.3.1, <del>9.2, 9.2.1,</del> 9.4.1, 9.5, 9.8.4,
9.9.1, 13.5.2, 15.2, 15.3
1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, <del>3.5, 3.5.1,</del> 3.7.4, 3.7.5, 3.9.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, 3.18,
4.1.2, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13.4.2, 13.5, 15.2
<del>9.7, 9.7.1, 11.3.9, 11.3.10, 13.1, 13.1.1, 15.2.5, 15.2.6.1, 15.3.1, 15.3.2, 15.4.1</del>
PAGE 4
4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, <del>9.7, 9.7.1</del>, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.3
1.1.1, <del>2.4, 2.4.1,</del> 3.4.2, 3.7.4, 3.8.2.3, <del>3.11, 3.11.1,</del> 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, 7.2, 7.3.2, 7.3.6, 7.3.9, 7.3.10,
8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3
2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9
3.2.4, 6.1.1, <del>6.3, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4</del>
3.2.4, <del>3.7.4, 6.1.1,</del> 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5
<del>1.6, 1.6.1, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 10.2.2, 11.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14.1.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 13.6, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.1.1, 14.</del>
```

Additions and Deletions Report for AIA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

14.2.1.3, 15.2.8, 15.4.2, 15.4.3

```
3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1
 Contract Documents, The
 1.1.1
 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.9, 9.7.1, 10.3.2, 12.1.1, 14.3.2,
 15.1.5.1, 15.2.5
 PAGE 5
 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3, 11.1.1, 11.3.7, 14.1, <del>14.2.1.1</del>14.2.1.1,
 1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, <del>3.5, 3.5.1, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.3, 4.2, 5.2, 6.2.2, 7, 3.2.2, 3.2.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.2.4, 3.3.1, 3.4.2, 3.5.3, 3.5.1, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.3, 4.2, 5.2, 6.2.2, 7, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.3.1, 3.4.2, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2.4, 3.2</del>
 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13.5, 15.1.2, 15.2.1
 ...
 3.2.1, 3.2.2, <del>3.5, 3.5.1,</del> 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2
 3.3.2, 3.18, <del>5.3, 5.3.1,</del> 6.1.3, 6.2, 9.5.1, 10.2.8
<del>2.4, 2.4.1,</del> 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3, 7.3.3, 7.3.7, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6, 11.3,
 12.1.2, 12.2.1, 12.2.4, 13.5, 14
 ...
3.14.2, 9.9.1, 10.2.1.2, 10.2.5, <del>10.4, 10.4.1, 11.3.1, 12.2.4</del>
3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 15.2, 6.3, 7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2, 9.2.1, 9.4, 9.5.1, 9.8.4, 9.9.1, 13.5.2,
 14.2.2, 14.2.4, 15.1, 15.2
<del>2.3, 2.4, 3.5, 2.3.1, 2.4.1, 3.5.1, 4.2.6, 6.2.5, 9.5.1, 9.5.2, 9.6.6, 9.8.2, 9.9.3, 9.10.4, 12.2.1</del>
Defective Work, Definition of
3.5.1
1.1, 2.1.1, 3.1.1, <del>3.5, 3.5.1,</del> 3.12.1, 3.12.2, 3.12.3, 4.1.1, 15.1.1, 5.1, 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1
<del>3.2, 3.2.,</del> 3.7.4, 5.2.3, 7.2.1, 7.3.1, <del>7.4, 7.4.1</del>, 8.3, 9.5.1, <del>9.7, 10.3.2, 10.4,</del> 9.7.1, 10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5
```

Additions and Deletions Report for AIA Document A201<sup>TM</sup> = 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

<del>6.3, 6.3.1, 7.3.9, 15.1, 15.2</del>

#### PAGE 6

```
1.1.3, 1.1.6, 3.4, <del>3.5, 3.5.1, 3.8.2, 3.8.3, 3.12, <u>3.13, 3.13.1, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3,</u></del>
 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2
 •••
 1.1.3, 1.2.1, 1.2.2, 2.2.3, 2.2.5, 3.1, 3.3.1, 3.4.1, <del>3.5,</del> <u>3.5.1,</u> 3.7.1, 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.5, 8.2, 9.5.1,
 9.9.1, 10.2, 10.3, 12.2, 14.2, 14.3.1, 15.1.3
 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, <del>7.4</del>, 9.5.1, 9.<del>7, 10.3.2, 10.4, 7.4.1, 9.5.1, 9.7.1, 10.3.2, 10.4.1, 14.3, 15.1.5, 15.2.5</del>
 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.3.1, 11.3.5, <del>12.3,</del> <u>12.3.1</u>, 14.2.4, 14.4.3
 ...
 <del>3.17, 3.17.1, 3.18,</del> 9.10.2, 10.3.3, 10.3.5, 10.3.6, 11.3.1.2, 11.3.7
 2.1.2, 2.2, 3.2.2, 3.12.4, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.2.1, 11.4, 13.5.1,
 13.5.2, 14.1.1.4, 14.1.4, 15.1.3
 10.2.8, 10.410.4.1
9.3.29.3.2, 11.4.1.4
Insurance Companies, Consent to Partial Occupancy
9.9.1, 11.4.1.5
Insurance Companies, Settlement with
9.9.111.4.10
1.1.3, 1.1.6, 3.4, 3.5, 3.5.1, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3,
9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2
1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 3.13.1, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1.1, 11.3, 13.1, 13.1.1, 13.4, 13.5.1, 13.5.2,
<del>13.6, 13.6.1, 14, 15.2.8, 15.4</del>
227
<del>2.3, 3.2.2, 3.5, 3.12.10, 3.17, 2.3.1, 3.2.2, 3.5.1, 3.12.10, 3.17.1, 3.18.1, 4.2.6, 4.2.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.6.4, 9.6.7, 4.2.12, 6.2.2, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4, 9.6.4,</del>
10.2.5, 10.3.3, 11.1.2, <del>11.2, <u>11.2.1,</u></del> 11.3.7, 12.2.5, 13.4.2
```

Additions and Deletions Report for AIA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

...

2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7, 5.2, <del>5.3, 5.3.1, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, <u>9.2, 9.2.1, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, <del>9.7, <u>9.7.1, 9.8, 9.9, 9.10, 11.1.3, 11.3.1.5, 11.3.6, 11.3.10, 12.2, 13.5, 13.7, 14, 15</del></del></del></u></del></u>

#### PAGE 7

1.1.3, 1.1.6, 1.5.1, 3.4.1, <del>3.5, 3.5.1,</del> 3.8.2, 3.8.3, 3.12, <del>3.13, 3.13.1,</del> 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1.2, 10.2.4, 14.2.1.1, 14.2.1.2

...

1.1.1, 1.1.2, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.<del>7,</del> 9.7.1, 10.3.2, 11.3.1

...

<del>2.3, 2.4, 3.5, 2.3.1, 2.4.1, 3.5.1, 4.2.6, 6.2.4, 9.5.1, 9.8.2, 9.9.3, 9.10.4, 12.2.1</del>

...

2.2.1, <del>2.3, 2.4, 2.3.1, 2.4.1, 3.2.4, 3.3.1, 3.7.2, 3.12.9, 5.2.1, <u>9.7, 9.7.1, 9.10, 10.2.2, 11.1.3, <u>11.4.6, 12.2.2.1, 13.3, 13.5.1, 13.5.2, 14.1, 14.2, 15.2.8, 15.4.1</u></del></u>

...

<del>2.3, 2.4, 2.3.1, 2.4.1,</del> 3.3.1, 3.9.2, 3.12.9, 3.12.10, 5.2.1, <del>9.7, <u>9.7.1,</u></del> 9.10, 10.2.2, 10.3, 11.1.3, 11.3.6, 12.2.2.1, **13.3**, 14, 15.2.8, 15.4.1

ere.

3.7.4, <u>4.5.</u> 10.2.8, **15.1.2**, 15.4

•••

2.1.2, **2.2**, 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, <del>11.2, 11.2.1, 11.3, 13.5.1, 13.5.2, 14.1.1.4, 14.1.4, 15.1.3</del>

...

1.5, 2.1.1, <del>2.3, 2.4, 2.3.1, 2.4.1, </del>3.4.2, 3.8.1, 3.12.10, 3.14.2, 4.1.2, 4.1.3, 4.2.4, 4.2.9, 5.2.1, 5.2.4, 5.4.1, 6.1, <del>6.3, 6.3.1, 7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.1, 9.3.2, 9.5.1, 9.6.4, 9.9.1, 9.10.2, 10.3.2, 11.1.3, 11.3.3, 11.3.10, 12.2.2, <del>12.3, 12.3.1, 13.2.2, 14.3, 14.4, 15.2.7</del></del>

...

# Owner's Loss of Use Insurance

#### 11.3.3

Owner's Relationship with Subcontractors

\*\*\*

1.1.1, 1.1.6, 1.1.7, **1.5**, 2.2.5, 3.2.2, <del>3.11, 3.17, 4.2.12, 5.3</del><u>3.11.1, 3.17.1, 4.2.12, 5.3.1</u>

. . .

4.2.5, 7.3.9, <del>9.2, <u>9.2.1,</u> **9.3**, 9.4, 9.5, 9.6.3, <del>9.7, <u>9.7.1,</u></del> 9.8.5, 9.10.1, 14.2.3, 14.2.4, 14.4.3</del>

```
4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, <del>9.7, 9.7.1, 9.10.1, 9.10.3, 13.7, 14.1.1.3, 14.2.4</del>
4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.4.1, <del>12.3, 11.4.5, 12.3.1, 13.7, 14.2.4, 14.4.3</del>
7.3.7.4, 9.6.7, 9.10.3, <u>11.4.9</u>, 11.4
PAGE 8
5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, <u>11.4.8</u>, 14.2.1.2
...
7.3.7.4, 9.6.7, 9.10.3, 11.4.9, 11.4
Project, Definition of the
3.5, 3.5.1, 4.2.6, 12.2.1
3.2.1, <del>3.5, 3.5.1,</del> 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.4.2, 9.5.1, 9.8.2, 9.10.1
3.3.2, 3.18, 4.2.3, <del>5.3,</del> <u>5.3.1,</u> 6.1.3, 6.2, 6.3, 9.5.1, 10
1.1.2, 2.3, 2.4, <del>3.5, 3.5.1,</del> 3.7.4, 3.15.2, 4.2.6, <u>4.5,</u> 5.3, 5.4, 6.1, 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.2, 12.2.4,
13.4, 14, 15.4
•••
3.3.1, 4.2.2, 4.2.7, <del>5.3, <u>5.3.1,</u> 10.1, 10.2, 10.4</del>
<u>1.4.1.2</u>, 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2
1.1.4, 3.12.5, 3.14.2, 4.2.4, 4.2.7, 6, 8.3.1, <u>11.4.7</u>, 12.1.2
```

Specifications, Definition of the

```
Specifications Specifications, The
6.2.1, 9.3.2, 10.2.1.2, <del>10.2.4</del> <u>10.2.4</u>, 11.4.1.4
5.3, 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, <u>11.4.7, 11.4.8,</u> 14.1, 14.2.1
PAGE 9
6.1.1, 11.4.5, 11.3.7
3.4.2, <del>3.5, 3.5.1, 7.3.8</del>
...
5.4.1.1, <u>11.4.9,</u> 14
3.1.3, 3.3.3, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 10.3.2, <del>11.4.1,</del> <u>11.4.1.1,</u> 12.2.1, 13.5
3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3.1, <del>7.4, 7.4.1, 8.3</del>, 9.5.1, <del>9.7, 10.3.2, 10.4, 9.7.1, 10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5</del>
2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2, <u>4.4, 4.5,</u> 5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1,
9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3, <u>11.4.1.5</u>, <u>11.4.6</u>, <u>11.4.10</u>, <u>12.2</u>, 13.5, 13.7, 14, 15.1.2, 15.4
...
9.10.5, <u>11.4.7</u>, 13.4.2, 15.1.6
9.9.3, 9.10.3, 9.10.4, 11.4.3, 11.4.5, 11.4.7, 12.2.2.1, 13.4.2, 14.2.4, 15.1.6
6.1.1, 11.4.5, 11.3.7
3.5, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.4, 12.2.2, 13.713.7.1
PAGE 10
```

Additions and Deletions Report for AIA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

2.3, 2.4, 3.3.1, 3.9, 3.12.9, 3.12.10, 5.2.1, 8.2.2, 9.7, 9.10, 10.2.2, 10.3, 11.1.3, 11.4.6, 12.2.2, 12.2.4, 13.3, 14, 15.4.1

•••

1.1.1, 2.3, 3.9, 7, 8.2.2, 11.4.9, 12.1, 12.2, 13.5.2, 14.3.1, 15.1.2

#### PAGE 11

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement Agreement, Advertisement for Bids, Invitation & Instructions to Bidders, Conditions (General, Supplementary and Special Supplementary), Proposal, Surety Bond, Performance Bond, Labor and Material Payment Bond, Plans, Drawings, Specifications, Addenda, Bulletins, Supplemental Plans, Supplemental Specifications, and Standards & Specifications pursuant to Paragraph 1.5.4 below, and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract approved by the Owner and signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding requirements.

٠.,

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, Sub-subcontractor provided, however, Owner shall be a third party beneficiary of any Subcontract agreement under the circumstances set forth in Article 5 herein, (3) between the Owner and the Architect or the Architect's consultants or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

...

## § 1.1.7 INSTRUMENTS OF SERVICE

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

#### § 1.1.8 INITIAL DECISION MAKER

The Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2 and certify termination of the Agreement under Section 14.2.2.

- §1.1.7 "As selected", or "as approved", or "satisfactory to", or "as required", or "as directed" mean as selected, approved, required or directed by the Owner or by the Architect acting for the Owner, unless otherwise specified in the specific provision in which the term is used.
- §1.1.8 "Final Completion" means the date the Contract has been fully performed, all the Work has been completed and a final Certificate for Payment approved by the Owner has been issued by the Architect.
- §1.1.9 "Indicated" and "shown" mean as described, detailed, discussed, scheduled, referenced, or called for in, or reasonably inferable from, the Contract Documents, in order to produce a first class Work product.
- §1.1.10 "Or equal", "approved equal", or "equal to" mean that the determination whether an alternative product or system is equal to that indicated or specified shall be made by the Owner.

- §1.1.11 "Provide" or derivatives thereof means the Contractor shall properly fabricate, supply, furnish or procure all labor, materials, equipment, apparatus, and accessory appurtenances necessary to transport, deliver, install, erect and construct the specified item, complete, in place, fully tested and ready for operation and use, including any final connections in strict accordance with the Drawings, Specifications and other Contract Documents. The words "Contractor shall" are implied and shall be so understood wherever the direction "provide" is used.
- §1.1.12 "Punch List Items" shall mean and shall be limited to uncompleted items of the Work (a) that do not interfere with the use and occupancy of any area of the Site for its intended purpose and (b) that, as a group, are capable of being completed by the Contractor within thirty (30) days of issuance of any Punch List. The "Punch List" is the list containing the Punch list items.
- §1.1.13 "Substantial Completion" means the date that all of the Work has been completed to the point where it can be occupied and used for all purposes intended by Owner and bas been accepted by Owner and Architect as such, subject only to minor Punch List Items which do not affect full use and enjoyment," and the Contractor has completed all Work necessary in order for the Owner to receive all required occupancy permits.

#### § 1.1.14 THE PROJECT MANUAL

The Project Manual is a volume assembled for the Work which may include the bidding requirements, sample forms, Conditions of the Contract and Specifications.

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Work shall consist of all items specifically included in the Contract Documents as well as all additional items of work "indicated" in or from the Contract Documents. The Contract Documents are complementary, and what is required by any one Contract Document shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them by all. By its execution of the Contract, the Contractor shall represents, acknowledges and agrees that any differences between the requirements of the Drawings and the Specifications or any differences noted within the Drawings themselves or within the Specifications themselves or any conflicts between the Contract Documents and applicable standards, codes and ordinances, have been referred to the Owner and Architect by Contractor prior to submission of bids and have been clarified by an Addendum issued to all bidders.

If any such differences or conflicts were not called to the Owner's and Architect's attention prior to submission of bids, the Architect shall decide which of the conflicting requirements will govern based upon the most stringent or highest quality of the requirements, and, subject to the approval of the Owner, the Contractor shall perform the Work at no additional cost and/or time to the Owner in accordance with the Architect's decision. Work not indicated in the Contract Documents will not be required unless it is consistent therewith and is reasonably inferable therefrom as being necessary to produce the indicated results.

Large scale drawings shall take precedence over small scale drawings; figured dimensions on drawings over scaled dimensions and noted material over graphic representations.

# **PAGE 12**

- § 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings. <u>Unless specifically stated as an obligation of the Owner, it is understood to be an obligation of the Contractor.</u>
- §1.2.4 All Work shall conform to the Contract Documents. No change therefrom shall be made without review and written acceptance by Architect and Owner.

...

# § 1.4 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

Additions and Deletions Report for AIA Document A201<sup>TM</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

(1950958438)

#### § 1.4.1 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

#### § 1.4.2 HEADINGS.

The headings for each paragraph of the Contract Documents are for convenience and reference purposes only and in no way define, limit or describe the scope or intent of said paragraphs or of the Contract Documents nor in any way affect the Contract Documents.

# § 1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENTS OF SERVICEOWNERSHIP AND USE OF DRAWINGS. SPECIFICATIONS AND OTHERPROJECT DOCUMENTS

§ 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, Contract Documents are owned and copyrighted by the Owner. No Contractor, Subcontractor or Materialman shall retain or claim ownership or copyright of Contract Documents. No Contractor, Subcontractor or Materialman shall use any Contract Document for another project. All Contractors and Subcontractors have a limited license to reproduce portions of Contract Documents for this project, which cannot be obtained from Architect or Owner.

The Architect and Owner shall be deemed the owners of the Project Documents, including the Drawings and Specifications, and will retain all common law, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Project Documents. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' Architect or Owner's reserved rights.

§ 1.5.2 The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Architect and the Architect's consultants. Contractor hereby specifically acknowledges and declares, and the execution of the Contract by the Contractor is a representation of the Contractor that the Contract Documents are full and complete, are sufficient to have enabled the Contractor to determine the cost of the Work and that the Contract Documents are sufficient to enable it to construct the Work outlined therein, in accordance with applicable laws and regulations, and otherwise to fulfill all its obligations hereunder, including, but not limited to, Contractor's obligation to construct the Work for an amount not in excess of the Contract Sum on or before the date(s) of Substantial Completion established in the Agreement. The Contractor further acknowledges and declares that it has visited and examined the site, examined all physical and other conditions affecting the Work and is fully familiar with all of the conditions thereon and thereunder affecting the same. In connection therewith, Contractor specifically represents and warrants to Owner that it has, by careful examination, satisfied itself as to: (1) the nature, location, and character of the Project and the site, including, without limitation, surface conditions of the site and subsurface conditions readily observable or ascertainable upon the exercise reasonable diligence and all structures and obstructions thereon and thereunder, both natural and manmade, and all surface and subsurface water conditions of the site and the surrounding area; (2) the nature, location, and character of the general area in which the Project is located, including without limitation, its climatic conditions, available labor supply and labor costs, and available equipment supply and equipment costs; and (3) the quality and quantity of all materials, supplies, tools, equipment, labor, and professional services necessary to complete the Work in the manner and within the cost and time frame required by the Contract Documents. In connection with the foregoing, and having carefully examined all Contract Documents, as aforesaid, and having visited the site, the Contractor acknowledges and declares that it has no knowledge of any discrepancies, omissions, ambiguities, or conflicts in said Contract Documents and that if it becomes aware of any such discrepancies, omissions, ambiguities, or conflicts, it has an obligation to and will promptly notify Owner and Architect of such fact, and will not proceed until it shall have received the written interpretation of Owner or Architect.

Claims for additional compensation or extensions of time because of the failure of the Contractor to familiarize himself with the conditions enumerated in the Contract Documents and other conditions of the site which might affect the Work, or because of the failure of the Contractor to obtain a needed interpretation before proceeding with the Work, will not be allowed.

#### **PAGE 14**

- § 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or euthorization. authorization, excluding matters requiring approval by the Board of Park Commissioners or the Executive Director. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.
- § 2.1.2 The Owner shall furnish to the Contractor within fifteen days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of or enforce mechanic's lion rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein. Intentionally omitted.

- § 2.2.1 Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Contractor may only request such evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.
- Decision of the Owner. All work done under this Contract shall be done to the satisfaction of the Owner who shall in all cases determine the amount of work done which is to be paid for under this Contract. The Owner shall decide all questions that may arise as to the measurements of quantities and the fulfillment of this Contract on the part of the Contractor, and shall determine all questions concerning the true intent or meaning of the Plans and Specifications and his determination and decision shall be final and conclusive.

- § 2.2.3 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work-furnishing of surveys by the Owner is not a guarantee of the accuracy of the information contained therein. and shall not relieve the Contractor from its duties under the Contract Documents in general. The submission of a bid for the work implies that the Contractor had examined the site, taking into consideration all such conditions that may affect the work, regardless of the information contained in the surveys.
- § 2.2.4 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control required to be furnished by Owner and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or repeatedly fails to carry out 12.2, or fails or refuses to provide a sufficient amount of

properly supervised and coordinated labor, materials, or equipment so as to be able to complete the Work within the Contract Time or fails to remove and discharge (within ten days) any lien filed upon Owner's funds by anyone claiming by, through, or under Contractor, or disregards the instructions of Architect or Owner when based on the requirements of the Contract Documents, or fails to carry out any portion of the Work in accordance with the Contract Documents, or in the event an emergency arises that requires the Work to be stopped, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; eliminated or the emergency no longer exists; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3. The Owner and Architect shall at all times have access to the Project Site for purposes of inspection whenever the Work is in preparation and progress.

#### **PAGE 15**

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten-day seven-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, or fails within such seven-day period to eliminate (or diligently commence to eliminate) the cause of any stop work order issued under Section 2.3 thereof, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable actual cost of correcting such deficiencies, including Owner's expenses and compensation for the Architect's additional services made necessary by such default, neglect or failure. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

## §2.5 OWNER'S REMEDIES NOT EXCLUSIVE

The rights and remedies of Owner stated in this Article 2 shall be in addition to and not in limitation of, any other rights of the Owner granted in the Contract Documents or at law or in equity

It is understood and agreed that the relationship of Contractor to Owner shall be that of an independent contractor. Nothing contained herein or inferable herefrom shall be deemed or construed to (1) make Contractor the agent, servant, or employee of the Owner, or (2) create any partnership, joint venture, or other association between Owner and Contractor. Any direction or instruction by Owner in respect of the Work shall relate to the results the Owner desires to obtain from the Work, and shall in no way affect Contractor's independent contractor status as described herein.

§ 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents. High quality craftsmanship will be expected in all phases of work. Any elements found unacceptable and not in compliance with the Contract Documents will be removed and replaced by the Contractor until satisfactory results are obtained.

...

§ 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with evaluated the local conditions under which the Work is to be performed and correlated personal observations and, except for concealed conditions or those unknown to Contractor, found the site suitable for completing the Work in compliance with requirements of the Contract Documents. Furthermore, execution of the Contract by the Contractor is a representation that the Contract Documents include the construction details, means, methods, procedures and techniques necessary to perform the Work.

Before submitting its bid proposal the Contractor shall carefully study and compare the Contract Documents with each other and with information furnished by Owner pursuant to Section 2.2.2 and shall at once report to the Architect and Owner errors, inconsistencies or omissions discovered. The Contractor shall not be liable to the Owner or Architect for damage resulting from errors, inconsistencies or omissions in the Contract Documents that could not have been discovered by a reasonably diligent contractor experienced in performance of the type of work of the Work of this Project in advance of performance and that are not of the nature of items described in and intended to be covered in Sections 1.2.1 and 1.5.2 hereof, unless the Contractor recognized or reasonably should

have recognized such error, inconsistency or omission and failed to report it to the Owner and Architect. If the Contractor performs any construction activity involving an error, inconsistency or omission in the Contract Documents that the Contractor recognized or reasonably should have recognized and of which Contractor failed to notify the Owner and Architect, the Contractor shall assume complete responsibility for such performance and shall bear the full amount of the attributable costs for correction.

- §3.2.1.1. If any errors, inconsistencies, or omissions in Contract Documents are recognized or reasonably should have been recognized by the Contractor, any employee or agent of the Contractor, or any of its Subcontractors, the Contractor shall be responsible for notifying the Owner and Architect in writing of such error, inconsistency, or omission before proceeding with the Work. The Architect will take such notice under advisement and within a reasonable time commensurate with job progress render an interpretation. The Architect's interpretation shall be subject to Owner's approval. If the Contractor fails to give such notice and proceeds with such Work, it shall correct any errors, inconsistencies, or omissions at no additional cost to the Owner.
- § 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe evaluate any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; Documents, except to the extent that such errors, omissions or inconsistencies affect Contractor's services; however, the Contractor shall promptly report to the Architect and the Owner any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.
- § 3.2.3 The Contractor is not Both Architect and the Contractor are required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the authorities. The Contractor shall promptly report to the Architect and the Owner any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require. Contractor.
- § 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, the Contractor shall make Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner as would have been avoided avoided, including any increases in construction costs, if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

## **PAGE 16**

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner and Architect and shall not proceed with that portion of the Work without further written instructions from the Owner and Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures.

#### **PAGE 17**

- §3.3.4 The Contractor has the responsibility to ensure that all material suppliers and Subcontractors, their agents, and employees adhere to the Contract Documents, and that they order materials on time, taking into account the current market and delivery conditions and that they provide materials on time. The Contractor shall coordinate its Work with that of all others on the Project including deliveries, storage, installations, and construction utilities. The Contractor shall be responsible for the space requirements, locations, and routing of its equipment. In areas and locations where the proper and most effective space requirements, locations, and routing cannot be made as indicated, the Contractor shall meet with all others involved, before installation, to plan the most effective and efficient method of overall installation.
- §3.3.5 The Contractor shall establish and maintain bench marks and all other grades, lines and levels necessary for the Work, report errors or inconsistencies to the Owner and Architect before commencing Work, and review the placement of the building(s) and permanent facilities on the site with the Owner and Architect after all lines are staked out and before foundation Work is started. The Contractor shall provide access to the Work for the Owner, Architect, other persons designated by Owner, and governmental inspectors. Any encroachments made by the Contractor or its Subcontractors (of any tier) on adjacent properties due to construction as revealed by an improvement survey, except for encroachments arising from errors or omissions not reasonably discoverable by the Contractor in the Contract Documents shall be the sole responsibility of the Contractor, and the Contractor shall correct such encroachments within thirty (30) days of the improvement survey (or as soon thereafter as reasonably possible), at the Contractor's sole cost and expense, either by the removal of the encroachment (and subsequent reconstruction on the Project site) or agreement with the adjacent property owner(s) (in form and substance satisfactory to Owner in its sole discretion) allowing the encroachments to remain.

...

- § 3.4.2 Except in the case of minor changes in the Work authorized by the Architect in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions The Contractor may make substitutions equivalent to or superior to the specified materials only with the consent of the Owner, after evaluation by the Architect and approval by the Owner and in accordance with a Change Order or Construction Change Directive.
- § 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them. Except as provided in 3.4.2, the materials specified have been determined to have characteristics appropriate for the purposes of this Project. No work will be acceptable which utilizes an alternate not approved during the bidding process.
- § 3.4.4 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.
- §3.4.5 Before ordering any material or doing any Work, the Contractor shall verify all measurements at the Project Site and he shall be responsible for the correctness of same. No extra charge or compensation will be allowed to the Contractor on account of any difference between actual dimensions and the measurements shown by the Project Drawings.
- §3.4.6 The Contractor shall carefully inspect all materials delivered on and to the Project Site and reject defective materials without waiting for the Architect or other representative of Owner to observe the materials.
- §3.4.7 Contractor shall maintain harmonious labor relations on the job site. If a labor problem arises or any person employed by the Contractor on the Work shall appear to the Owner to be incompetent or conduct himself in a disorderly or improper manner, such person or persons shall be removed from the Work immediately on the request of the Owner. Said removal shall not create any additional cost to Owner and shall not extend the time for

#### completion of the Work.

#### **PAGE 18**

The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. §3.5.1 The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. The warranty will not be affected by the specification of any product or procedure, unless the Contractor objects

promptly to such product or procedures and advises the Architect and Owner of possible substitute products or procedures which will not affect the warranty. This warranty shall not be restricted by the limitations of any manufacturer's warranty. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal-wear and tear and normal usage. If required by the Architect, or improper operation. If required by the Architect or the Owner, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

All warranties shall include labor and materials and shall be signed by the manufacturer or Subcontractor as the case may be and countersigned by the Contractor. All warranties shall be addressed to the Owner and delivered to the Architect upon completion of the Work and before or with the submission of request for final payment. Except as otherwise provided in these Supplementary Conditions, elsewhere in the Contract Documents, or in any Certificate of Substantial Completion approved by Owner and Contractor and/or Subcontractor, as applicable, all warranties shall become effective on the date of Final Completion of the entire Work, and shall run for a twelve (12) month period, unless a longer period is provided for in the Contract Documents or by law. Where warranties overlap, the more stringent requirement shall govern.

- §3.5.2.1 If materials or equipment are replaced during the original warranty period, a new warranty period thereon shall then begin from the date that such corrective action is completed and approved.
- §3.5.2.2 For concrete work, warranty protection for a repaired item shall be for twenty-four months after final acceptance of concrete work or the length of the original warranty period, whichever is longer. This will cover structural failures, as well as surface erosion due to spalling caused by frost popping soft aggregates within the concrete and surface erosion due to faulty workmanship. All concrete work not meeting high industry standards will be removed and replaced at no charge to the Owner.
- §3.5.2.3 Defective materials, equipment or workmanship occurring within the Warranty period may be repaired where such produces results conforming to the Contract Documents relating to appearance, performance and reliability. Where the nature of the defective materials, equipment or workmanship is such that acceptable results cannot be obtained by repair, such defective items shall be removed and replaced with new materials, equipment or workmanship complying with the Contract Documents.
- §3.5.2.4 Correction of defective or non-conforming Work shall include, in addition to that described in Article 12, any damage to the Project or other property that may result from such defective or nonconforming Work or from such corrective action, including without limitation any damage to any contents, to the work of other contractors, or to adjacent property.
- §3.5.2.5 The Contractor shall furnish maintenance and twenty-four (24) hour callback service for the equipment provided by Contractor, Subcontractor or supplier for a period of at least six (6) months after Final Completion and acceptance of the Work, or for such longer period as shall otherwise be provided in any of the Contract Documents. This service shall include regular examinations of the installation by competent and trained employees of the Contractor, or manufacturer, and shall include all necessary adjustments, greasing, oiling, cleaning, supplies and parts to keep the equipment in proper operation except such parts made necessary by misuse, accidents or negligence not caused by the Contractor or any of its Subcontractors.

§3.5.2.6 The warranty provided in this Paragraph 3.5 shall be in addition to and not in limitation of, any other warranty or remedy required under the Contract Documents or under applicable law.

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect. The Illinois State sales tax is not applicable to the Project with respect to materials, equipment and supplies incorporated in the Work or totally consumed in the performance of the Work. The Contractor shall pay unemployment and Social Security taxes and other taxes imposed by local, city, state or federal government with respect to Contractor's own personnel and certify to Owner that this has been done before payment is made to the Contractor.

## § 3.7 PERMITS, FEES, NOTICES AND COMPLIANCE WITH LAWSPERMITS, FEES, NOTICES, AND COMPLIANCE WITH

- § 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations eeneluded concluded except as indicated in 2.2.2 herein.
- § 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work-laws, ordinances, codes, rules, regulations and lawful orders of public authorities having jurisdiction over all or any part of the Work or any insurance organizations (collectively, "Legal Requirements") relating to the Work of the performance thereof. If the Contractor fails to give any such notice or to permit any inspection, it shall be responsible for and shall indemnify and hold harmless, the Owner, the Architect and their respective consultants, employees, officers and agents against and from any resulting Work delays, fines, penalties, judgments or damages, including without limitation reasonable attorneys' fees and court costs imposed on, or suffered, sustained or incurred by, any of the parties indemnified hereunder. The Contractor shall also be liable to the Owner for any delay in the performance of the Work or increase in the cost of the Work resulting from the Contractor's failure to fully comply with the provisions of this Subparagraph 3.7.2. If the Contractor performs Work that it knows or reasonably should have known would be contrary to Legal Requirements, without such notice to the Architect and the Owner, the Contractor shall assume full responsibility for such Work and delays and shall bear the attributable costs.
- § 3.7.3 If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.
- § 3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Decuments or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor in writing, stating the reasons. If either party disputes the Architect's determination or recommendation, that party may proceed as provided in Article 15.
- § 3.7.2.1 If the Contractor observes that portions of the Contract Documents are at variance with Legal Requirements, the Contractor shall promptly notify the Architect and Owner in writing, and necessary changes shall be accomplished by appropriate Modification.

- <u>S3.7.2.2</u> Copies of any and all permits, licenses and certificates shall be delivered to the Architect and Owner as soon as they are obtained. The Contractor shall deliver the originals of such permits, licenses and certificates to the Architect together with the Contractor's application for final payment.
- §3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 7 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor in writing, stating the reasons and the Contract Sum or time for performance of the Work shall not be adjusted. If either party disputes this determination or recommendation, that party may proceed as provided in Article 15.
- § 3.7.6 Contractor shall comply with all public and private utility requirements relating to the Work or the performance thereof. If the Contractor performs Work contrary to applicable utility requirements, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.
- § 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

  direct.

## **PAGE 20**

- .1 Allowances allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
- .3 Whenever whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.
- § 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. The Contractor's competent superintendent shall have the knowledge and control of all work under this Contract and shall communicate directly to the Owner upon request.
- § 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the and Architect the name and qualifications of a proposed superintendent. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to the proposed superintendent or (2) that the Architect or Owner requires additional time to review. Failure of the Architect to reply within the 14 day period period, or any extension of said time period, shall constitute notice of no reasonable objection.

Additions and Deletions Report for AIA Document A201<sup>TM</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1968, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

§ 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall be which shall not unreasonably be withheld or delayed.

...

§ 3.10.2 The Contractor shall prepare a submittal schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit the schedule(s) for the Architect's and Owner's approval. The Architect's and Owner's approval shall not unreasonably be delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Owner and Architect reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.

#### **PAGE 21**

- § 3.10.4 Construction Engineerting Check. The Contractor shall notify the Owner three (3) business days in advance of all grading, drainage, and other major items of construction for field checking of construction engineering. All questions pertaining to the Plans, Specifications and details of the work shall be directed to the Owner and cleared prior to construction.
- § 3.10.5 Contractor's Construction Schedule. The Contractor shall provide regular monitoring and updating of the Progress Schedule with monthly Update Reports submitted contemporaneously with the monthly pay application, or more frequently as required by the conditions of the Work. The update report shall indicate progress achieved and activities commenced or completed within the last month.

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Architect and Owner and shall be delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

٠..

- § 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved by the <a href="Owner and Architect.">Owner and Architect.</a>
- § 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Owner and the Architect in writing of such deviation at the time of submittal and (1) the Architect has Owner and Architect have given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Owner's or Architect's approval thereof.

#### **PAGE 22**

§ 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner will and the Architect will-shall specify all performance and

design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings. calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Owner and Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Owner and Architect have Architect has specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review, approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

The Contractor-shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment. § 3.13.1 The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment

§ 3.13.2 General Use. The Contractor shall enforce the Owner's instructions regarding the conduct and use of the site by his employees.

§ 3.13.3 Property Corners. Existing property corners on the site shall be replaced by a registered Land Surveyor at the Contractor's expense.

#### § 3.13.4 Parking & Traffic.

.1 Parking of construction vehicles on the site by the Contractor shall not inhibit construction nor prevent access for emergency or other official vehicles. Parking of private vehicles on the site by the Contractor is prohibited unless said vehicle is necessary in the execution of the Contract. No construction vehicles shall be parked near or under any existing vegetation on the site.

.2 Construction traffic and staging shall be permitted only within construction limits as indicated on plan. The contractor is responsible for repair of any areas disturbed outside of this area, including grading and sodding. No staging will be permitted on the existing asphalt.

§ 3.13.5 Fencing. The Contractor will be responsible for erecting and maintaining all construction fencing required by applicable law, regulation, rule, ordinance or code at all times of construction. Failure to erect or maintain this fencing will result in the correction of the problem by the Owner at the expense of the Contractor. The Contractor's expense will be back charged to the contract, and may include, but are not limited to, the cost of any materials and staff time. Required fencing must be installed and fully erected before construction operations beginning and tied-up at the end of each working day. All construction fencing must conform to the Specifications and as required by applicable law. specification.

§ 3.13.6 Water Removal. If, during construction, standing water caused by heavy rains or poor drainage becomes a hazard in the proper execution of the Contract, it shall be the responsibility of the Contractor to provide and make payment for removal of said water to existing drainage swales, storm sewers or other natural or man-made drainage ways.

#### **PAGE 23**

§3.15.3 The Contractor shall walk the site at the close of every work day to assure it is either free of waste material and rubbish, or the waste material and rubbish is secured in a container that is inaccessible to park patrons.

...

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if the Contractor knows or has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Owner and Architect.

•••

§ 3.18.1 To the fullest extent permitted by law the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against law, the Contractor shall waive all right of contribution and shall indemnify and hold harmless the Owner and the Architect and their officers, officials, employees, volunteers and agents from and against all claims, damages, losses and expenses, including but not limited to attorneys' fees, legal fees (attorney's and paralegals fees and court costs), arising out of or resulting from the performance of the Work, Contractor's work, provided that any such claim, damage, loss or expense (i) is attributable to bodily injury, sickness, disease or death, or to-injury to or destruction of tangible property (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the Contractor, a property, other than the work itself, including the loss of use resulting therefrom and (ii) is caused in whole or in part by any wrongful or negligent act or omission of the Contractor, any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts they any of them may be liable, regardless of whether or not such claim, damage, loss or expense it is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.18 otherwise reduce any other right or obligation of indemnity which would otherwise exist as to any party or person described in this Paragraph. Contractor shall similarly protect, indemnify and hold and save harmless the Owner, its officers, officials, employees, volunteers and agents against and from any and all claims, costs, causes, actions and expenses including but not limited to legal fees, incurred by reason of Contractor's breach of any of its obligations under, or Contractor's default of, any provision of the Contract.

In any and all claims against the Owner or the Architect or any of their agents or employees and consultants by any employee of any Subcontractor, anyone directly or indirectly employed by any of them or anyone for whose acts any of them may be liable, the indemnification obligation under this Paragraph shall not be limited in anyway by any limitation on the amount or type of damages, compensation or benefits payable by or for the Contractor or any Subcontractor under workers' or workmen's compensation acts, disability benefit acts or other employee benefit acts.

"Claims," "damages,"" losses" and "expenses" as these words are used in this Agreement shall be construed to include, but not limited to (1) injury or damage consequent upon the failure of or use or misuse by Contractor, its Subcontractors, agents, servants or employees, of any hoist, rigging, blocking, scaffolding, or any and all other kinds of items of equipment or other mechanical or structural contrivance erected or constructed by any person, or any or all other kinds of equipment whether or not the same be owned, furnished or loaned by Owner; (2) all attorneys' fees and costs incurred in bringing an action to enforce the provisions of this indemnity or any other indemnity contained herein; (3) (3) all costs and expenses incurred by the indemnified party and (4) error or omission or defect in any submission made to Architect / Architect for its approval or review.

## PAGE 24

§ 3.18.3 The indemnification obligations of the Contractor under this Contract are limited only to the extent required under the Construction Contract Indemnification for Negligence Act (740 ILCS 35/0.01 et seq.).

§ 4.1.2 Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner, Contractor Owner and Architect. Consent shall not be unreasonably withheld.

- § 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate for Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.
- § 4.2.2 The Architect as a representative of the Owner, will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, to become generally familiar with to supervise and to keep the Owner informed about the progress and quality of the portion of the Work completed, to endeavor to guard the Owner against defects and deficiencies in the Work, and to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.

#### **PAGE 25**

Except as otherwise provided in the Contract Documents or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate with each other through the Architect about matters arising out of or relating to the Contract. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and material suppliers shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.

- § 4.2.5 Based on the Architect's observations and evaluations of the of the progress and quality of Work and Contractor's Applications for Payment, the Architect will review and and, after consultation with Owner, certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.
- § 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Owner and the Architect considers it necessary or advisable, the Architect will have authority the Owner or the Architect will have the responsibility and authority, subject to Owner's approval, to require inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Owner or the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Owner or the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.
- § 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect Owner or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5 and 3.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means,

methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

- § 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Owner and Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4. All Change Orders, Construction Change Directives, and field directives shall require the written approval of Owner in order to be binding on Owner.
- § 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and records, approval, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.

#### **PAGE 26**

- § 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable to the Contractor for results of interpretations or decisions rendered in good faith.
- § 4.2.13 The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.
- § 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's Owner's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

§ 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Architect may reply within 14-30 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to any such proposed person or entity or (2) that the Architect or Owner requires additional time for review. Failure of the Owner or Architect to reply within the 14 day period 30 day period or any extension thereof shall constitute notice of no reasonable objection.

## **PAGE 27**

\$5.2.5 In the event of a conflict between the Owner and Architect regarding the selection of Subcontractors, the Owner's decision shall govern.

By appropriate agreement, written where legally required for validity, written agreement, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate. the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The

Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

...

- § 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that
  - .1 assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor in writing; and
  - assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract. All subcontract agreements shall conform to the requirements of the Contract Documents and the Contractor hereby assigns to Owner (and Owner's permitted assigns) all its interest in any subcontract agreements and purchase orders now existing or hereinafter entered into by the Contractor for performance of any part of the Work, which assignment will be effective in the event of the Contractor's failure to perform the Work in accordance with the Contract Documents and upon acceptance by Owner in writing and only as to those subcontract agreements and purchase orders that Owner designates in said writing. It is agreed and understood that the Owner may accept said assignment at any time during the course of construction prior to Final Completion. Upon such acceptance by Owner, (1) the Contractor shall promptly furnish to the Owner true and correct copies of the designated subcontract agreements and purchase orders, and (2) the Owner shall only be required to compensate the designated Subcontractor(s) or supplier(s) for compensation accruing to such party(ies) for Work done or materials delivered from and after the date on which the Owner accepts the subcontract agreement(s) or purchase order(s). All sums due and owing by the Contractor to the designated Subcontractor(s) or supplier(s) for work performed or material supplied prior to Owner's acceptance of the subcontract agreement(s) or purchase order(s) shall constitute a debt between such parties and the Contractor. It is further agreed that no subcontract agreement or purchase order shall contain any restriction that would prohibit assignment under the terms and conditions stated hereinabove. It is further agreed and understood that such assignment is part of the consideration to Owner for entering into the Contract with the Contractor and may not be withdrawn prior to Final Completion.

\*\*\*

§ 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract.

#### **PAGE 28**

§ 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect and Owner apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.

---

§ 6.2.4 The Contractor shall promptly remedy damage the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or Owner, separate contractors as provided in Section 10.2.5.

§6.2.6 Should the Contractor cause damage to the work or property of any separate contractor and/or in the event of any other claim, dispute, or matter in question between the Contractor and any separate contractor, the Contractor shall promptly attempt to settle with such other contractor by agreement, or otherwise to resolve the dispute. In any event, the Contractor shall indemnify, defend, and hold harmless the Owner, its officers, park commissioners, employees and agents, to the full extent as agreed to under Section 3.18 of these General Conditions.

If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect-will allocate the cost among those responsible.

#### **PAGE 29**

§ 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor and Architect; Architect and Contractor; a Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Architect alone.

...

§ 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Architect and Contractor and Architect stating their agreement upon all of the following:

.3 The extent of the adjustment, if any, in the Contract Time.

Methods used in determining adjustments to the Contract sum may include those listed in Section 7.3.3. Notwithstanding any provision to the contrary in the Contract Documents, the overhead and profit for Contractor and Subcontractors on any Change Order or Construction Change Directive shall be determined as specified in Section 4.5 of the Agreement.

§ 7.3.1 A Construction Change Directive is a written order prepared and signed by the Architect and signed by the Owner and Architect, directed to the Contractor, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

§ 7.3.4 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Construction Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

§ 7.3.7 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect and Owner shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall be limited to the following:

#### PAGE 30

§7.3.11 No change in the Work, whether by way of alteration or addition to the Work, shall be the basis of an addition to the Contract Sum or a change in the Contract Time unless and until such alteration or addition has been authorized by a Change Order executed and issued in accordance with and in strict compliance with the requirements of the Contract Documents and applicable law. Accordingly, no course of conduct or dealing between the parties, nor any express or implied acceptance of alterations or additions to the Work and no claim that the Owner has been unjustly enriched shall be the basis of any claim to an increase in the Contract Sum or change in the Contract Time.

...

## §7.5 CONTINUATION OF WORK PENDING RESOLUTION

Pending final determination of cost to the Owner or extension of time to the Contractor, unless otherwise directed by Owner, Contractor shall continue to perform the Work in accordance with the Contract Documents.

§ 8.1.3 The date of Substantial Completion is the date certified by the Architect and approved by Owner in accordance with Section 9.8.

#### PAGE 31

- § 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.
- § 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time carry the Work forward regularly, diligently, uninterruptedly and expeditiously and in a good workmanlike and professional manner at such a rate of progress and with an adequate work force as will insure the completion of the Work in accordance with the Contract Documents by the date established in the Contract. It is expressly understood and agreed by and between Contractor and Owner that the time for completion of the Work is a reasonable time, taking into consideration the average climatic range, usual industrial conditions, and all other conditions and factors prevailing in this locality.

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a separate contractor employed by the Owner; or by changes ordered in the Work; or by labor disputes, fire, unusual delay in deliveries, unavoidable easualties or other causes beyond the Contractor's control; or by delay authorized by the Owner-pending mediation and arbitration; or by other causes that the Architect determines may justify delay, then progress of the Work prior to Substantial Completion by: any wrongful act or neglect of Owner; changes ordered in the Work which are not caused by the wrongful or negligent acts, errors or omissions of Contractor, its agents, employees or Subcontractors; or by occurrences beyond the control and without the fault or negligence of the Contractor and which by the exercise of reasonable diligence the Contractor is unable to prevent or provide against, including regional labor disputes (not disputes limited to the work force of, or provided by, the Contractor or its Subcontractors) as they affect the Work that cannot be resolved by Contractor's agreeing to the wages, hours, working conditions and other terms as they have been or will be established as the pattern settlement with respect to said dispute, provided that prior to

execution of the Contract by Owner, Contractor has advised Owner in writing of the expiration during the Contract Time of applicable labor contracts; fire, unusual delay in deliveries not reasonably foreseeable, unavoidable casualties, adverse weather conditions not reasonably foreseeable, unusual subsurface or concealed Site conditions which could not have been discovered in the exercise of due diligence and which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the type contemplated in the Contract Documents, or by other occurrences which the Architect, subject to the Owner's approval, determines may justify delay, then, provided that the Contractor is in compliance with all other relevant provisions of the Contract Documents, the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine, or Construction Change Directive for the length of time actually and directly caused by such occurrence as determined by the Architect and approved by the Contractor and Owner; provided, however, that such extension of Contract Time shall be net of any delays caused by or due to the fault or negligence of the Contractor or which are otherwise the responsibility of the Contractor and shall also be net of any contingency or "float" time allowance included in the Contractor's construction schedule. The Contractor shall, in the event of any occurrence likely to cause a delay, cooperate in good faith with the Architect and Owner to minimize and mitigate the impact of any such occurrence and do all things reasonable under the circumstances to achieve this goal.

...

§ 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under other provisions of the Contract Documents. Extension of time provided for the completion of the Work shall be the Contractor's sole remedy for delay (except for the Contractor's right to terminate the Contract pursuant to the provisions of Article 14 hereof), unless the same shall have been caused by acts constituting intentional interference by Owner with Contractor's performance of the Work and where to the extent that such acts of the Owner continue after Contractor's written notice to Owner of such interference. The Owner's exercise of any of its rights under the Contract, including, without limitation, its rights under Article 7, 'Changes in the Work,' regardless of the extent or number of such Changes, or the Owner's exercise of any of its remedies of suspension of the Work, or requirement of correction or re-execution of any defective Work, shall not under any circumstances be construed as intentional interference with Contractor's performance of the Work.

#### **PAGE 32**

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit to the Architect, Architect and Owner, before the first Application for Payment, a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Architect and Owner may require. This schedule, unless objected to by the Architect, Architect or Owner, shall be used as a basis for reviewing the Contractor's Applications for Payment.

§ 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, 9.2., for completed portions of the Work. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents. Until Final Completion, the Owner will pay ninety percent (90%) of the amount due the Contractor on account of progress payments. No interest will be paid on retention amounts. Provided, however, that after the Work is seventy percent (70%) complete, Owner may, without reduction of previous retainage, determine to pay Contractor remaining progress payments for each work category in full.

§9.3.1.3 The following forms must be used for pay requests: (1) AIA Application & Certificate for Payment (G702 & G703); and (2) a Sworn Statement from Contractor and Subcontractor to Owner.

§9.3.1.4 When the contract work has been awarded on a unit price bid basis, the form of each application shall follow the Bid Proposal Form, listing each item number, the total quantity of units completed to date of the estimate, the unit price and subtotal. The subtotal column shall be added to show the total cost of work completed to date, less ten

(10%) percent to be withheld giving the total amount requested for payment. Previous applications for payment paid by the Owner shall be shown on each subsequent request and subtracted after the ten (10%) percent has been withheld.

§ 9.3.1.5 Failure to supply waivers of lien or acceptable evidence of payment of all current accounts incurred by this Contract work will be considered grounds for withholding final payment.

#### **PAGE 33**

- § 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, and subject to Owner's approval, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Section 9.5.1.
- § 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, and not Contractor, based on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation to the Owner and not the Contractor, that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.
- All applications for payment shall be accompanied by affidavits, in triplicate, from the Contractor and Subcontractors containing such information and in such form as to comply with the Illinois Mechanics Lien Act (770 ILCS 60/0.01 et seq.) and showing in detail the sources of all labor and materials used and contracted to be used on the Project, including names and addresses of subcontractors and material suppliers; amounts paid and remaining to be paid to each; lien waivers in a form acceptable to Owner; together with all other documents as shall be necessary, in the sole judgment of the Architect and Owner, to waive all claims of liens to date and comply with all applicable state and local laws.

§ 9.5.1 The Architect Architect, after consultation with Owner, may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because efof, but not limited to:

#### PAGE 34

§9.5.4 No interest will be paid on payments withheld.

§ 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents and shall so notify the Architect. .

§ 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Architect shall-The Owner shall not have an obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law.

§9.6.8 Anything to the contrary contained or implied herein notwithstanding, no progress payment need be made by Owner until such time as Contractor, Subcontractors or any other persons performing the Work or furnishing materials or equipment for the Project furnishes such documents as Owner may reasonably require (including without limitation sworn notarized contractor's statements, affidavits and waivers of lien).

If the Architect does not issue a recommendation for a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner either does not reject the Architect's recommendations for the Certificate of Payment or does not pay the Contractor within seven-10 days after the date established in the Contract Documents the amount certified by the Architect or awarded by binding dispute resolution, accepted for payment by the Owner, then the Contractor may, upon seven additional days' written notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. Architect stop the Work. .. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut-down, delay incurred of shutdown, and start-up, plus interest as provided for in the Contract Documents.

#### **PAGE 35**

- § 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect and Owner a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.
- § 9.8.3 Upon receipt of the Contractor's list, the Owner and Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Owner's and Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Owner and the Architect to determine Substantial Completion.
- § 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Final Completion of the Work or designated portion thereof unless otherwise provided in the Contract Documents or the Certificate of Substantial Completion.

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer as required under Section 11.3.1.5 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect and Owner as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.

...

§ 9.10.1 Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect finds find the Work acceptable under the Contract Documents and the Contract fully performed, and the Architect has advised Owner of that finding and Owner has not advised Architect of any objection to such finding, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation to Owner and not Contractor, that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled. Owner's failure to object to and Owner's acceptance of Architect's findings hereunder shall not limit Architect's obligation to properly perform his duties under the Contract Documents and shall not constitute Owner's acceptance of Work not complying with the requirements of the Contract Documents or Owner's waiver of any claims or remedies it may have with respect to any such defective or delayed Work.

#### **PAGE 36**

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims. The final payment by Owner—shall not relieve the Contractor of the responsibility for the correction of any and all defects in the work performed. Contractor shall correct all defects as notified for the applicable warranty period after final payment.

- § 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from
  - .1 liens, Claims, security interests or encumbrances arising out of the Contract and unsettled;
  - 2 failure of the Work to comply with the requirements of the Contract Documents; or
- .3 terms of special warranties required by the Contract Documents. Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

§ 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

. . .

§ 10.2.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their

protection from damage, injury or loss. Contractor shall be responsible for security of all tools, materials, and equipment left on site.

#### **PAGE 37**

- § 10.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, such activities shall only be done with written consent of Owner and the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- § 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose wrongful acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.
- § 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect. The person designated as responsible for prevention of accidents shall hold regularly scheduled meetings with representatives of Subcontractors, and in the event of separate contracts, hold meetings with other contractors, to promote compliance with governing safety regulations.

§ 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or the Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have has no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Contractor's reasonable additional costs of shut-down, delay and start-up.

#### **PAGE 38**

## § 11.1 CONTRACTOR'S LIABILITY INSURANCE

§ 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- .1 Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed;
- -Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees;
- Claims for damages because of bodily injury, sickness or disease, or death of any person other than the Contractor's employees;
- Claims for damages insured by usual-personal injury liability coverage;

- Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle;
- Claims for bodily injury or property damage arising-out of completed operations; and
- -Claims involving contractual liability insurance applicable to the Contractor's obligations under Section 3.18.

§ 11.1.2 The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims made basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment, and, with respect to the Contractor's completed operations coverage, until the expiration of the period for correction of Work or for such other period for maintenance of completed operations coverage as specified in the Contract Documents.

§ 11.1.3 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance. These certificates and the insurance policies required by this Section 11.1 shall contain a provision that coverages afforded under the policies will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner. An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the final Application for Payment as required by Section 9.10.2 and thereafter upon renewal or replacement of such coverage until the expiration of the time required by Section 11.1.2. Information concerning reduction of coverage on account of revised limits or claims paid under the General-Aggregate, or both, shall be furnished by the Contractor with reasonable promptness.

§ 11.1.4 The Contractor shall cause the commercial liability coverage required by the Contract Documents to include (1) the Owner, the Architect and the Architect's consultants as additional insureds for claims-caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's operations; and (2) the Owner as an additional insured for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's completed operations.

#### § 11.2 OWNER'S LIABILITY INSURANCE

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance.

§11.1 CONTRACTOR'S INSURANCE REQUIREMENTS. Contractor shall procure and maintain for the duration of the contract, insurance against claims for death, injuries, sickness to persons, or damages to property which may arise from or in connection with the performance of work hereunder by the Contractor, his agents, representatives, employees or subcontractors, anyone directly or indirectly employed by them or anyone for whose acts any of them may be liable, of the types and in the amounts listed below.

§11.1.1 Commercial General and Umbrella Liability Insurance. Contractor shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less that \$1,000,000 each occurrence. If such CGL insurance contains a general aggregate limit, it shall apply separately to this project/location. CGL insurance shall be written on Insurance Services Office (ISO) occurrence form CG 00 01, or a substitute form providing equivalent coverage, and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract (including the tort liability of another assumed in a business contract). Owner, its elected and appointed officials, employees, agents and volunteers, and Architect shall be included as an insured under the CGL, using ISO additional insured endorsement CG 20 10 or a substitute providing equivalent coverage, and under the commercial umbrella, if any. This insurance shall apply as primary insurance with respect to any other insurance or self-insurance afforded to Owner and Architect. . There shall be no endorsement or modification of the CGL limiting the scope of coverage for liability arising from pollution, explosion, collapse, or underground property damage.

§11.1.2 Continuing Completed Operations Liability Insurance. Contractor shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella liability insurance with a limit of not less than \$1,000,000 each occurrence for at least three years following substantial completion of the work. Continuing CGL insurance shall be written on ISO occurrence form CG 00 01, or substitute form providing equivalent coverage, and shall, at minimum, cover liability arising from products-completed operations and liability assumed under an insured contract. Continuing CGL insurance shall have a products-completed operations aggregate of at least two times its each occurrence limit. Continuing commercial umbrella coverage, if any, shall include liability coverage for damage to the insured's completed work equivalent to that provided under ISO form CG 00 01.

- §11.1.3 Business Auto and Umbrella Liability Insurance. Contractor shall maintain business auto liability and, if necessary, commercial umbrella liability insurance with a limit of not less than \$1,000,000 each accident. Such insurance shall cover liability arising out of any auto including owned, hired and non-owned autos. Business auto insurance shall be written on Insurance Services Office (ISO) form CA 00 01, CA 00 05, CA 00 12, CA 00 20, or a substitute form providing equivalent liability coverage. If necessary, the policy shall be endorsed to provide contractual liability coverage equivalent to that provided in the 1990 and later editions of CA 00 01.
- §11.1.4 Workers Compensation Insurance. Contractor shall maintain workers compensation as required by statute and employers liability insurance. The commercial umbrella and/or employers liability limits shall not be less than \$1,000,000 each accident for bodily injury by accident or \$1,000,000 each employee for bodily injury by disease. If Owner has not been included as an insured under the CGL using ISO additional insured endorsement CG 20 10 under the Commercial General and Umbrella Liability Insurance required in this Contract, the Contractor waives all rights against Owner and its officers, officials, employees, volunteers and agents for recovery of damages arising out of or incident to the Contractors work.
- §11.1.5 Contractor's Obligation to Insure for Bodily Injury Claims. In addition to the above, the Owner will require all Contractor's to purchase insurance to cover claims and expenses asserted against Architect, its employees and consultants for bodily injury, sickness, disease, or death cause by any negligent act or omission of the Contractor, any Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts any of them may be liable.

#### §11.1.6 General Insurance Provisions

- .1 Evidence of Insurance Prior to beginning work, Contractor shall furnish Owner with a certificate(s) of insurance and applicable policy endorsement(s), executed by a duly authorized representative of each insurer, showing compliance with the insurance requirements set forth above. All certificates shall provide for 30 days written notice to Owner prior to the cancellation or material change of any insurance referred to therein. Written notice to Owner shall be by certified mail, return receipt requested. Failure of Owner to demand such certificate, endorsement or other evidence of full compliance with these insurance requirements or failure of Owner to identify a deficiency from evidence that is provided shall not be construed as a waiver of Contractor's obligation to maintain such insurance. Owner shall have the right, but not the obligation, of prohibiting Contractor or any subcontractor from entering the project site until such certificates or other evidence that insurance has been placed in complete compliance with these requirements is received and approved by Owner. Failure to maintain the required insurance may result in termination of this Contract at Owner's option. With respect to insurance maintained after final payment in compliance with a requirement above, an additional certificate(s) evidencing such coverage shall be promptly provided to Owner whenever requested. Contractor shall provide certified copies of all insurance policies required above within 10 days of Owner's written request for said copies.
- . 2 Acceptability of Insurers. For insurance companies which obtain a rating from A.M. Best, that rating should be no less than A VII using the most recent edition of the A.M. Bests Key Rating Guide. If the Bests rating is less than A VII or a Best's rating is not obtained, the Owner has the right to reject insurance written by an insurer it deems unacceptable.
- .3 Cross-Liability Coverage. If Contractor's liability policies do not contain the standard ISO separation of insureds provision, or a substantially similar clause, they shall be endorsed to provide cross-liability coverage.
- .4 Deductibles and Self-Insured Retentions. Any deductibles or self-insured retentions must be declared to the At the option of the Owner, the Contractor may be asked to eliminate such deductibles or self-insured retentions as respects the Owner, its officers, officials, employees, volunteers and agents or required to procure a bond guaranteeing payment of losses and other related costs including but not limited to investigations, claim administration and defense expenses.

.5 Subcontractors. Contractor shall cause each subcontractor employed by Contractor to purchase and maintain insurance of the type specified above. When requested by the Owner, Contractor shall furnish copies of certificates of insurance evidencing coverage for each subcontractor.

## §11.2 PERFORMANCE AND PAYMENT BONDS

- § 11.2.1 The Contractor shall deposit with the Owner before commencing any Work an AIA A312-2010

  Performance Bond and Payment Bond for 110% of the Contract Sum, guaranteeing the faithful performance of the work in accordance with the Contract, the payment of all indebtedness incurred for labor and materials, payment of the prevailing wage in accordance with paragraph 13.8.1, and guarantee correction of work. The Surety must be approved by the Owner, and be licensed to conduct business in the State of Illinois and be named in the current list of "Surety Companies Acceptable on Federal Bonds" as published in the Federal Register by the Audit Staff Bureau of Accounts, U.S. Treasury.
- § 11.2.2 The Contractor and all subcontractors shall name the Owner as an obligee on all bonds.
- § 11.3.1 Unless otherwise provided, the Owner shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance written on a builder's risk "all risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project Sum as well as subsequent modifications thereto for the entire Work at the site on a replacement cost basis without optional voluntary deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section Paragraph 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Section Paragraph 11.3 to be covered, whichever is later. carlier. This insurance shall include the respective interests of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Project. Work.
- § 11.3.1.1 Property insurance shall be on an "all risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) a course of construction policy form and shall insure against the perils of fire and extended coverage and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, false work, windstorm, testing and start-up, temporary buildings and debris removal, including demolition, and shall cover reasonable compensation for Architect's and Contractor's the Architect's, any of the Owner's Consultant's services and expenses required as a result of such insured loss. Owner shall not be required to provide coverage for other perils unless otherwise provided in the Contract Documents. Property insurance provided by the Owner shall not cover Contractor's, Subcontractor's or Sub-subcontractor's liability or any tools, apparatus, machinery, scaffolding, hoists, forms, staging, shoring or other similar items commonly referred to as construction equipment, which may be on the site and the capital value of which is not included in the Work. The Contractor shall make his own arrangements for any insurance he may require on such construction equipment.
- § 11.3.1.2 If the Owner does not intend to purchase such property insurance required by the Contract and with all of the coverages in the amount described above, the Owner shall so inform the Contractor in writing prior to commencement of the Work. The Contractor may then effect insurance that The Contractor shall effect insurance which will protect the interests of the Contractor, Subcontractors and Sub-subcontractors in the Work, and by appropriate Change Order the cost thereof shall be charged to the Owner. If the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain insurance as described above, without so notifying the Contractor in writing, then the Owner shall bear all reasonable costs properly attributable thereto. Work.
- § 11.3.1.3 If the property insurance requires deductibles, minimum deductibles and such deductibles are identified in the Contract Documents, the Contractor shall pay costs not covered because of deductibles. If the Owner or insurer increases the required minimum deductibles above the amounts so identified or if the Owner elects to purchase this insurance with voluntary deductible amounts, the Owner shall be responsible for payment of the additional costs not

covered because of such increased or voluntary deductibles. If deductibles are not identified in the Contract Documents, the Owner shall pay costs not covered because of such deductibles.

§ 11.3.1.4 This Unless otherwise provided in the Contract Documents, this property insurance shall cover portions of the Work stored off the site, the site and paid for by Owner after written approval of the Owner at the value established in the approval, and also portions of the Work in transit and paid for by Owner.

#### **PAGE 41**

The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner; this insurance shall include interests of the Owner, Contractor, Contractor Subcontractors and Sub-subcontractors in the Work, and the Owner and the Contractor shall be named insureds.

#### § 11.3.3 LOSS OF USE INSURANCE

The Owner, at the Owner's option, may purchase and maintain such insurance as will insure the Owner against loss of use of the Owner's property due to fire or other hazards; however caused. The Owner waives all rights of action against the Contractor for loss of use of the Owner's property, including consequential losses due to fire or other hazards however caused.

§ 11.3.5 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, the Owner-shall waive all rights in accordance with the terms of Section 11.3.7 for damages caused by fire or other causes of loss covered by this separate property insurance. All separate policies shall provide this waiver of subrogation by endorsement or otherwise.

§ 11.3.6 Before an exposure to loss may occur, the Owner shall file with the Contractor a copy of each policy that includes insurance coverages required by this Section 11.3. Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project. Each policy shall contain a provision that the policy will not be canceled or allowed to expire, and that its limits will not be reduced, until at least 30 days? prior written notice has been given to the Contractor.

## § 11.3.7 WAIVERS OF SUBROGATION

The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents and employees, each of the other, and (2) the Architect, Architect's consultants, separate contractors described in Article 6, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for damages caused by fire or other causes of loss to the extent covered by property insurance obtained pursuant to this Section 11.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Owner as fiduciary. The Owner or Contractor, as appropriate, shall require of the Architect, Architect's consultants, separate contractors described in Article 6, if any, and the subcontractors, sub subcontractors, agents and employees of any of them, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endersement or otherwise. A waiver of subrogation shall be effective as to a person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged.

§ 11.3.8 A loss insured under the Owner's property insurance shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11:3.10. The Contractor shall pay Subcontractors their just shares of insurance proceeds received by the Contractor, and by appropriate agreements, written where legally required for validity, shall require Subcontractors to make payments to their Sub-subcontractors in similar manner.

- § 11.3.9 If required in writing by a party in interest, the Owner as fiduciary shall, upon occurrence of an insured loss, give bond for proper performance of the Owner's duties. The cost of required bonds shall be charged against proceeds received as fiduciary. The Owner shall deposit in a separate account proceeds so received, which the Owner shall distribute in accordance with such agreement as the parties in interest may reach, or as determined in accordance with the method of binding dispute resolution selected in the Agreement between the Owner and Contractor. If after such loss no other special agreement is made and unless the Owner terminates the Contract for convenience, replacement of damaged property shall be performed by the Contractor after notification of a Change in the Work in accordance with Article 7.
- § 11.3.10 The Owner as fiduciary shall have power to adjust and settle a loss with insurers unless one of the parties in interest shall object in writing within five days after occurrence of loss to the Owner's exercise of this power; if such objection is made, the dispute shall be resolved in the manner selected by the Owner and Contractor as the method of binding dispute resolution in the Agreement. If the Owner and Contractor have selected arbitration as the method of binding dispute resolution, the Owner as fiduciary shall make settlement with insurers or, in the case of a dispute over distribution of insurance proceeds, in accordance with the directions of the arbitrators.

#### § 11.4 PERFORMANCE-BOND-AND-PAYMENT-BOND

- § 11.4.1 The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.
- § 11.4.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.
- §11.3.11 Notwithstanding any provision contained in Section 11.3 including paragraphs 11.3.1 through and including 11.3.11, the Owner's obligation to purchase insurance shall herein be deemed satisfied by the Owner's membership in a self-insured risk management agency or pool. The Contractor agrees that any obligation the Owner has to purchase property insurance shall be satisfied by the Owner's membership in a self-insured risk management agency or pool. The Contractor further agrees that it will only have rights allowable to it under any coverage provided through the Owner's membership in a self-insured risk management agency or pool.

...

- § 12.1.1 If a portion of the Work is covered contrary to the Owner's or Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Owner or the Architect, be uncovered for the Owner's and Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.
- § 12.1.2 If a portion of the Work has been covered that the Owner or the Architect has not specifically requested to examine prior to its being covered, the Architect the Architect, with the consent of the Owner, may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

...

The Contractor shall promptly correct Work rejected by the Owner or Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Centractor a written acceptance of such condition. so. The Owner shall give such notice promptly after discovery of the condition. During the one-year-applicable period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.4. Notwithstanding the foregoing, Contractor shall correct Work deficiently or defectively performed, and replace defective or nonconforming materials, even though such deficiency, defect or nonconformity may be discovered more than one year after final completion, if the correction is of a latent defect and arises from poor workmanship or improper materials or is required to be made to workmanship or materials covered by Contractor or Subcontractors contrary to the Architect's request or to requirements specifically expressed in the Contract Documents and was therefore not visible for inspection by Architect or Owner at the time the Work was performed.

#### **PAGE 42**

§ 12.2.2.3 The one year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2.

•••

The Contract shall be governed by the law of the place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4.located.

...

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to covenants, agreements and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract. This Contract is nonassignable in whole or in part by Contractor, and an assignment shall be void without the prior written consent of Owner, which consent, shall not be unreasonably withheld.

#### **PAGE 43**

§ 13.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Owner and Architect timely notice of when and where tests and inspections are to be made so that the Owner and Architect may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor.

§ 13.5.2 If the Owner, Architect, Owner-or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Section 13.5.1, the Owner or Architect

will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Owner and Architect of when and where tests and inspections are to be made so that the Owner and Architect may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.

§ 13.5.4 Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Owner and Architect.

...

§ 13.5.7 Retests. The cost of a retest will be borne by the party requesting the retest, unless the retest shows that the original test or the Work being tested was in error or defective, and in such event, the cost of the retest shall be borne by the other party.

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located the legal rate established in the Illinois Local Governmental Prompt Payment Act.

...

The Owner and Contractor shall commence all claims and causes of action, whether in contract, tort, breach of warranty or otherwise, against the other arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in the Agreement within the time period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all claims and causes of action not commenced in accordance with this Section 13.7. law.

## § 13.8. COMPLIANCE WITH LAWS

Contractor shall abide by and comply with all applicable Federal, State and local laws and rules and regulations including without limitation those relating to 1) fair employment practices, affirmative action and prohibiting discrimination in employment; 2) workers' compensation; 3) workplace safety; 4) wages and claims of laborers, mechanics and other workers, agents, or servants in any manner employed in connection with contracts involving public funds or the development or construction of public works, buildings or facilities; and 5) steel products procurement.

§ 13.8.1 As a condition of the award of the Contract to contractor, Contractor shall certify, affirm and agree as follows, which certifications, affirmations and agreements shall be incorporated in and hereunder as a part of the Contract:

The Contractor shall comply with the requirements of the Illinois Prevailing Wage Act (820 ILCS 130/0.01 et seq.) and the Owner's Ordinances requiring payment of prevailing wages. The Contractor shall pay or cause to be paid not less than the prevailing rate of hourly wage in the county the work is performed as determined by the Illinois Department of Labor for the month in which the work is performed including but not limited to all laborers, workers and mechanics. All contractors and subcontractors rendering services under this contract must comply with all requirements under the Act, including but not limited to, all wage, notice and record keeping duties.

The Contractor is required to verify current prevailing wage prior to the first day of each month and to pay the then-current prevailing wage rate as determined by the Illinois Department of Labor, regardless of the rates contained in the Contract Documents. Any increases in costs to the Contractor due to the changes in the prevailing wage during the term of this Contract shall be at the expense of Contractor and not at the expense of Owner. Current prevailing wage rates are published at the following website: https://www.illinois.gov/idol/Laws-Rules/CONMED/Rates/2015/july/COUNTY.HTM. agrees to indemnify and hold harmless the Owner for any violations of the Prevailing Wage Act.

The Contractor shall also: (1) insert into each subcontract and the project specifications for each subcontract, a written stipulation that the subcontractor shall not pay less than the prevailing rate of hourly wage to all laborers, workers, and mechanics performing work under the contract; and (2) require each subcontractor to insert into each lower-tiered contract and the project specifications for each lower-tiered subcontract, a stipulation that the subcontractor shall not pay less than prevailing rate of hourly wage to all laborers, workers, and mechanics performing work under the contract.

The Contractor shall include on all bonds and shall cause all subcontractors' bonds required under the Contract Documents to guarantee compliance with the Prevailing Wage Act.

Additionally, the Contractor and each subcontractor shall make and keep, for a period of not less than five (5) years from the date of the last payment on a contract or subcontract, records of all laborers, mechanics, and other workers employed by them on the Project; the records shall include each worker's name, address, telephone number when available, social security number, classification or classifications, the hourly wages paid in each pay period, the number of hours worked each day, and the starting and ending times of work each day. The Contractor shall submit monthly, no later than the 10th day of each calendar month, in person, by mail, or electronically a certified payroll to the Owner with each monthly pay request in the form attached to the Contract Documents. The certified payroll shall be accompanied by a statement signed by the Contractor or subcontractor which states that: (i) he or she has examined the certified payroll and such records are true and accurate; (ii) the hourly rate paid to each worker is not less than the general prevailing rate of hourly wages required by this Act; and (iii) the Contractor or subcontractor is aware that filing a certified payroll that he or she knows to be false is a Class A misdemeanor. The Contractor may rely on the certification of a lower tier subcontractor, provided the Contractor does not knowingly rely upon a subcontractor's false certification. The records submitted in accordance with this payroll submittal provision shall be considered public records pursuant to Section 5 of the Prevailing Wage Act, 820 ILCS 130/5 (2004, as amended by P.A. 94-515). The Owner may, at its option, immediately terminate the Contract in the event that Contractor violates any provision of this paragraph or the Prevailing Wage Act.

Contractor shall also post the prevailing wage rates for each craft or type of worker or mechanic needed to complete the project at either: (1) a location on the project site easily accessible to the workers engaged on the project; or (2) in lieu of posting on the project site, if the Contractor has a business location where laborers, workers, ands mechanics may regularly visit, the Contractor may either post the prevailing rate of wages in each county the Contractor works in a conspicuous location or provide the laborers, workers or mechanics engaged on the project a written notice indicating the prevailing rate of wages for the project.

Upon seven business days notice, the Contractor and each subcontractor shall make available for inspection and copying at a location within this State during reasonable hours, the records identified in 820 ILCS 130/5(a)(1) to the Owner, and its officers and agents.

- .2 To the best of Contractor's knowledge, no officer or employee of Contractor has been convicted of bribery or attempting to bribe an officer or employee of the State of Illinois, or any unit of local government, nor has any officer or employee made an admission of guilt of such conduct which is a matter of record.
- .3 Contractor is not barred from bidding on or entering into public contracts due to having been convicted of bid-rigging or bid rotating under paragraphs 33E-3 or 33E-4 of the Illinois Criminal Code. Contractor also certifies that no officers or employees of the Contractor have been so convicted and that Contractor is not the successor company or a new company created by the officers or owners of one so convicted. Contractor further certifies that any such conviction occurring after the date of this certification will be reported to the Owner, immediately in writing, if it occurs during the bidding process, or otherwise prior to entering into the Contract therewith.
- .4 Pursuant to the Illinois Human Rights Act (775 ILCS 5/2-105), Contractor has a written sexual harassment policy that includes, at a minimum, the following information: (i) a statement on the illegality of sexual harassment; (ii) the definition of sexual harassment under State law; (iii) a description of sexual harassment utilizing examples; (iv) the Contractor's internal complaint process including penalties; (v) the legal recourse, investigative and complaint process available through the Illinois Department of Human Rights and the Human Rights Commission and directions on how to contact both; and (vi)

User Notes:

protection against retaliation as provided by Section 6-101 of the Illinois Human Rights Act. Contractor further certifies that such policy shall remain in full force and effect. A copy of the policy shall be provided to the Illinois Department of Human Rights upon request. Contractor shall abide by the "Employment of Illinois Workers on Public Works Act" (30 ILCS 570/0.01 et. seq.) which stipulates that whenever there is a period of excessive unemployment in Illinois, defined as any month immediately following two (2) consecutive calendar months during which the level of unemployment in Illinois exceeds five percent (5%) as measured by the U.S. Bureau of Labor Statistics in its monthly publication of employment and unemployment figures, the Contractor shall employ only Illinois laborers unless otherwise exempted as so stated in the Act. ("Illinois laborer" means any person who has resided in Illinois for at least 30 days and intends to become or remain an Illinois resident). Other laborers may be used if Illinois laborers are not available or are incapable of performing the particular type of work involved if so certified by the Contractor and approved by the Architect. (i) Contractor's bid proposal was made without any connection or common interest in the profits anticipated to be derived from the Contract by Contractor with any other persons submitting any bid or proposal for the Contract; (ii) the Contract terms are in all respects fair and the Contract will be entered into by Contractor without collusion or fraud; and (iii) no official, officer or employee of the Owner has any direct or indirect financial interest in Contractor's bid proposal or in Contractor. Contractor knows and understands the Equal Employment Opportunity Clause administrated by the Illinois Department of Human Rights, which is incorporated herein by this reference, and agrees to comply with the provisions thereof. Contractor further certifies that Contractor is an "equal opportunity employer" as defined by Section 2000 (e) of Chapter 21, Title 42 of the United States Code Annotated and Executive Orders #11246 and #11375 as amended, which are incorporated herein by this reference. Neither Contractor nor any substantially owned affiliated company is participating or shall participate in an international boycott in violation of the provisions of the U.S. Export Administration Act of 1979 or the regulations of the U.S. Department of Commerce promulgated under that Act. Contractor is not barred from contracting with the Owner because of any delinquency in the payment of any tax administrated by the Illinois Department of Revenue, unless it is being contested. Contractor further certifies that it understands that making a false statement regarding delinquency in taxes is a Class A. misdemeanor and, in addition, voids the Contract and allows the Owner, a municipal entity, to recover in a civil action all amounts paid to the Contractor. If Contractor has 25 or more employees at the time of letting of the Contract, Contractor knows, understands and acknowledges its obligations under the Illinois Drug Free Workplace Act (30 ILCS 580/1 et seq.) and certifies that it will provide a drug-free workplace by taking the actions required under, and otherwise implementing on a continuing basis, Section 3 of the Drug Free Workplace Act. Contractor further certifies that it has not been debarred and is not ineligible for award of this Contract as the result of a violation of the Illinois Drug Free Workplace Act. The Contractor shall cause all Subcontractors to comply with the requirements and provisions of the Illinois Substance Abuse Prevention on Public Works Projects Act (820 ILCS 265/1 et seq.) (the "Act") by: Prohibiting the use, possession, distribution or delivery of any drug or alcohol (as defined under the Act) or allowing any employee to be under the influence of any said drug or alcohol while performing the Work;

following:

**User Notes:** 

Filing a written substance abuse prevention program with the Owner for the

prevention of substance abuse among its employees prior to the commencement of the Work. Said program shall be available to the general public and, at a minimum, contain the

- i A minimum requirement of a 9 panel urine drug test plus a test for alcohol. Testing an employee's blood may only be used for post-accident testing, however, blood testing is not mandatory for the employer where a urine test is sufficient;
- ii. A prohibition against the actions for the use, possession, distribution or delivery of any drug or alcohol (as defined under the Act) or any employee under the influence of any said drug or alcohol while performing the Work;
- iii. A requirement that employees performing the Work submit to pre-hire, random, reasonable suspicion, and post-accident drug and alcohol testing. Testing of an employee before commencement of the Work is not required if the employee participated in a random testing program during the ninety (90) days preceding the date on which the employee commenced work hereunder; and
- iv. A procedure for notifying an employee that he or she may not perform any of the Work if he or she: 1) uses, possess, delivers or is under the influence of a drug or alcohol as prohibited under the Act; 2) tests positive for the presence of a drug as outlined in the Act; or 3) refuses to submit to drug or alcohol testing as required under the Contractor's substance abuse program until the employee tests negative for the presence of drugs or alcohol as outlined in the Act or has been approved to commence or return to work in accordance with the Contractor's substance abuse program.
- c. Immediately removing and/or prohibiting access to the Work site of any employee who: 1) uses, possess, delivers or is under the influence of a drug or alcohol as prohibited under the Act; 2) tests positive for the presence of a drug as outlined in the Act; or 3) refuses to submit to drug or alcohol testing as required under the Contractor's substance abuse program. Said employee shall be prohibited from the Work site until he or she tests negative for the presence of drugs or alcohol as outlined in the Act or has been approved to commence or return to work in accordance with the Contractor's substance abuse program; and
  - d. Complying with all other requirements of the Act.
- .12 The Contractor hereby acknowledges that pursuant to 5 ILCS 140/7(2) any record of the Contractor that relates directly to a governmental function being performed by the Contractor pursuant to this Agreement is considered a public record of the Owner for purposes of the Freedom of Information Act (FOIA), and upon request of the Owner's FOIA Officer, Contractor shall within two (2) business days turn over to the FOIA Officer any record requested that is in possession of the Contractor.
- § 13.8.2 Contractor and any subcontractor shall keep and maintain accurate books of record and account, in accordance with sound accounting principles, of all expenditures made and all costs, liabilities and obligations incurred under this Contract, and all papers, files, accounts, reports, cost proposals with backup data and all other material relating to work under this Contract and shall make all such materials available at the office of the Owner at any reasonable time during the term of this contract and for the length of time established by law or five (5) years, whichever is longer from the date of final payment to Contractor or termination of this Contract for audit, inspection and copying upon Owner's request.
- § 13.8.3 The Contractor shall be required to remain for the entirety of the Contract in compliance with the foregoing legal requirements. A violation is grounds for the immediate termination of the Contractor for cause. However, any forbearance in delay by the Owner in terminating Contractor or canceling the Contract shall not constitute a waiver of any right the Owner may have, including without limitation termination of Contractor, cancellation of the Contract and recovery of damages.

#### §13.8.4 Record Keeping.

§13.8.4.1 Contractor and any subcontractor shall keep and maintain accurate books of record and account, in accordance with sound accounting principles, of all expenditures made and all costs, liabilities and obligations incurred under this Contract, and all papers, files, accounts, reports, cost proposals with backup data and all other material relating to work under this Contract and shall make all such materials available at the office of the Park

District Executive Director at any reasonable time during the term of this contract and for the length of time established by law or five (5) years, whichever is longer from the date of final payment to Contractor or termination of this Contract for audit, inspection and copying upon Owner's request.

#### **PAGE 47**

§ 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30-90 consecutive days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons:

- An act of government, such as a declaration of national emergency that requires all Work to be stopped;stopped;or
- .3 Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents; orDocuments.

- § 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, including reasonable overhead and profit, costs incurred by reason of such termination, and damages.
- § 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

#### **PAGE 48**

- § 14.2.1 The Owner may terminate the Contract if the Contractor
  - .1 repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
  - .2 fails to make If the Contractor shall institute proceedings or consent to proceedings requesting relief or arrangement under the Federal Bankruptcy Act or any applicable Federal or State Law, or if a petition under any federal or state insolvency law is filed against the Contractor and such petition is not dismissed within sixty (60) days from the date of said filing, or if the Contractor admits in writing his inability to pay his debts generally as they become due, or if he makes a general assignment for the benefit of his creditors, or if a receiver, liquidator, trustee, or assignee is appointed on account of his bankruptcy or insolvency; or if a receiver of all or any substantial portion of the Contractor's properties is appointed; or if the Contractor abandons the Work; or if he fails, except in cases for which extension of time is provided, to prosecute promptly and diligently the Work or to supply enough properly skilled workmen or proper materials for the Work; or if the Contractor submits an application for payment, sworn statement, waiver of lien, affidavit or document of any nature whatsoever which is intentionally falsified; or if the Contractor fails to make prompt payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors:
  - repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
- otherwise is guilty of substantial breach of a provision of the Contract Documents or otherwise breaches obligations under any subcontract with a Subcontractor; or if a mechanic's or materialman's lien or a notice of lien is filed against any part of the Project and is not promptly bonded or insured over by the Contractor in a manner reasonably satisfactory to the Owner, or if the Contractor disregards any laws, statutes, ordinances, rules, regulations

or orders of any governmental body or public or quasi-public authority having jurisdiction of the Work or the Project premises; or if the Contractor otherwise violates any material provision of the Contract Documents, then, without prejudice to any right or remedy available Owner may, after giving the Contractor seven (7) days' written notice, terminate the employment of the Contractor, and take possession of the Project and of all materials, equipment, tools, construction equipment and machinery thereon owned by the Contractor and accept assignment of Subcontracts and may complete the Work by whatever reasonable method the Owner may deem expedient. If requested by the Owner, the Contractor shall remove any part or all of his equipment, machinery and supplies from the Project within seven (7) days from the date of such request, and in the event of the Contractor's failure to so, the Owner shall have the right to remove or store, or remove and store, such equipment, machinery and supplies at the Contractor's expense. In case of such termination, the Contractor shall not be entitled to receive any further payment for Work performed by the Contractor through the date of termination until final completion of the Work

- § 14.2.2 When any of the above reasons exist, the Owner, upon certification by the Initial Decision Maker that sufficient cause exists to justify such action, may without prejudice to any other rights or remedies of the Owner and after-giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:
  - .1 Exclude the Contractor from the site and take possession of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
  - 2 Accept assignment of subcontracts pursuant to Section 5.4; and
- .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.
- § 14.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished. If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. This obligation for payment shall survive termination of the Contract.
- § 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, upon application, and this obligation for payment shall survive termination of the Contract.
- §14.2.4 The Owner's right to terminate the Contract pursuant to Section 14.2 shall be in addition to and not in limitation of its right to stop the Work without terminating the Contract pursuant to Section 2.3.

## **PAGE 49**

...

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 14.3.1. Adjustment of the Contract Sum shall <u>not</u> include profit. No adjustment shall be made to the extent

§ 14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

Additions and Deletions Report for AIA Document A201<sup>TM</sup> ~ 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 11:11:21 on 10/04/2016 under Order No.1549258185\_1 which expires on 08/24/2017, and is not for resale.

User Notes:

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, "Claim" is a written demand or assertion by the Contractor seeking adjustment to interpretation of Contract terms, payment of money, extension of time or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim arising out of the Contract.

...

Claims by either the Owner or Contractor must be initiated by written notice to the other party and to the Initial Decision Maker Owner with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. Claims by either party must be initiated within 21 Architect. Claims by Contractor must be initiated within 15 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant 15 days after the Contractor first recognizes the condition giving rise to the Claim, whichever is later.

...

Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, Contractor's Claim, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents. The Architect will prepare Change Orders and issue Certificates for Payment in accordance with the decisions of the Initial Decision Maker. the performance required of him under the Contract.

...

§ 15.1.5.3 Unit Prices. The Contractor shall be responsible for notifying the Owner of any discrepancies or additions to work items completed on a unit price basis. This notification must take place prior to the execution of the Work. The purpose of this requirement is to make sure the Owner is aware of the extra items affecting the cost of the original contract amount. Discrepancies in the multiplication of units of work and unit prices will be resolved in favor of the unit prices. Discrepancies between the indicated sum of any column of figures and the corrected sum thereof will be resolved in favor of the corrected sum.

## § 15.1.6 CLAIMS FOR CONSEQUENTIAL DAMAGESCLAIMS

The Contractor and Owner waive Claims against each other waives Claims against the Owner for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- .1 damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- 2 ——waiver includes, but is not limited to damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's Owner's termination in accordance with Article 14. Nothing contained in this Section 15.1.6 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

In any suit or action arising under this Contract the Owner shall be entitled to an award of reasonable attorney's fees and costs of litigation.

## § 15.2 INITIAL DECISION

§ 15.2.1 Claims, excluding those arising under Sections 10.3, 10.4, 11.3.9, and 11.3.10, shall be referred to the Initial Decision Maker for initial decision. The Architect will serve as the Initial Decision Maker, unless otherwise indicated in the Agreement. Except for those Claims excluded by this Section 15.2.1, an initial decision shall be required as a condition precedent to mediation of any Claim arising prior to the date final payment is due, unless 30 days have passed after the Claim has been referred to the Initial Decision Maker with no decision having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner.

- § 16.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to resolve the Claim.
- § 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision. The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense.
- § 16.2.4 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of such request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part.
- § 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution.
- § 15.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section
- § 15.2.6.1 Either party may, within 30 days from the date of an initial decision, demand in writing that the other party file for mediation within 60 days of the initial decision. If such a demand is made and the party receiving the demand fails to file for mediation within the time required, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.
- § 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.
- § 15.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

#### § 15.3 MEDIATION

- § 15.3.1 Claims, disputes, or other matters in controversy arising out of or related to the Contract except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.6 shall be subject to mediation as a condition precedent to binding dispute resolution.
- § 15.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration is stayed pursuant to this Section 15.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

§ 15.3.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

#### § 15.4 ARBITRATION

- § 15.4.1 If the parties have selected arbitration as the method for binding dispute resolution in the Agreement, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.
- § 15.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.
- §-15.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.
- § 15.4.3 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

## § 15.4.4 CONSOLIDATION OR JOINDER

- § 15.4.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).
- § 15.4.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

#### §15.1.7 RESOLUTION OF DISPUTES

- §15.1.7.1 Venue. Any suit or action arising under this Contract shall be commenced in DuPage County, Illinois, but only after exhausting all possible administrative remedies.
- §15.1.7.2 Limitations On Contractor's Claims. No suit or action shall be maintained by Contractor, its successors or assigns, against Owner on any claim based upon or arising out of this Contract or out of anything done in connection with this Contract unless such action shall be commenced within one year of the termination of this Contract.
- §15.1.7.3 Waiver of Punitive Damages. The Contractor and Owner waive all claims against each other for all punitive damages arising out of or relating to this Contract, but nothing in this paragraph shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

§ 15.4.4.3 The Owner and Contractor grant to any person or entity made a party to an arbitration conducted under this Section 15.4, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Contractor under this Agreement.

# **Certification of Document's Authenticity**

AIA® Document D401™ - 2003

I, Nicole L. Karas, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 11:11:21 on 10/04/2016 under Order No. 1549258185\_1 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA® Document A201<sup>TM</sup> – 2007, General Conditions of the Contract for Construction, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

Nicole	Kaws			
(Signed)		<u> </u>		
Attorney				
(Title)			_	
10/4/16				
(Dated)			_	

# Specifications

## SPECIFICATIONS SECTION

### PRE-ENGINEERED BUILDING INSULATION

### PART 1 GENERAL

## 1.1 SECTION INCLUDES

- A. Pre-Engineered Building Insulation for New Construction.
- B. Pre-Engineered Building Insulation for Existing Construction.

## 1.2 RELATED SECTIONS

- A. Section 13121 Pre-Engineered Metal Buildings.
- B. Section 13900 Fire Protection Systems.
- C. Division 15 Mechanical; Rough-in utilities.
- D. Division 16 Electrical; Rough-in utilities

## 1.3 REFERENCES

- A. ASTM E 84 Standard Test Method for Surface Burning Characteristics of Building Materials
- B. ASTM E 96 Standard Test Method for Water Vapor Transmission of Materials in Sheet Form (Procedure B).
- C. ASTM E 665 Standard Specification for Mineral-Fiber Blanket Thermal Insulation for Light Frame Construction and Manufactured Housing.
- NFPA 255 Standard Method of Test of Surface Burning Characteristics of Building Materials.
- E. UL 723 Tests for Surface Burning Characteristics of Building Materials.
- F. ASTM C 1136 Standard Specification for Flexible, Low Permeance Vapor Retarders for Thermal Insulation.

## 1.4 DESIGN REQUIREMENTS

- A. Thermal Resistance of Installed System: R-Value of 19.
- B. Insulation system shall have a continuous vapor barrier inside of building purlins, girts, and insulation to provide complete isolation from inside conditioned air.

### 1.5 SUBMITALS

- A. Submit under provisions of Section 01300.
- B. [Product Data]: Manufacturer's data sheet on each product to be used, including:
  - 1. Preparation instructions and recommendations.
  - 2. Storage and handling requirements and recommendations.

- 3. Installation instructions.
- C. Shop Drawings: Indicate locations of connections and attachments, general details, anchorages and method of anchorage and installation.
- D. Verification Samples: For each finish product specified, two samples, minimum size 6 inches (150 mm) square or long, representing actual products required for this project.
- D. Manufacturer's Certificate: Certify products meet or exceed specified requirements.

## 1.6 QUALITY ASSURANCE

- A. Manufacturer Qualifications: Company specializing in manufacturing product systems specified in this section with minimum five years documented experience.
- B. Installer Qualifications: Company with a minimum of five years documented experience installing in tennis facilities.

## 1.7 DELIVERY, STORAGE, AND HANDLING

- A. Store products in manufacturer's unopened packaging until ready for installation.
- B. Store products indoors and protect from moisture, construction traffic, and damage.

### 1.8 PROJECT CONDITIONS

A. Maintain environmental conditions (temperature, humidity, and ventilation) within limits recommended by manufacturer for optimum results. Do not install products under environmental conditions outside manufacturer's absolute limits.

## PART 2 PRODUCTS

## 2.1 MANUFACTURERS

- A. Acceptable Manufacturer: Sports Interiors, Inc. 23380 Indian Creek Road, Prairie View, IL 60069. Tel: (847) 948-5335 fax (847) 948-0162. Email: <a href="mailto:sportsinteriors@hotmail.com">sportsinteriors@hotmail.com</a>, <a href="mailto:www.sports-interiors.com">www.sports-interiors.com</a> or pre-approved equal.
- Requests for substitutions will be considered in accordance with provisions of Section L
  of the Instructions to Bidders.

## 2.2 MATERIALS

- A. SporTuff System, or approved equal, consists of Batt Insulation, Roof Insulation, Wall Insulation, Vapor Barrier Liner Fabric, Thermal Breaks, Straps, and other devices and components in a proprietary insulation system as follows:
  - 1. Batt Insulation: ASTM C 665; preformed glass fiber batt conforming to the following:
    - a. Thermal Resistance: R of 19 (RSI of 152 mm).
    - b. Batt Size: Equal to purlin/girt spacing by manufacturer's standard lengths.
    - c. Unfaced.
  - 2. Roof Insulation: Fiberglass batt or fiberglass blanket complying with ASTM C 665 and ASTM E 84 with thermal resistance and thickness as follows:
    - a. As indicated on the drawings.
    - b. R-19; 6 inches (152 mm).

- 3. Wall Insulation: Fiberglass blanket of batt complying with ASTM C 665 and ASTM E 84 with thermal resistance and thickness as follows:
  - a. R-19; 6 inches (152 mm)
- 4. Vapor Barrier Liner Fabric: SporTuff Syseal type woven, reinforced, high-density polyethylene yarns coated on both sides with a continuous white polyethylene coating, as follows:
  - a. Product complies with ASTM C 1136, Types 1 through VI.
  - b. Perm rating: 0.02 for fabric and for seams in accordance with ASTM E
  - c. Flame/Smoke Properties:
    - 1) 25/50 in accordance with ASTM E 84
    - Self-extinguishes with field test using matches or butane lighter.
  - d. Ultra violet radiation inhibitor to minimum UVMax rating of 8
  - e. Size and seaming: Manufactured in large custom pieces by extrusion welding from roll goods, and fabricated to substantially fit defined building area with minimum practicable job site sealing.
  - f. Provide with factory double, extrusion-welded seams. Stapled seams or heat-melted seams are not acceptable due to degradation of fabric.
  - g. Factory-folded to allow for rapid installation
  - h. Color: High reflective white
- Vapor Barrier Lap Sealant: Solvent based, SporTuff polyethylene fabric adhesive.
- 6. Vapor Barrier Tape: Double-sided sealant tape ¾ inch (19mm) wide by 1/32 inch (.79 mm) thick.
- 7. Vapor Barrier Patch Tape: Single-sided, adhesive backed sealant tape 3 inches (76 mm) wide made from same material as Syseal type liner fabric.
- 8. Straps:
  - a. 100 KSI minimum yield tempered high-tensile-strength steel.
  - b. Size: Not less than 0.020 inch (0.50 mm) thick by 1 inch (25 mm) by continuous length.
  - c. Galvanized, primed, and painted to match finished white color on the exposed side.
- 9. Fasteners:
  - a. For light gage steel: #12 by ¾ (19 mm) inch plated Tek 2 type screws with sealing washer, painted to match specified color.
  - b. For heavy gage steel" #12 by 1-1/2 inch (38 mm) plated Tek 4 type screws with sealing washer, painted to match specified color.

## PART 3 EXECUTION

## 3.1 EXAMINATION

- A. Verify that building structure including all bracing and any concealed building systems are completed and approved prior to installing liner system and insulation in the structure.
- B. Correct any unsatisfactory conditions before proceeding.
- C. If conditions are the responsibility of another installer, notify the Tennis Center Club Manager of unsatisfactory preparation before proceeding

## 3.2 INSTALLATION - GENERAL

- A. Install pre-engineered building insulation system in accordance with manufacturer's installation instructions and the approved shop drawings.
- B. Prepare surfaced using the methods recommended by the manufacturer for achieving the best results for the substrate under the project conditions.
- C. Install in purlin spaces without gaps or voids. Do not compress insulation.
- D. Trim insulation neatly to fit spaces.
- E. Fit insulation tight in spaces and tight to exterior side of sealed liner fabric and around mechanical and electrical services within plane of insulation.

## 3.3 ROOF INSULATION INSTALLATION

## A. Straps:

- 1. Cut straps to length and install in the pattern and spacing indicated on shop drawings.
- 2. Tension straps to required value.

## B. Vapor Barrier Fabric:

- 1. Install vapor barrier fabric in large one piece custom fabricated pieces to substantially fit defined building areas with minimum practicable job site sealing.
- 2. Position pre-folded fabric on the strap platform along one peak purlin.
- 3. Attach the two top corners at the peak and also centered on the bay.
- 4. Pull the material from the peak to the eave line and fasten at all intersections of the purlins and straps.
- 5. Repeat process for the material to other eave, sealing the two pieces together with sealant and strap running along purlin.
- 6. Trim edges and seal along the rafters.
- 7. All seams must be completely sealed and stapled seams are not acceptable.

## C. Insulation:

- 1. Unpack, and shake to a thickness exceeding the specified thickness.
- 2. Ensure that cavities are filled completely with insulation.
- 3. Pull through the vapor barrier liner fabric without any voids or gaps.
- D. Seal vapor barrier fabric to the wall fabric (to eave purlin if wall material is not being installed) and elsewhere as required to provide a continuous vapor barrier.

## 3.4 CLEANING

- A. Clean dirt or exposed sealant from the exposed vapor barrier fabric
- B. Remove scraps and debris from the site.

## 3.5 PROTECTION

- A. Protect system products until completion of installation.
- B. Repair or replace damaged products before completion of insulation system installation.

## 3.6 SCHEDULE

Project should be completed during Nov 21<sup>st</sup> 2016 – Apr 3<sup>rd</sup> 2017 period.

END OF SECTION 608449

# Appendix

• Oak Brook Park District Prevailing Wage Resolution and the July 2015 pay rates from the Illinois Department of Labor.

## **RESOLUTION NO.** <u>16-0620</u>

## A RESOLUTION OF THE OAK BROOK PARK DISTRICT REGARDING ILLINOIS PREVAILING WAGE ACT

WHEREAS, the State of Illinois has enacted "the Prevailing Wage Act," as amended, 820 ILCS 130/0.01through 130/12 (the "Act"); and

WHEREAS, the Act requires that the Board of Park Commissioners of the Oak Brook Park District (the "District") investigate and ascertain for the District the prevailing rate of wages as defined in the Act, for laborers, mechanics, and other workers in the locality of the District employed in performing construction or demolition of public works for the Park District.

# NOW, THEREFORE, BE IT RESOLVED BY THE BOARD OF PARK COMMISSIONERS OF THE OAK BROOK PARK DISTRICT, AS FOLLOWS:

Section 1. To the extent and as required by the Act, the general prevailing rate of wages in this locality for laborers, mechanics and other workers engaged in the construction and demolition of public works coming under the jurisdiction of the District is hereby ascertained to be the same as the prevailing rate of wages for construction and demolition work in DuPage County as determined by the Department of Labor of the State Of Illinois (the "Department") as of July 1, 2015 (the latest Illinois Prevailing Wage Rates available on the Illinois Department of Labor website), a copy of those determinations being attached hereto and incorporated herein by reference. As required by said Act, any and all revisions of the prevailing rate of wages by the Department shall supersede the Department's July 2015 determination and apply to any and all public works construction or demolition undertaken by the District. The definition of any terms appearing in this Resolution, which are also used in aforesaid Act, shall be the same as in the Act.

Section 2. Nothing herein contained is intended to apply nor shall be construed to apply said general prevailing rate of wages, as herein ascertained, to any work or employment performed on behalf of the District except public works construction or demolition of the District to the extent required by the Act.

Section 3. The Secretary of the District's Board of Park Commissioners shall publicly post or keep available for inspection by any interested party in the main office of the District this determination of the prevailing rate of wages. A copy of this determination or of the current revised determination of prevailing rate of wages then in effect shall be attached to all contract specifications, and for works awarded to a contractor without a public bid, contract or project specification, shall be attached to the purchase order related to the work to be done or in a separate document.

Section 4. The Secretary of the District's Board of Park Commissioners shall mail a copy of this Resolution to any employer, and to any association of employers and to any person or association of employees who have filed or file their names and addresses, requesting copies of any determination stating the particular rates and the particular class of workers whose wages will be affected by such rates.

<u>Section 5.</u> The Secretary of the District's Board of Park Commissioners shall promptly file a certified copy of the Resolution with the Department of Labor of the State of Illinois.

Section 6. Within thirty (30) days after filing a certified copy of this Resolution with the Illinois Department of Labor, the Secretary of the District's Board of Park Commissioners shall cause to be published in a newspaper of general circulation within the area a notice that this determination is effective and constitutes the determination of the District.

## ADOPTED BY ROLL CALL VOTE THIS 20th day of June, 2016

AYES: _	Truedson,	Tan	Carson	and	Knitter
NAYS:	Trombet	ta			
ABSEN	Γ:				

APPROVED:

President

STATE OF ILLINOIS	)	
	)	SS
COUNTY OF DU PAGE	)	

## **CERTIFICATION**

I, the undersigned, do hereby certify that I am the Secretary of Oak Brook Park District, DuPage County, Illinois, and as such official, I am keeper of the records, Resolutions, files and seal of said Park District, and I HEREBY CERTIFY that the foregoing instrument is a true and correct copy of Resolution <u>16-0620</u>

## A RESOLUTION OF THE OAK BROOK PARK DISTRICT REGARDING ILLINOIS PREVAILING WAGE ACT

adopted at a duly called Regular Meeting of the Board of Park Commissioners of the Oak Brook Park District, held in Oak Brook, Illinois, in said Park District at 6:30 p.m. on the 20<sup>nd</sup> day of June, 2016.

I do further certify that the deliberations of the Board on the adoption of said resolution were conducted openly, that the vote on the adoption of said Resolution was taken openly, that said meeting was called and held at a specified time and place convenient to the public, that notice of said meeting was duly given to all of the news media requesting such notice, that said meeting was called and held in strict compliance with the provisions of the Open Meetings Act of the State of Illinois, as amended, and with the provisions of the Park District Code of the State of Illinois, as amended, and that the Board has complied with all of the provisions of said Act and said Code and with all of the procedural rules of the Board.

IN WITNESS WHEREOF, I hereunto affix my official signature and the seal of said Park District this 20<sup>nd</sup> day of June, 2016.

Oak Brook Park District

## **Du Page County Prevailing Wage for July 2015**

(See explanation of column headings at bottom of wages)

Trade Name				Base	FRMAN 1						Vac	_		
ASBESTOS ABT-GEN	(Accepted)	ALL	5576		39.950					10.72				
ASBESTOS ABT-MEC BOILERMAKER		BLD			38.840									
		BLD		47.070										
BRICK MASON		BLD ALL			48.160					14.43				
CARPENTER CEMENT MASON		ALL			46.350					16.39 14.45				
CERAMIC TILE FNSHER		BLD			0.000					9.230				
COMMUNICATION TECH		BLD			34.750					15.16				
ELECTRIC PWR EQMT OP		ALL			51.480					11.75				
ELECTRIC PWR EQMT OP		HWY		39.220	53.290	1.5				12.17				
ELECTRIC PWR GRNDMAN		ALL			51.480					9.090				
ELECTRIC PWR GRNDMAN		HWY			53.290					9.400				
ELECTRIC PWR LINEMAN ELECTRIC PWR LINEMAN		${ t ALL}$			51.480					14.06				
ELECTRIC PWR TRK DRV		ALL			53.290 51.480					14.56				
ELECTRIC PWR TRK DRV		HWY			53.290					9.730				
ELECTRICIAN		BLD			41.980					18.29				
ELEVATOR CONSTRUCTOR		BLD			57.150					14.21				
FENCE ERECTOR	NE	ALL		37.340	39.340	1.5	1.5	2.0	13.05	12.06	0.000	0.300		
FENCE ERECTOR	M	ALL			48.660		2.0	2.0	10.52	20.76	0.000	0.700		
GLAZIER		BLD			42.000					16.99				
HT/FROST INSULATOR		BLD			50.950					12.16				
IRON WORKER IRON WORKER	E	ALL			46.200 48.660					21.14				
LABORER	VV	ALL ALT.		39 200	39.950					10.72				
LATHER		ALL		44.350	46.350					16.39				
MACHINIST		BLD		45.350	47.850					8.950				
MARBLE FINISHERS		ALL BLD ALL		32.400	34.320					13.75				
MARBLE MASON		BLD		43.030	47.330	1.5	1.5	2.0	10.05	14.10	0.000	0.780		
MATERIAL TESTER I		ALL			0.000		1.5	2.0	13.98	10.72	0.000	0.500		
MATERIALS TESTER II		ALL			0.000					10.72				
MILLWRIGHT		ALL	1		46.350					16.39				
OPERATING ENGINEER OPERATING ENGINEER				48.100						12.65 12.65				
OPERATING ENGINEER				44.250						12.65				
OPERATING ENGINEER				42.500						12.65				
OPERATING ENGINEER				51.850						12.65				
OPERATING ENGINEER		BLD	6	49.100	52.100	2.0	2.0	2.0	17.55	12.65	1.900	1.250		
OPERATING ENGINEER			7	51.100						12.65				
OPERATING ENGINEER		FLT	7		36.000					11.80				
OPERATING ENGINEER OPERATING ENGINEER				46.300										
OPERATING ENGINEER				45.750										
OPERATING ENGINEER				42.300						12.65				
OPERATING ENGINEER				41.100										
OPERATING ENGINEER		HWY	6	49.300	50.300	1.5	1.5	2.0	17.55	12.65	1.900	1.250		
OPERATING ENGINEER			7	47.300			1.5	2.0	17.55	12.65	1.900	1.250		
ORNAMNTL IRON WORKER					47.500					17.94				
ORNAMNTL IRON WORKER	W				48.660					20.76				
PAINTER PAINTER SIGNS		ALL BLD			43.730 38.090					8.200				
PILEDRIVER		ALL			46.350					16.39				
PIPEFITTER		BLD			49.000					15.85				
PLASTERER		BLD		43.430	46.040	1.5				14.43				
PLUMBER		BLD		46.650	48.650	1.5	1.5	2.0	13.18	11.46	0.000	0.880		
ROOFER		BLD			44.000					10.54				
SHEETMETAL WORKER		BLD			46.720					13.31				
SPRINKLER FITTER	-	BLD			51.200					9.650				
STEEL ERECTOR STEEL ERECTOR		ALL ALL			44.070					19.59 20.76				
STONE MASON	VV.	BLD			48.160					14.43				
	>]		ΙN	EFFECT					50 1.5			2.97 9.930	0.000	0.500
TERRAZZO FINISHER	92.0	BLD	08382		0.000					11.22				
TERRAZZO MASON		BLD		41.880	44.880	1.5				12.51				
TILE MASON		BLD			47.840					11.40				
TRAFFIC SAFETY WRKR		HWY			34.350					6.450				
TRUCK DRIVER TRUCK DRIVER				35.920 32.700						8.760				
TRUCK DRIVER				32.700						4.350				
IJOH DILLYDIN			J	52.500	33.100	1.5	1.5	2.0	0.500	1.550	5.000	0.100		

TRUCK DRIVER

ALL 4 33.100 33.100 1.5 1.5 2.0 6.500 4.350 0.000 0.150 BLD 42.620 43.620 1.5 1.5 2.0 10.05 13.34 0.000 0.670

Legend: RG (Region)

TYP (Trade Type - All, Highway, Building, Floating, Oil & Chip, Rivers)

C (Class)

Base (Base Wage Rate)

FRMAN (Foreman Rate)

M-F>8 (OT required for any hour greater than 8 worked each day, Mon through Fri.

OSA (Overtime (OT) is required for every hour worked on Saturday)

OSH (Overtime is required for every hour worked on Saturday)

H/W (Health & Welfare Insurance)

Pensn (Pension)

Vac (Vacation)

Trng (Training)

### Explanations

DUPAGE COUNTY

IRON WORKERS AND FENCE ERECTOR (WEST) - West of Route 53.

The following list is considered as those days for which holiday rates of wages for work performed apply: New Years Day, Memorial Day, Fourth of July, Labor Day, Thanksgiving Day, Christmas Day and Veterans Day in some classifications/counties. Generally, any of these holidays which fall on a Sunday is celebrated on the following Monday. This then makes work performed on that Monday payable at the appropriate overtime rate for holiday pay. Common practice in a given local may alter certain days of celebration. If in doubt, please check with IDOL.

#### EXPLANATION OF CLASSES

ASBESTOS - GENERAL - removal of asbestos material/mold and hazardous materials from any place in a building, including mechanical systems where those mechanical systems are to be removed. This includes the removal of asbestos materials/mold and hazardous materials from ductwork or pipes in a building when the building is to be demolished at the time or at some close future date.

ASBESTOS - MECHANICAL - removal of asbestos material from mechanical systems, such as pipes, ducts, and boilers, where the mechanical systems are to remain.

TRAFFIC SAFETY - work associated with barricades, horses and drums used to reduce lane usage on highway work, the installation and removal of temporary lane markings, and the installation and removal of temporary road signs.

### CERAMIC TILE FINISHER

The grouting, cleaning, and polishing of all classes of tile, whether for interior or exterior purposes, all burned, glazed or unglazed products; all composition materials, granite tiles, warning detectable tiles, cement tiles, epoxy composite materials, pavers, glass, mosaics, fiberglass, and all substitute materials, for tile made in tile-like units; all mixtures in tile like form of cement, metals, and other materials that are for and intended for use as a finished floor surface, stair treads, promenade roofs, walks, walls, ceilings, swimming pools, and all other places where tile is to form a finished interior or exterior. The mixing of all setting mortars including but not limited to thin-set mortars, epoxies, wall mud, and any other sand and cement mixtures or adhesives when used in the preparation, installation, repair, or maintenance of tile and/or similar materials. The handling and unloading of all sand, cement, lime, tile, fixtures, equipment, adhesives, or any other materials to be used in the preparation, installation, repair, or maintenance of tile and/or similar materials. Ceramic Tile Finishers shall fill all joints and voids regardless of method on all tile work, particularly and especially after installation of said tile work. Application of any and all protective coverings to all types of tile installations including, but not be limited to, all soap compounds, paper products, tapes, and all polyethylene coverings, plywood, masonite, cardboard, and any new type of products that may be used to protect tile installations, Blastrac equipment, and all floor scarifying equipment used in preparing floors to receive tile. The clean up and removal of all waste and materials. All demolition of existing tile floors and walls to be re-tiled.

### COMMUNICATIONS TECHNICIAN

Low voltage installation, maintenance and removal of telecommunication facilities (voice, sound, data and video) including telephone and data inside wire, interconnect, terminal equipment, central offices, PABX, fiber optic cable and equipment, micro waves, V-SAT, bypass, CATV, WAN (wide area networks), LAN (local area networks), and ISDN (integrated system digital network), pulling of wire in raceways, but not the installation of raceways.

MARBLE FINISHER

Loading and unloading trucks, distribution of all materials (all stone, sand, etc.), stocking of floors with material, performing all rigging for heavy work, the handling of all material that may be needed for the installation of such materials, building of scaffolding, polishing if needed, patching, waxing of material if damaged, pointing up, caulking, grouting and cleaning of marble, holding water on diamond or Carborundum blade or saw for setters cutting, use of tub saw or any other saw needed for preparation of material, drilling of holes for wires that anchor material set by setters, mixing up of molding plaster for installation of material, mixing up thin set for the installation of material, mixing up of sand to cement for the installation of material and such other work as may be required in helping a Marble Setter in the handling of all material in the erection or installation of interior marble, slate, travertine, art marble, serpentine, alberene stone, blue stone, granite and other stones (meaning as to stone any foreign or domestic materials as are specified and used in building interiors and exteriors and customarily known as stone in the trade), carrara, sanionyx, vitrolite and similar opaque glass and the laying of all marble tile, terrazzo tile, slate tile and precast tile, steps, risers treads, base, or any other materials that may be used as substitutes for any of the aforementioned materials and which are used on interior and exterior which are installed in a similar manner.

MATERIAL TESTER I: Hand coring and drilling for testing of materials; field inspection of uncured concrete and asphalt.

MATERIAL TESTER II: Field inspection of welds, structural steel, fireproofing, masonry, soil, facade, reinforcing steel, formwork, cured concrete, and concrete and asphalt batch plants; adjusting proportions of bituminous mixtures.

### OPERATING ENGINEER - BUILDING

Class 1. Asphalt Plant; Asphalt Spreader; Autograde; Backhoes with Caisson Attachment; Batch Plant; Benoto (requires Two Engineers); Boiler and Throttle Valve; Caisson Rigs; Central Redi-Mix Plant; Combination Back Hoe Front End-loader Machine; Compressor and Throttle Valve; Concrete Breaker (Truck Mounted); Concrete Conveyor; Concrete Conveyor (Truck Mounted); Concrete Paver Over 27E cu. ft; Concrete Paver 27E cu. ft. and Under: Concrete Placer; Concrete Placing Boom; Concrete Pump (Truck Mounted); Concrete Tower; Cranes, All; Cranes, Hammerhead; Cranes, (GCI and similar Type); Creter Crane; Spider Crane; Crusher, Stone, etc.; Derricks, All; Derricks, Traveling; Formless Curb and Gutter Machine; Grader, Elevating; Grouting Machines; Heavy Duty Self-Propelled Transporter or Prime Mover; Highlift Shovels or Front Endloader 2-1/4 yd. and over; Hoists, Elevators, outside type rack and pinion and similar machines; Hoists, One, Two and Three Drum; Hoists, Two Tugger One Floor; Hydraulic Backhoes; Hydraulic Boom Trucks; Hydro Vac (and similar equipment); Locomotives, All; Motor Patrol; Lubrication Technician; Manipulators; Pile Drivers and Skid Rig; Post Hole Digger; Pre-Stress Machine; Pump Cretes Dual Ram; Pump Cretes: Squeeze Cretes-Screw Type Pumps; Gypsum Bulker and Pump; Raised and Blind Hole Drill; Roto Mill Grinder; Scoops - Tractor Drawn; Slip-Form Paver; Straddle Buggies; Operation of Tie Back Machine; Tournapull; Tractor with Boom and Side Boom; Trenching Machines.

Class 2. Boilers; Broom, All Power Propelled; Bulldozers; Concrete Mixer (Two Bag and Over); Conveyor, Portable; Forklift Trucks; Highlift Shovels or Front Endloaders under 2-1/4 yd.; Hoists, Automatic; Hoists, Inside Elevators; Hoists, Sewer Dragging Machine; Hoists, Tugger Single Drum; Laser Screed; Rock Drill (Self-Propelled); Rock Drill (Truck Mounted); Rollers, All; Steam Generators; Tractors, All; Tractor Drawn Vibratory Roller; Winch Trucks with "A" Frame.

Class 3. Air Compressor; Combination Small Equipment Operator; Generators; Heaters, Mechanical; Hoists, Inside Elevators (remodeling or renovation work); Hydraulic Power Units (Pile Driving, Extracting, and Drilling); Pumps, over 3" (1 to 3 not to exceed a total of 300 ft.); Low Boys; Pumps, Well Points; Welding Machines (2 through 5); Winches, 4 Small Electric Drill Winches.

Class 4. Bobcats and/or other Skid Steer Loaders; Oilers; and Brick Forklift.

Class 5. Assistant Craft Foreman.

Class 6. Gradall.

Class 7. Mechanics; Welders.

### OPERATING ENGINEERS - HIGHWAY CONSTRUCTION

Class 1. Asphalt Plant; Asphalt Heater and Planer Combination; Asphalt Heater Scarfire; Asphalt Spreader; Autograder/GOMACO or other similar type machines: ABG Paver; Backhoes with Caisson Attachment; Ballast Regulator; Belt Loader; Caisson Rigs; Car Dumper; Central Redi-Mix Plant; Combination Backhoe Front Endloader Machine, (1 cu. yd. Backhoe Bucket or over or with attachments); Concrete Breaker (Truck Mounted); Concrete Conveyor; Concrete Paver over 27E cu. ft.; Concrete

Placer; Concrete Tube Float; Cranes, all attachments; Cranes, Tower Cranes of all types: Creter Crane: Spider Crane; Crusher, Stone, etc.; Derricks, All; Derrick Boats; Derricks, Traveling; Dredges; Elevators, Outside type Rack & Pinion and Similar Machines; Formless Curb and Gutter Machine; Grader, Elevating; Grader, Motor Grader, Motor Patrol, Auto Patrol, Form Grader, Pull Grader, Subgrader; Guard Rail Post Driver Truck Mounted; Hoists, One, Two and Three Drum; Heavy Duty Self-Propelled Transporter or Prime Mover; Hydraulic Backhoes; Backhoes with shear attachments up to 40' of boom reach; Lubrication Technician; Manipulators; Mucking Machine; Pile Drivers and Skid Rig; Pre-Stress Machine; Pump Cretes Dual Ram; Rock Drill - Crawler or Skid Rig; Rock Drill - Truck Mounted; Rock/Track Tamper; Roto Mill Grinder; Slip-Form Paver; Snow Melters; Soil Test Drill Rig (Truck Mounted); Straddle Buggies; Hydraulic Telescoping Form (Tunnel); Operation of Tieback Machine; Tractor Drawn Belt Loader; Tractor Drawn Belt Loader (with attached pusher - two engineers); Tractor with Boom; Tractaire with Attachments; Traffic Barrier Transfer Machine; Trenching; Truck Mounted Concrete Pump with Boom; Raised or Blind Hole Drills (Tunnel Shaft); Underground Boring and/or Mining Machines 5 ft. in diameter and over tunnel, etc; Underground Boring and/or Mining Machines under 5 ft. in diameter; Wheel Excavator; Widener (APSCO).

Class 2. Batch Plant; Bituminous Mixer; Boiler and Throttle Valve; Bulldozers; Car Loader Trailing Conveyors; Combination Backhoe Front Endloader Machine (Less than 1 cu. yd. Backhoe Bucket or over or with attachments); Compressor and Throttle Valve; Compressor, Common Receiver (3); Concrete Breaker or Hydro Hammer; Concrete Grinding Machine; Concrete Mixer or Paver 7S Series to and including 27 cu. ft.; Concrete Spreader; Concrete Curing Machine, Burlap Machine, Belting Machine and Sealing Machine; Concrete Wheel Saw; Conveyor Muck Cars (Haglund or Similar Type); Drills, All; Finishing Machine -Concrete; Highlift Shovels or Front Endloader; Hoist - Sewer Dragging Machine; Hydraulic Boom Trucks (All Attachments); Hydro-Blaster; Hydro Excavating (excluding hose work); Laser Screed; All Locomotives, Dinky; Off-Road Hauling Units (including articulating) Non Self-Loading Ejection Dump; Pump Cretes: Squeeze Cretes - Screw Type Pumps, Gypsum Bulker and Pump; Roller, Asphalt; Rotary Snow Plows; Rototiller, Seaman, etc., self-propelled; Self-Propelled Compactor; Spreader - Chip - Stone, etc.; Scraper - Single/Twin Engine/Push and Pull; Scraper - Prime Mover in Tandem (Regardless of Size); Tractors pulling attachments, Sheeps Foot, Disc, Compactor, etc.; Tug Boats.

Class 3. Boilers; Brooms, All Power Propelled; Cement Supply Tender; Compressor, Common Receiver (2); Concrete Mixer (Two Bag and Over); Conveyor, Portable; Farm-Type Tractors Used for Mowing, Seeding, etc.; Forklift Trucks; Grouting Machine; Hoists, Automatic; Hoists, All Elevators; Hoists, Tugger Single Drum; Jeep Diggers; Low Boys; Pipe Jacking Machines; Post-Hole Digger; Power Saw, Concrete Power Driven; Pug Mills; Rollers, other than Asphalt; Seed and Straw Blower; Steam Generators; Stump Machine; Winch Trucks with "A" Frame; Work Boats; Tamper-Form-Motor Driven.

Class 4. Air Compressor; Combination - Small Equipment Operator; Directional Boring Machine; Generators; Heaters, Mechanical; Hydraulic Power Unit (Pile Driving, Extracting, or Drilling); Light Plants, All (1 through 5); Pumps, over 3" (1 to 3 not to exceed a total of 300 ft.); Pumps, Well Points; Vacuum Trucks (excluding hose work); Welding Machines (2 through 5); Winches, 4 Small Electric Drill Winches.

Class 5. SkidSteer Loader (all); Brick Forklifts; Oilers.

Class 6. Field Mechanics and Field Welders

Class 7. Dowell Machine with Air Compressor; Gradall and machines of like nature.

OPERATING ENGINEER - FLOATING

Diver. Diver Wet Tender, Diver Tender, ROV Pilot, ROV Tender

SURVEY WORKER - Operated survey equipment including data collectors, G.P.S. and robotic instruments, as well as conventional levels and transits.

TRUCK DRIVER - BUILDING, HEAVY AND HIGHWAY CONSTRUCTION

Class 1. Two or three Axle Trucks. A-frame Truck when used for transportation purposes; Air Compressors and Welding Machines, including those pulled by cars, pick-up trucks and tractors; Ambulances; Batch Gate Lockers; Batch Hopperman; Car and Truck Washers; Carry-alls; Fork Lifts and Hoisters; Helpers; Mechanics Helpers and Greasers; Oil Distributors 2-man operation; Pavement Breakers; Pole Trailer, up to 40 feet; Power Mower Tractors; Self-propelled Chip Spreader; Skipman; Slurry Trucks, 2-man operation; Slurry Truck Conveyor Operation, 2 or 3 man; Teamsters; Unskilled Dumpman; and Truck Drivers hauling warning lights, barricades, and portable toilets on the job site.

Class 2. Four axle trucks; Dump Crets and Adgetors under 7 yards; Dumpsters, Track Trucks, Euclids, Hug Bottom Dump Turnapulls or Turnatrailers when pulling other than self-loading equipment or similar equipment under 16 cubic yards; Mixer Trucks under 7 yeards; Ready-mix Plant Hopper Operator, and Winch Trucks, 2 Axles.

Class 3. Five axle trucks; Dump Crets and Adgetors 7 yards and over; Dumpsters, Track Trucks, Euclids, Hug Bottom Dump Turnatrailers or turnapulls when pulling other than self-loading equipment or similar equipment over 16 cubic yards; Explosives and/or Fission Material Trucks; Mixer Trucks 7 yards or over; Mobile Cranes while in transit; Oil Distributors, 1-man operation; Pole Trailer, over 40 feet; Pole and Expandable Trailers hauling material over 50 feet long; Slurry trucks, 1-man operation; Winch trucks, 3 axles or more; Mechanic--Truck Welder and Truck Painter.

Class 4. Six axle trucks; Dual-purpose vehicles, such as mounted crane trucks with hoist and accessories; Foreman; Master Mechanic; Self-loading equipment like P.B. and trucks with scoops on the front.

### TERRAZZO FINISHER

The handling of sand, cement, marble chips, and all other materials that may be used by the Mosaic Terrazzo Mechanic, and the mixing, grinding, grouting, cleaning and sealing of all Marble, Mosaic, and Terrazzo work, floors, base, stairs, and wainscoting by hand or machine, and in addition, assisting and aiding Marble, Masonic, and Terrazzo Mechanics.

### Other Classifications of Work:

For definitions of classifications not otherwise set out, the Department generally has on file such definitions which are available. If a task to be performed is not subject to one of the classifications of pay set out, the Department will upon being contacted state which neighboring county has such a classification and provide such rate, such rate being deemed to exist by reference in this document. If no neighboring county rate applies to the task, the Department shall undertake a special determination, such special determination being then deemed to have existed under this determination. If a project requires these, or any classification not listed, please contact IDOL at 217-782-1710 for wage rates or clarifications.

### LANDSCAPING

Landscaping work falls under the existing classifications for laborer, operating engineer and truck driver. The work performed by landscape plantsman and landscape laborer is covered by the existing classification of laborer. The work performed by landscape operators (regardless of equipment used or its size) is covered by the classifications of operating engineer. The work performed by landscape truck drivers (regardless of size of truck driven) is covered by the classifications of truck driver.

## MATERIAL TESTER & MATERIAL TESTER/INSPECTOR I AND II

Notwithstanding the difference in the classification title, the classification entitled "Material Tester I" involves the same job duties as the classification entitled "Material Tester/Inspector I". Likewise, the classification entitled "Material Tester II" involves the same job duties as the classification entitled "Material Tester/Inspector II".

## **Cook County Prevailing Wage for July 2015**

(See explanation of column headings at bottom of wages)

Trade Name				Base							Vac			
ASBESTOS ABT-GEN		ALL								10.72				
ASBESTOS ABT-MEC				36.340						10.96				
BOILERMAKER		BLD		47.070	51.300	2.0	2.0	2.0	6.970	18.13	0.000	0.400		
BRICK MASON		BLD		43.780						14.43				
CARPENTER		ALL		44.350						16.39				
BOILERMAKER BRICK MASON CARPENTER CEMENT MASON CERAMIC TILE FNSHER		ALL		43.750	0.000					14.45				
COMM. ELECT.		BLD			42.800					9.230 12.57				
COMM. ELECT. ELECTRIC PWR EQMT OP		ALL			51.100					14.87				
ELECTRIC PWR GRNDMAN		ALL			52.500					12.28				
ELECTRIC PWR LINEMAN		ALL			52.500					15.75				
ELECTRICIAN		ALL			48.000					15.27				
ELEVATOR CONSTRUCTOR FENCE ERECTOR		BLD			57.150					14.21				
CTACTED		DID			39.340 42.000					12.06 16.99				
HT/FROST INSULATOR IRON WORKER LABORER LATHER MACHINIST		BLD		48.450						12.16				
IRON WORKER		ALL		44.200						21.14				
LABORER		ALL		39.200	39.950	1.5	1.5	2.0	13.98	10.72	0.000	0.500		
LATHER		ALL		44.350						16.39		150 F. T.		
				45.350						8.950				
MARBLE FINISHERS MARBLE MASON		ALL BLD		32.400 43.030						13.75 14.10				
MATERIAL TESTER I		ALL		29.200						10.72				
MATERIALS TESTER II		ALL		34.200						10.72				
MILLWRIGHT		ALL			46.350		1.5	2.0	11.79	16.39	0.000	0.630		
OPERATING ENGINEER				48.100			2.0	2.0	17.55	12.65	1.900	1.250		
OPERATING ENGINEER OPERATING ENGINEER				46.800						12.65				
OPERATING ENGINEER OPERATING ENGINEER				44.250						12.65				
OPERATING ENGINEER				42.500 51.850						12.65 12.65				
OPERATING ENGINEER				49.100						12.65				
OPERATING ENGINEER				51.100						12.65				
OPERATING ENGINEER		FLT	1	53.600	53.600	1.5	1.5	2.0	17.10	11.80	1.900	1.250		
OPERATING ENGINEER				52.100						11.05				
OPERATING ENGINEER OPERATING ENGINEER				46.400						11.80				
OPERATING ENGINEER				55.100						11.80				
OPERATING ENGINEER				35.000						11.05				
OPERATING ENGINEER				46.300						12.65				
OPERATING ENGINEER				45.750						12.65				
OPERATING ENGINEER OPERATING ENGINEER		HWY	3	43.700	50.300	1.5	1.5	2.0	17.55	12.65	1.900	1.250		
OPERATING ENGINEER OPERATING ENGINEER				42.300 41.100										
OPERATING ENGINEER				49.300						12.65				
OPERATING ENGINEER				47.300										
ORNAMNTL IRON WORKER		ALL		45.000	47.500	2.0	2.0	2.0	13.55	17.94	0.000	0.650		
PAINTER		ALL		41.750						11.10				
PAINTER SIGNS		BLD			38.090					2.710				
PILEDRIVER PIPEFITTER		ALL BLD		44.350	49.000					16.39				
PLASTERER		BLD		43.430						15.85 14.43				
PLUMBER		BLD			48.650					11.46				
ROOFER		BLD		41.000			1.5	2.0	8.280	10.54	0.000	0.530		
SHEETMETAL WORKER		BLD		42.230						20.68				
SIGN HANGER		BLD		31.310						3.280				
SPRINKLER FITTER STEEL ERECTOR		BLD		49.200						9.650				
STONE MASON		BLD			48.160					19.59 14.43				
	>		ΙN	EFFECT	ALL				0 1.5			2.97 9.93	0 0.000	0.500
TERRAZZO FINISHER		BLD		38.040						11.22				
TERRAZZO MASON		BLD		41.880			1.5	2.0	10.55	12.51	0.000	0.940		
TILE MASON		BLD		43.840						11.40				
TRAFFIC SAFETY WRKR TRUCK DRIVER	F	HWY AT.I.	1	32.750						6.450				
TRUCK DRIVER				34.100						8.500				
TRUCK DRIVER				34.300						8.500				
TRUCK DRIVER				34.500						8.500				
TRUCK DRIVER	W			35.600						9.140				
TRUCK DRIVER	W	ALL	2	32.700	33.100	1.5	1.5	2.0	6.500	4.350	0.000	0.000		

TRUCK DRIVER W ALL 3 32.900 33.100 1.5 1.5 2.0 6.500 4.350 0.000 0.000 TRUCK DRIVER W ALL 4 33.100 33.100 1.5 1.5 2.0 6.500 4.350 0.000 0.000 TUCKPOINTER BLD 43.800 44.800 1.5 1.5 2.0 8.280 13.49 0.000 0.670

Legend: RG (Region)

TYP (Trade Type - All, Highway, Building, Floating, Oil & Chip, Rivers)

C (Class)

Base (Base Wage Rate)

FRMAN (Foreman Rate)

M-F>8 (OT required for any hour greater than 8 worked each day, Mon through Fri.

OSA (Overtime (OT) is required for every hour worked on Saturday)

OSH (Overtime is required for every hour worked on Sunday and Holidays)

H/W (Health & Welfare Insurance)

Pensn (Pension)

Vac (Vacation)

Trng (Training)

### **Explanations**

COOK COUNTY

The following list is considered as those days for which holiday rates of wages for work performed apply: New Years Day, Memorial Day, Fourth of July, Labor Day, Thanksgiving Day, Christmas Day and Veterans Day in some classifications/counties. Generally, any of these holidays which fall on a Sunday is celebrated on the following Monday. This then makes work performed on that Monday payable at the appropriate overtime rate for holiday pay. Common practice in a given local may alter certain days of celebration. If in doubt, please check with IDOL.

TRUCK DRIVERS (WEST) - That part of the county West of Barrington Road.

### EXPLANATION OF CLASSES

ASBESTOS - GENERAL - removal of asbestos material/mold and hazardous materials from any place in a building, including mechanical systems where those mechanical systems are to be removed. This includes the removal of asbestos materials/mold and hazardous materials from ductwork or pipes in a building when the building is to be demolished at the time or at some close future date.

ASBESTOS - MECHANICAL - removal of asbestos material from mechanical systems, such as pipes, ducts, and boilers, where the mechanical systems are to remain.

### CERAMIC TILE FINISHER

The grouting, cleaning, and polishing of all classes of tile, whether for interior or exterior purposes, all burned, glazed or unglazed products; all composition materials, granite tiles, warning detectable tiles, cement tiles, epoxy composite materials, pavers, glass, mosaics, fiberglass, and all substitute materials, for tile made in tile-like units; all mixtures in tile like form of cement, metals, and other materials that are for and intended for use as a finished floor surface, stair treads, promenade roofs, walks, walls, ceilings, swimming pools, and all other places where tile is to form a finished interior or exterior. The mixing of all setting mortars including but not limited to thin-set mortars, epoxies, wall mud, and any other sand and cement mixtures or adhesives when used in the preparation, installation, repair, or maintenance of tile and/or similar materials. The handling and unloading of all sand, cement, lime, tile, fixtures, equipment, adhesives, or any other materials to be used in the preparation, installation, repair, or maintenance of tile and/or similar materials. Ceramic Tile Finishers shall fill all joints and voids regardless of method on all tile work, particularly and especially after installation of said tile work. Application of any and all protective coverings to all types of tile installations including, but not be limited to, all soap compounds, paper products, tapes, and all polyethylene coverings, plywood, masonite, cardboard, and any new type of products that may be used to protect tile installations, Blastrac equipment, and all floor scarifying equipment used in preparing floors to receive tile. The clean up and removal of all waste and materials. All demolition of existing tile floors and walls to be re-tiled.

### COMMUNICATIONS ELECTRICIAN

Installation, operation, inspection, maintenance, repair and service of radio, television, recording, voice sound vision production and reproduction, telephone and telephone interconnect, facsimile, data apparatus, coaxial, fibre optic and wireless equipment, appliances and systems used for the transmission and reception of signals of any nature, business, domestic, commercial, education, entertainment, and residential purposes, including but not limited to, communication and telephone, electronic and sound equipment, fibre optic and data communication systems, and the performance of any task directly related to such installation or service whether at new or existing sites, such tasks to include the placing of wire and cable and electrical power conduit or other raceway work within the equipment room and pulling wire and/or cable through conduit and the

installation of any incidental conduit, such that the employees covered hereby can complete any job in full.

### MARBLE FINISHER

Loading and unloading trucks, distribution of all materials (all stone, sand, etc.), stocking of floors with material, performing all rigging for heavy work, the handling of all material that may be needed for the installation of such materials, building of scaffolding, polishing if needed, patching, waxing of material if damaged, pointing up, caulking, grouting and cleaning of marble, holding water on diamond or Carborundum blade or saw for setters cutting, use of tub saw or any other saw needed for preparation of material, drilling of holes for wires that anchor material set by setters, mixing up of molding plaster for installation of material, mixing up thin set for the installation of material, mixing up of sand to cement for the installation of material and such other work as may be required in helping a Marble Setter in the handling of all material in the erection or installation of interior marble, slate, travertine, art marble, serpentine, alberene stone, blue stone, granite and other stones (meaning as to stone any foreign or domestic materials as are specified and used in building interiors and exteriors and customarily known as stone in the trade), carrara, sanionyx, vitrolite and similar opaque glass and the laying of all marble tile, terrazzo tile, slate tile and precast tile, steps, risers treads, base, or any other materials that may be used as substitutes for any of the aforementioned materials and which are used on interior and exterior which are installed in a similar manner.

MATERIAL TESTER I: Hand coring and drilling for testing of materials; field inspection of uncured concrete and asphalt.

MATERIAL TESTER II: Field inspection of welds, structural steel, fireproofing, masonry, soil, facade, reinforcing steel, formwork, cured concrete, and concrete and asphalt batch plants; adjusting proportions of bituminous mixtures.

### OPERATING ENGINEER - BUILDING

Class 1. Asphalt Plant; Asphalt Spreader; Autograde; Backhoes with Caisson Attachment; Batch Plant; Benoto (requires Two Engineers); Boiler and Throttle Valve; Caisson Rigs; Central Redi-Mix Plant; Combination Back Hoe Front End-loader Machine; Compressor and Throttle Valve; Concrete Breaker (Truck Mounted); Concrete Conveyor; Concrete Conveyor (Truck Mounted); Concrete Paver Over 27E cu. ft; Concrete Paver 27E cu. ft. and Under: Concrete Placer; Concrete Placing Boom; Concrete Pump (Truck Mounted); Concrete Tower; Cranes, All; Cranes, Hammerhead; Cranes, (GCI and similar Type); Creter Crane; Spider Crane; Crusher, Stone, etc.; Derricks, All; Derricks, Traveling; Formless Curb and Gutter Machine; Grader, Elevating; Grouting Machines; Heavy Duty Self-Propelled Transporter or Prime Mover; Highlift Shovels or Front Endloader 2-1/4 yd. and over; Hoists, Elevators, outside type rack and pinion and similar machines; Hoists, One, Two and Three Drum; Hoists, Two Tugger One Floor; Hydraulic Backhoes; Hydraulic Boom Trucks; Hydro Vac (and similar equipment); Locomotives, All; Motor Patrol; Lubrication Technician; Manipulators; Pile Drivers and Skid Rig; Post Hole Digger; Pre-Stress Machine; Pump Cretes Dual Ram; Pump Cretes: Squeeze Cretes-Screw Type Pumps; Gypsum Bulker and Pump; Raised and Blind Hole Drill; Roto Mill Grinder; Scoops - Tractor Drawn; Slip-Form Paver; Straddle Buggies; Operation of Tie Back Machine; Tournapull; Tractor with Boom and Side Boom; Trenching Machines.

Class 2. Boilers; Broom, All Power Propelled; Bulldozers; Concrete Mixer (Two Bag and Over); Conveyor, Portable; Forklift Trucks; Highlift Shovels or Front Endloaders under 2-1/4 yd.; Hoists, Automatic; Hoists, Inside Elevators; Hoists, Sewer Dragging Machine; Hoists, Tugger Single Drum; Laser Screed; Rock Drill (Self-Propelled); Rock Drill (Truck Mounted); Rollers, All; Steam Generators; Tractors, All; Tractor Drawn Vibratory Roller; Winch Trucks with "A" Frame.

Class 3. Air Compressor; Combination Small Equipment Operator; Generators; Heaters, Mechanical; Hoists, Inside Elevators (remodeling or renovation work); Hydraulic Power Units (File Driving, Extracting, and Drilling); Pumps, over 3" (1 to 3 not to exceed a total of 300 ft.); Low Boys; Pumps, Well Points; Welding Machines (2 through 5); Winches, 4 Small Electric Drill Winches.

Class 4. Bobcats and/or other Skid Steer Loaders; Oilers; and Brick Forklift.

Class 5. Assistant Craft Foreman.

Class 6. Gradall.

Class 7. Mechanics; Welders.

OPERATING ENGINEERS - HIGHWAY CONSTRUCTION

Class 1. Asphalt Plant; Asphalt Heater and Planer Combination; Asphalt Heater Scarfire; Asphalt Spreader; Autograder/GOMACO or other similar

type machines: ABG Paver; Backhoes with Caisson Attachment; Ballast Regulator; Belt Loader; Caisson Rigs; Car Dumper; Central Redi-Mix Plant; Combination Backhoe Front Endloader Machine, (1 cu. yd. Backhoe Bucket or over or with attachments); Concrete Breaker (Truck Mounted); Concrete Conveyor; Concrete Paver over 27E cu. ft.; Concrete Placer; Concrete Tube Float; Cranes, all attachments; Cranes, Tower Cranes of all types: Creter Crane: Spider Crane; Crusher, Stone, etc.; Derricks, All; Derrick Boats; Derricks, Traveling; Dredges; Elevators, Outside type Rack & Pinion and Similar Machines; Formless Curb and Gutter Machine; Grader, Elevating; Grader, Motor Grader, Motor Patrol, Auto Patrol, Form Grader, Pull Grader, Subgrader; Guard Rail Post Driver Truck Mounted; Hoists, One, Two and Three Drum; Heavy Duty Self-Propelled Transporter or Prime Mover; Hydraulic Backhoes; Backhoes with shear attachments up to 40' of boom reach; Lubrication Technician; Manipulators; Mucking Machine; Pile Drivers and Skid Rig; Pre-Stress Machine; Pump Cretes Dual Ram; Rock Drill - Crawler or Skid Rig; Rock Drill - Truck Mounted; Rock/Track Tamper; Roto Mill Grinder; Slip-Form Paver; Snow Melters; Soil Test Drill Rig (Truck Mounted); Straddle Buggies; Hydraulic Telescoping Form (Tunnel); Operation of Tieback Machine; Tractor Drawn Belt Loader; Tractor Drawn Belt Loader (with attached pusher - two engineers); Tractor with Boom; Tractaire with Attachments; Traffic Barrier Transfer Machine; Trenching; Truck Mounted Concrete Pump with Boom; Raised or Blind Hole Drills (Tunnel Shaft); Underground Boring and/or Mining Machines 5 ft. in diameter and over tunnel, etc; Underground Boring and/or Mining Machines under 5 ft. in diameter; Wheel Excavator; Widener (APSCO).

Class 2. Batch Plant; Bituminous Mixer; Boiler and Throttle Valve; Bulldozers; Car Loader Trailing Conveyors; Combination Backhoe Front Endloader Machine (Less than 1 cu. vd. Backhoe Bucket or over or with attachments); Compressor and Throttle Valve; Compressor, Common Receiver (3); Concrete Breaker or Hydro Hammer; Concrete Grinding Machine; Concrete Mixer or Paver 78 Series to and including 27 cu. ft.; Concrete Spreader; Concrete Curing Machine, Burlap Machine, Belting Machine and Sealing Machine; Concrete Wheel Saw; Conveyor Muck Cars (Haglund or Similar Type); Drills, All; Finishing Machine Concrete; Highlift Shovels or Front Endloader; Hoist - Sewer Dragging Machine; Hydraulic Boom Trucks (All Attachments); Hydro-Blaster; Hydro Excavating (excluding hose work); Laser Screed; All Locomotives, Dinky; Off-Road Hauling Units (including articulating) Non Self-Loading Ejection Dump; Pump Cretes: Squeeze Cretes - Screw Type Pumps, Gypsum Bulker and Pump; Roller, Asphalt; Rotary Snow Plows; Rototiller, Seaman, etc., self-propelled; Self-Propelled Compactor; Spreader - Chip - Stone, etc.; Scraper - Single/Twin Engine/Push and Pull; Scraper - Prime Mover in Tandem (Regardless of Size); Tractors pulling attachments, Sheeps Foot, Disc, Compactor, etc.; Tug Boats.

Class 3. Boilers; Brooms, All Power Propelled; Cement Supply Tender; Compressor, Common Receiver (2); Concrete Mixer (Two Bag and Over); Conveyor, Portable; Farm-Type Tractors Used for Mowing, Seeding, etc.; Forklift Trucks; Grouting Machine; Hoists, Automatic; Hoists, All Elevators; Hoists, Tugger Single Drum; Jeep Diggers; Low Boys; Pipe Jacking Machines; Post-Hole Digger; Power Saw, Concrete Power Driven; Pug Mills; Rollers, other than Asphalt; Seed and Straw Blower; Steam Generators; Stump Machine; Winch Trucks with "A" Frame; Work Boats; Tamper-Form-Motor Driven.

Class 4. Air Compressor; Combination - Small Equipment Operator; Directional Boring Machine; Generators; Heaters, Mechanical; Hydraulic Power Unit (Pile Driving, Extracting, or Drilling); Light Plants, All (1 through 5); Pumps, over 3" (1 to 3 not to exceed a total of 300 ft.); Pumps, Well Points; Vacuum Trucks (excluding hose work); Welding Machines (2 through 5); Winches, 4 Small Electric Drill Winches.

Class 5. SkidSteer Loader (all); Brick Forklifts; Oilers.

Class 6. Field Mechanics and Field Welders

Class 7. Dowell Machine with Air Compressor; Gradall and machines of like nature.

OPERATING ENGINEER - FLOATING

Class 1. Craft Foreman; Master Mechanic; Diver/Wet Tender; Engineer; Engineer (Hydraulic Dredge).

Class 2. Crane/Backhoe Operator; Boat Operator with towing endorsement; Mechanic/Welder; Assistant Engineer (Hydraulic Dredge); Leverman (Hydraulic Dredge); Diver Tender.

Class 3. Deck Equipment Operator, Machineryman, Maintenance of Crane (over 50 ton capacity) or Backhoe (115,000 lbs. or more); Tug/Launch Operator; Loader/Dozer and like equipment on Barge, Breakwater Wall, Slip/Dock, or Scow, Deck Machinery, etc.

Class 4. Deck Equipment Operator, Machineryman/Fireman (4 Equipment Units or More); Off Road Trucks; Deck Hand, Tug Engineer, Crane Maintenance (50 Ton Capacity and Under) or Backhoe Weighing (115,000 pounds or less); Assistant Tug Operator.

Class 5. Friction or Lattice Boom Cranes.

Class 6. ROV Pilot, ROV Tender

SURVEY WORKER - Operated survey equipment including data collectors, G.P.S. and robotic instruments, as well as conventional levels and transits.

### TERRAZZO FINISHER

The handling of sand, cement, marble chips, and all other materials that may be used by the Mosaic Terrazzo Mechanic, and the mixing, grinding, grouting, cleaning and sealing of all Marble, Mosaic, and Terrazzo work, floors, base, stairs, and wainscoting by hand or machine, and in addition, assisting and aiding Marble, Masonic, and Terrazzo Mechanics.

### TRAFFIC SAFETY

Work associated with barricades, horses and drums used to reduce lane usage on highway work, the installation and removal of temporary lane markings, and the installation and removal of temporary road signs.

TRUCK DRIVER - BUILDING, HEAVY AND HIGHWAY CONSTRUCTION - EAST & WEST

Class 1. Two or three Axle Trucks. A-frame Truck when used for transportation purposes; Air Compressors and Welding Machines, including those pulled by cars, pick-up trucks and tractors; Ambulances; Batch Gate Lockers; Batch Hopperman; Car and Truck Washers; Carry-alls; Fork Lifts and Hoisters; Helpers; Mechanics Helpers and Greasers; Oil Distributors 2-man operation; Pavement Breakers; Pole Trailer, up to 40 feet; Power Mower Tractors; Self-propelled Chip Spreader; Skipman; Slurry Trucks, 2-man operation; Slurry Truck Conveyor Operation, 2 or 3 man; Teamsters; Unskilled Dumpman; and Truck Drivers hauling warning lights, barricades, and portable toilets on the job site.

Class 2. Four axle trucks; Dump Crets and Adgetors under 7 yards; Dumpsters, Track Trucks, Euclids, Hug Bottom Dump Turnapulls or Turnatrailers when pulling other than self-loading equipment or similar equipment under 16 cubic yards; Mixer Trucks under 7 yards; Ready-mix Plant Hopper Operator, and Winch Trucks, 2 Axles.

Class 3. Five axle trucks; Dump Crets and Adgetors 7 yards and over; Dumpsters, Track Trucks, Euclids, Hug Bottom Dump Turnatrailers or turnapulls when pulling other than self-loading equipment or similar equipment over 16 cubic yards; Explosives and/or Fission Material Trucks; Mixer Trucks 7 yards or over; Mobile Cranes while in transit; Oil Distributors, 1-man operation; Pole Trailer, over 40 feet; Pole and Expandable Trailers hauling material over 50 feet long; Slurry trucks, 1-man operation; Winch trucks, 3 axles or more; Mechanic--Truck Welder and Truck Painter.

Class 4. Six axle trucks; Dual-purpose vehicles, such as mounted crane trucks with hoist and accessories; Foreman; Master Mechanic; Self-loading equipment like P.B. and trucks with scoops on the front.

### Other Classifications of Work:

For definitions of classifications not otherwise set out, the Department generally has on file such definitions which are available. If a task to be performed is not subject to one of the classifications of pay set out, the Department will upon being contacted state which neighboring county has such a classification and provide such rate, such rate being deemed to exist by reference in this document. If no neighboring county rate applies to the task, the Department shall undertake a special determination, such special determination being then deemed to have existed under this determination. If a project requires these, or any classification not listed, please contact IDOL at 217-782-1710 for wage rates or clarifications.

### LANDSCAPING

Landscaping work falls under the existing classifications for laborer, operating engineer and truck driver. The work performed by landscape plantsman and landscape laborer is covered by the existing classification of laborer. The work performed by landscape operators (regardless of equipment used or its size) is covered by the classifications of operating engineer. The work performed by landscape truck drivers (regardless of size of truck driven) is covered by the classifications of truck driver.

MATERIAL TESTER & MATERIAL TESTER/INSPECTOR I AND II

Notwithstanding the difference in the classification title, the classification entitled "Material Tester I" involves the same job duties as the classification entitled "Material Tester/Inspector I". Likewise, the classification entitled "Material Tester II" involves the same job duties as the classification entitled "Material Tester/Inspector II".